NUREG/CR-4374 BNL-NUREG-51917 Vol. 2

A Review of the Oconee-3 Probabilistic Risk Assessment

External Events, Core Damage Frequency

Prepared by N. A. Hanan, D. Ilberg, D. Xue, R. Youngblood, J. W. Reed, M. McCann, T. Talwani, J. Wreathall, P. D. Kurth, K. Bandyopadhyay, C. Costantino

Brookhaven National Laboratory

Prepared for U.S. Nuclear Regulatory Commission

> 8603280339 860331 PDR ADOCK 05000287 P PDR

NOTICE

This report was prepared as an account of work sponsored by an agency of the United States Government. Neither the United States Government nor any agency thereof, or any of their employees, makes any warranty, expressed or implied, or assumes any legal liability of responsibility for any third party's use, or the results of such use, of any information, apparatus, product or process disclosed in this report, or represents that its use by such third party would not infringe privately owned rights.

NOTICE

Availability of Reference Materials Cited in NRC Publications

Most documents cited in NRC publications will be available from one of the following sources

- The NRC Public Document Room, 1717 H Street, N.W. Washington, DC 20555
- The Superintendent of Documents, U.S. Government Printing Office, Post Office Box 37082, Washington, DC 20013-7082
- 3. The National Technical Information Service, Springfield, VA 22161

Although the listing that follows represents the majority of documents cited in NRC publications, it is not intended to be exhaustive.

Referenced documents available for inspection and copying for a fee from the NRC Public Document Room include NRC correspondence and internal NRC memoranda; NRC Office of Inspection and Enforcement bulletins, circulars, information notices, inspection and investigation notices; Licensee Event Reports; vendor reports and correspondence; Commission papers; and applicant and licensee documents and correspondence.

The following documents in the NUREG series are available for purchase from the GPO Sales Program: formal NRC staff and contractor reports, NRC sponsored conference proceedings, and NRC booklets and brochures. Also available are Regulatory Guides, NRC regulations in the Code of Federal Regulations, and Nuclear Regulatory Commission Issuances.

Documents available from the National Technical Information Service include NUREG series reports and technical reports prepared by other federal agencies and reports prepared by the Atomic Energy Commission, forerunner agency to the Nuclear Regulatory Commission.

Documents available from public and special technical libraries include all open literature items, such as books, journal and periodical articles, and trancactions. *Federal Register* notices, federal and state legislation, and congressional reports can usually be obtained from these libraries.

Documents such as theses, dissertations, foreign reports and translations, and non-NRC conference proceedings are available for purchase from the organization sponsoring the publication cited.

Single copies of NRC draft reports are available free, to the extent of supply, upon written request to the Division of Technical Information and Document Control, U.S. Nuclear Regulatory Commission, Washington, DC 20555.

Copies of industry codes and standards used in a substantive manner in the NRC regulatory process are maintained at the NRC Library, 7920 Norfolk Avenue, Bethesda, Maryland, and are available there for reference use by the public. Codes and standards are usually copyrighted and may be purchased from the originating organization or, if they are American National Standards, from the American National Standards Institute, 1430 Broadway, New York, NY 10018.

NUREG/CR-4374 BNL-NUREG-51917 Vol. 2

A Review of the Oconee-3 Probabilistic Risk Assessment

External Events, Core Damage Frequency

Manuscript Completed: January 1986 Date Published: March 1986

Prepared by N. A. Hanan, D. Ilberg, D. Xue, R. Youngblood, J. W. Reed*, M. McCann*, T. Talwani**, J. Wreathall***, P. D. Kurth***, K. Bandyopadhyay, C. Costantino

Brookhaven National Laboratory Upton, NY 11973

*Jack R. Benjamin & Associates, Inc. **University of South Carolina ***Battelle Columbus Laboratorie

Prepared for Division of Safety Review and Oversight Office of Nuclear Reactor Regulation U.S. Nuclear Regulatory Commission Washington, D.C. 20555 NRC FIN A3797

ABSTRACT

A review of the Oconee-3 Probabilistic Risk Assessment (OPRA) was conducted with the broad objective of evaluating qualitatively and quantitatively (as much as possible) the OPRA assessment of the important sequences that are "externally" generated and lead to core damage. The review included a technical assessment of the assumptions and methods used in the OPRA within its stated objective and with the limited information available. Within this scope, BNL performed a detailed reevaluation of the accident sequences generated by internal floods and earthquakes and a less detailed review (in some cases a scoping review) for the accident sequences generated by fires, tornadoes, external floods, and aircraft impact.

CONTENTS

										Page
ABS LIS LIS ACKI EXE	TRACT TOF TOF NOWLE CUTIV	FIGURES TABLES. DGMENTS E SUMMA	RY							ix x xi xii
1.	INTR	ODUCTIO	N							1-1
	1.1 1.2	Object Organi	ive, Scope zation of	e, and A Report.	pproach	to the	e Review			·····1-1 ····1-1
2.	A RE SOUR	VIEW OF CES WIT	THE OCON HIN PLANT	EE UNIT BUILDIM	3 ANALY	SIS OF	FLOODIN	NG EVENT	TS FROM	2-1
	2.1	Introd 2.1.1 2.1.2	Backgrou Flood-Re	nd lated Co with Flo	ndition	s and	General al	Outline	of	2-1 2-1
	2.2	Turbin 2.2.1 2.2.2 2.2.3	e Buildin Identifi Critical Definiti Estimati 2.2.3.1	g Flood- cation of Floodir on of a on of Fl Approac	Initiat In Impor Ing Areas Set of ood-Ini th to th	Discretiatin	ents lood Sou te Flood g Events d Initia	d-Initia	nd ating Ev requency	2-2 ents2-4 2-6
		2.2.4	2.2.3.2 Consider Some Par	Estimat Categor Broad (ation of ticular	ion ization roups Flood Conside	Isolat	ood-Init	sures wed by l	Events	in 2-9 2-10 2-12 2-12
	2.3	2.3.1	amage Seq The Acci 2.3.1.1 2.3.1.2 2.3.1.3	dent Sec Success The Mai Support	uences criter n Seque	ia ia ences gic and	Fault	Trees		2-12 2-12 2-13 2-14 2-15
		2.3.2	Quantifi 2.3.2.1 2.3.2.2 Analysis	cation of Long-Te The Res Quantif of Add	of the C erm Assu sults of fication itional	Core Da France the C in th Flood	mage Sec of ASW S ore Dama e BNL Re Sequence	quences Suction age Sequences eview	Jences	2-16 2-17 2-18 2-19
			2.3.3.1 2.3.3.2	Other 1 Auxilia	furbine ary Buil	Buildi Iding F	ng Flood looding	d Seque	nces	2-19
	2.4	Summar Refere	y nces							

٧

3.	REVIE	EW OF SI	EISMIC EVENTS
	2 1	Intradi	ation 2.1
	2.1	Introdi Coderd	- Namend
	3.2	Seismi	c Hazard
		3.2.1	Review Approach
		3.2.2	Seismic Hazard Methodology
		3.2.3	Seismotectonic Regions
		3.2.4	Seismicity Parameters
		3.2.5	Ground Motion Characterization
			3.2.5.1 Ground Motion Attenuation
			3.2.5.2 Unner-Round Accelerations 3-14
		3 2 6	Saismic Hazard/Engoility Intonfaco
		2 2 7	Companying with USCC Deculter
		3.2.1	Comparison with 0505 Results
		3.2.8	Sensitivity calculations
		3.2.9	Summary
	3.3	Seismi	c Fragility Analysis
		3.3.1	Introduction
		3.3.2	Plant Background and Frayility Methodology
		3.3.3	General Comments on the Seismic Fragility Analysis3-23
		3.3.4	Review of Significant Components
		3.3.5	Conclusions
	3.4	System	s Analysis
		3.4.1	Introduction
		3 4 2	Seismic Sequences Definition - A Qualitative Peview 3-38
		3 4 3	Paviaw Paculte and Concitivity Analysis 3-30
	3 5	Dafe to	neview Results and Sensitivity Analysis
	3.5	Refere	1005
4.	REVIE	EW OF O	THER EXTERNAL EVENTS4-1
	4.1	Review	of Fire Events
		4.1.1	Summary of OPRA Analysis
		4.1.2	RNL Review
	4 2	Review	of Tornado Events
	4.6	A 2 1	Summary of ODDA Analysis
		1 2 2	DNI Daviau
	4 2	4.6.6	DNL REVIEW
	4.3	Review	of External Flood
		4.3.1	Summary of UPRA Analysis
		4.3.2	BNL Review
	4.4	Review	of Aircraft Impact Events
5	SIIMM/	ARV	5-1
	30/11/2		
APP	ENDIX	A: RE	VIEW OF OPRA FAULT TREE APPROACH TO QUANTIFICATION OF
		TH	F FLOOD-INITIATING EVENTS
APP	ENDTY		PRINE BUILDING FLOODING. QUANTIEICATION OF THE SEQUENCES
ALL.	enorn.	CO	DE DAMAGE BINS AND CONTAINMENT_SAFECHADD STATES
		CO	The UNITAL DIAS, AND CONTAINENT "SAFEGUARD STATES
APP	ENDIX	C . FA	ILT TREES FOR THE POSSIBILITIES OF FLOODING PEACHT
Arr	LUDIX	CP.	ITTAL LEVELS
		UR	

Page

APPENDIX	D:	REVIEW OF SEISMIC GROUND MOTION HAZARD, OCONEE NUCLEAR POWER PLANT SITED-1
APPENDIX	E:	A REVIEW OF THE SEISMIC ANALYSES FOR JOCASSEE DAM AND OCONEE DIKES USED IN OCONEE PRAE-1
APPENDIX	F:	SEISMIC FAULT TREESF-1
APPENDIX	G:	A REVIEW OF THE MODELING OF FIRE PHENOMENA AND THE ESTIMATION OF FIRE FREQUENCIES IN THE OCONEE UNIT 3 PROBABILISTIC RISK ASSESSMENT FOR BROOKHAVEN NATIONAL LABORATORY BY BATTELLE COLUMBUS LABORATORIESG-1

-

4

Page

B

LIST OF FIGURES

Figure	Page
2.1 2.2 2.3	Schematic of the CCW system
	after a flood
3.1	An example of the uncertainty in seismic hazard estimates3-40
3.2	Oconee study seismotectonic model
3.3	U.S. Geological Survey seismic sources near the Oconee site3-48
3.4	Seismic source areas for Expert 3 in LLNL seismic hazard study3-49
3.5	Seismic source areas for Expert 11 in LLNL seismic hazard study3-50
3.5	Seismic source areas suggested by Bollinger
3.7	Comparison of the best estimate ground motion models from the
	LLNL ground motion panel for magnitudes 5 and 7 and the Oconee
	attenuation relationships
3.8	Comparison of the Oconee attenuation models and other models
	provided by the SHCP ground motion experts
3.9	USGS seismic source areas defined in their 1982 study
3.10	Comparison of the 1976 and 1982 USGS studies and the mean Oconee
2.11	seismic nazard curve
3.11	Comparison of cases 1 and 2
3.12	doministion truncation 2 results using a three standard
2 12	Comparison of cases 1 and 2 using a five standard doviation
5.15	comparison of cases 1 and 2 using a five scandard deviation
3 14	Comparison for cases 1 and 2 hunchhosis 1 I = VIII for
5.14	the Piedmont and upper coastal plain source 3-59
3.15	Comparison for cases 1 and 2, hypothesis 2, Image = 1% for
0110	the Piedmont and upper coastal plain source
3.16	Comparison for cases 1 and 2, hypothesis 3, $I_{max} = X$ for the
	Piedmont and upper coastal plain source
3.17	OPRA seismic event tree
3.18a	Sequence and core melt fragility curves
3.18b	Sequence fragility curves
4.1	OPRA cable-shaft fire event tree
4.2	OPRA event tree for sequences initiated by a severe tornado4-8

LIST OF TABLES

Table	Page
0.1 2.1	Summary of Annual Core Damage Frequencies for External Eventsxvi Key Events Occurring at Four Critical Elevations in the Turbine Building (Modified Plant)
2.2	Time to Reach Critical Levels as a Function of Flood Flow Rate
2.3	(in minutes)
2.5	BNL Review
2.5	Unit 3 LPSW or Condensate Cooler System
2.6	BNL Selection of Flooding Categories for the Flood-Isolation Failure Possibilities2-33
2.7	Values for the Failure to Maintain Long-term SSF ASW Suction
2.8	Core Damage Conditional Probabilities of the Various Flood
2.9	Summary of Core Damage Frequencies for Turbine Building CCW
2.10	Floods2-36 BNL Review Summary of Core Damage Frequencies for Floods From
2.11	Sources Within the Plant2-37 BNL Review Core Damage Frequency Distribution for Floods From
2 1	Sources Within the Plant
3.2	LLNL Seismic Hazard Characterization Project Seismicity
3.3	Parameter Estimates
3.4	USGS Seismic Source Parameters for Sources Near the Oconee Site3-68 Component Fragility Parameter Values
3.6	Top Events for the Seismic Event Tree
3./	in OPRA
3.8	Effect of Including Components 38 Through 41
3.10	Effect of Alternative Random Failure and Operator Error
3.11	Bounding Cases: No Credit for Operators Actions Frequencies
3.12	Mean Frequency of Failures (per year)
3.13	Effect of Alternative Block Wall and Condenser Capacities
3.14	Coolant Pump Components From Analysis
3.15	Components Removed From Base Case Plant System Model
4.1	Summary of BCL Review
4.2	Summary of Fire Core Damage Annual Frequency

ACKNOWLEDGMENTS

The authors wish to thank their colleagues in the Department of Nuclear Energy at Brookhaven National Laboratory for many discussions and comments throughout the course of this work.

This work was performed for the Reliability and Risk Assessment Branch (RRAB) of the U.S. Nuclear Regulatory Commission. Mr. E. Chelliah of RRAB was the technical monitor of the project. The authors wish to acknowledge F. Coffman, E. Chelliah, and A. Thadani for comments on the draft of this report.

Lastly, the authors would like to acknowledge the relentless effort and the high standard of quality of C. Conrad in editing and preparing this document for publication, and of S. Flippen for her excellent typing.

EXECUTIVE SUMMARY

This review of the Oconee-3 Probabilistic Risk Assessment (OPRA) by Brookhaven National Laboratory (BNL) was sponsored by the U.S. Nuclear Regulatory Commission (NPC). The OPRA, which was performed by the Nuclear Safety Analysis Center (NSAC), Duke Power Company (DPC), and other participating institutions, includes estimates of the frequency of accidents (internally and externally initiated events) that may lead to core damage, the frequency of release of radionuclides, and the frequency of occurrence of public health effects resulting from the various initiating events. This review presents an assessment of the frequency of core damage due to externally initiated events and other physical phenomena, i.e., internal floods, earthquakes, fires, tornadoes, external floods, and aircraft impact. A companion review of the frequency of core damage due to internal events is reported in Volume 1 of this report.

The broad objective of the BNL review of the external events portion of the OPRA was to evaluate qualitatively and quantitatively (as much as possible) the study's assessment of the important sequences that are "externally" generated and lead to core damage. To carry out this objective, BNL reviewed the assumptions and methods of the OPRA within its stated objective and with the limited information available. Within this scope, BNL performed a detailed reevaluation of the accident sequences generated by internal floods and earthquakes and a less detailed review (in some cases a scoping review) for the accident sequences generated by fires, tornadoes, external floods, and aircraft impact.

The review process included a site visit and a meeting with Duke Power Company and NRC. The DPC staff was helpful and cooperative throughout the course of the review.

Overall, the assessment of core damage frequency due to "external" events presented in the OPRA appears to use state-of-the-art methodologies, and within the stated scope the OPRA is a good piece of work. Table 0.1 presents the results of this review, and the main conclusions follow:

a. Floods from sources within the plant buildings - The OPRA presents a very detailed analysis of the accident sequences due to internal floodings, and BNL also performed a detailed reevaluation of this portion of the OPRA. The main results of this part of the review are the following:

- a.1 The total core damage frequency for floods from sources in the plant calculated in this review is equal to 9.7E-5/yr, i.e.,
 - turbine building CCW floods: 8.0E-5/yr (OPRA = 8.8E-5/yr),

other floods in turbine building: 4.8E-6/yr,

auxiliary building floods: 1.2E-5/yr (OPRA = 1.3E-5/yr).

Note that the OPRA has done some analysis for the other floods in the turbine building and for the auxiliary building floods. However, in their final results, they fail to present the core damage frequency contribution from these sources.

a.2 A scoping assessment of the parameter uncertainties about the frequency of core damage for internal flooding was performed and the resulting distribution is

 $X_{05} = 1.3E-5/yr$ $y_{50} = 5.2E-5/yr$ $y_{95} = 2.8E-4/yr$ $y_{1ean} = 9.1E-5/yr$

b. Seismic - The OPRA presents a detailed analysis of the accident sequences due to seismic events, and this review has performed an equivalent reevaluation of this part of the OPRA. The following comments and results are appropriate.

- b.1 Seismic Hazard Analysis
 - The methodology used to evaluate the frequency of exceedance is adequate and appropriate to characterize the seismic hazard at the Oconee site.
 - Consideration of only one seismotectonic model is considered inappropriate and probably unconservative. The seismotectonic model used in the analysis is reasonable.
 - Estimates of maximum intensity for the Piedmont source area are believed to be underestimated by one intensity unit. Sensitivity calculations indicate that this results in a moderate increase in the siesmic hazard (i.e., a factor of 2 to 10).
 - Comparisons with the Seismic Hazard Characterization Program (SHCP) show that the attenuation models used in the Oconee study provide estimates of the seismic hazard that are lower. Overall, we judge this would have a small effect on the hazard estimates.
 - The estimates of seismicity parameters, activity rates, and b-values are in reasonable agreement with estimates from other studies.
 - In our opinion, the uncertainty in the estimate of the frequency of exceedance is underestimated, leading to an unconservative estimate of the site hazard. If a complete family of seismotectonic models and attenuation relationships is included in the analysis, we estimate there will be a moderate increase in the frequency of exceedance (i.e., a factor of 3 to 10).
- b.2 Fragility Analysis From the review of the fragility analysis and the resulting parameter values, the capacities used in the Oconee PRA are believed to be conservatively low. Generic median values were used for many of the components which resulted in conservative capacities. It was difficult to follow the capacity calculations which contain many inconsistencies and

in general are not well organized. In addition, the highconfidence low-probability values which correspond to the fragility parameter values are in many cases inconsistent with earthquake experience.

The median capacity for the block walls was particularly low. The analysis did not take advantage of the additional capacity provided by arching action, which is currently being verified by testing. This is an important consideration since the block walls are important contributors to the mean frequency of core melt.

b.3 Core Damage Frequency: Event Tree/Fault Tree Analysis - In this review, the fault trees used in the seismic events analysis were modified, and the resulting core damage frequency due to seismic events was equal to 8.2E-5/yr as compared to 6.3E-5/yr in the OPRA; in this review the OPRA seismic hazard and fragilities were used. Note that this review tried to replicate the OPRA analysis and a core damage frequency of 7.5E-5 was obtained (compared to 6.3E-5 in the OPRA); this result indicates a small difference due to the quantification tools used. It is also important to note that because of the complexity of the expressions for the core damage sequences, only a mean value of core damage frequency was estimated in this review, i.e., a distribution was not obtained.

c. Fires - In this review, a limited assessment of the modeling of fire growth and suppression and its effects on initiating events was performed. Several areas were found where a more detailed analysis than that performed in the OPRA would change the core damage frequency; for these areas, upper- and lower-bound factors were used in this review. Also, in the analysis of the fire event tree, this review found an area where a more detailed analysis (with much more information than BNL had access to) would change the results of the fire sequences. On the basis of these facts, this review assessed the core damage frequency from fire events to be between 6.9E-6/yr to 2.2E-4/yr as compared to the 1.0E-5/yr calculated in the OPRA.

Duke Power Company has stated, in a comment to this review, that a sequence added in this report (see Section 4.1.2) is theoretically possible but it was not appropriate or within the bounds of the OPRA scoping fire analysis because of the low probability combination of burned/not burned control and power cables. However, the information provided to BNL was not sufficient for the elimination of such sequences. If these postulated sequence, were to be eliminated, the core damage frequency from fire events would change from between 6.9E-6/yr to 2.2E-4/yr to between 2.5E-6/yr to 8.1E-5/yr.

d. Tornadoes - In this review, only a scoping analysis of tornado events was performed. The review of the tornado-wind accident sequences found a sequence not presented in the OPRA (see Section 4.2) and this difference is responsible for the change in core damage frequency from 1.3E-5/yr to 2.3E-5/yr.

e. External Floods - In this review, no reanalysis was done for the external floods because of the lack of information and the restricted scope of

this review. Therefore, the core damage frequency of 2.5E-5/yr calculated in the OPRA for external flooding was not reassessed at BNL. This core damage frequency comes exclusively from the random failure of the Jocassee Dam and the assumption that a failure of the dam will result in core damage.

f. Aircraft Impact - Because of the negligible contribution to core damage, no effort was spent in the reassessment of the core damage frequency due to aircraft impact.

	OPRA	BNL Review
Turbine-Building Floods		
 CCW Floods Other Floods 	8.8E-5 _a	8.0E-5 4.8E-6
Auxiliary-Building Floods	_a	1.2E-5
Seismic	6.3E-5	8.2E-5
Fires ^b	1.0E-5	<u>6.9E-6</u> 2.2E-4
Tornadoes	1.3E-5	2.3E-5
External Floods	2.5E-5	2.5E-5 ^C
Aircraft Impact	Negligible	Negligible

Table 0.1 Summary of Annual Core Damage Frequencies for External Events

aThe OPRA addresses these floods but does not include them in its final results.

^bBNL presents an upper and lower bound based on the fire phenomenology modeling (see Appendix G). ^cNot reassessed in this review.

1. INTRODUCTION

1.1 Objective, Scope, and Approach to the Review

The Duke Power Company, in collaboration with the Nuclear Safety Analysis Center, has carried out a full-scope PRA of Oconee Unit 3. The Oconee PRA (OPRA) treats "internally" initiated scenarios (accidents initiated by a functional equipment failure or an external loss of offsite power), as well as externally initiated scenarios (i.e., earthquakes, tornadoes, floods, and aircraft impact) and other physical phenomena (fires and internal floods); here, both the externally initiated scenarios and other physical phenomena will be referred to as "external events." Containment analysis was also performed in the OPRA. BNL has conducted a full-scope review of the "internal" and "external event" scenarios defined in the OPRA. However, only a limited review of the containment response and radiological source term analyses in the OPRA has been performed. This report describes the review of the "external events" scenarios out to core damage; the review of the "internally" initiated events is presented in Volume 1 of this report. The review of the containment performance and radiological source terms will be presented in a separate report.

The broad objective of the BNL review of the external events portion of the OPRA was to evaluate qualitatively and quantitatively (as much as possible) the study's assessment of the important sequences that are "externally" generated and lead to core damage. To carry out this objective, BNL reviewed the assumptions and methods of the OPRA within its stated objective and with the limited information available. Within this scope, BNL performed a detailed reevaluation of the accident sequences generated by internal floods and earthquakes and a less detailed review (in some cases a scoping review) for the accident sequences generated by fires, tornadoes, external floods, and aircraft impact.

This review has been conducted principally by the Risk Evaluation Group within the Safety and Risk Evaluation Division at BNL. In addition, selected areas were reviewed by other groups at BNL and by contractors. Jack R. Benjamin and Associates (JBA) reviewed the seismic hazard and fragility analysis; modeling of fire growth and suppression was reviewed by Battelle Columbus Laboratories (BCL); and the Structural Analysis Division at BNL reviewed parts of the Jocassee seismic fragility analysis and other portions of the PRA.

The project monitor was E. Chelliah of the Reliability and Risk Assessment Branch of the U.S. NRC.

The review process was facilitated by several discussions with Duke Power Company and its consultants. BNL, JBA, and BCL reviewers visited the Oconee plant in May 1985.

1.2 Organization of Report

Section 2 presents a review of the flooding events from sources within plant buildings. Section 3 contains a review of seismic events. Section 4 presents the review of fires, tornadoes, external floods, and aircraft impact, and Section 5 presents a summary of the results of this review. 2. A REVIEW OF THE OCONEE UNIT 3 ANALYSIS OF FLOODING EVENTS FROM SOURCES WITHIN PLANT BUILDINGS

2.1 Introduction

2.1.1 Background

The Oconee Probabilistic Risk Assessment¹ (OPRA) has provided a very detailed analysis of flooding events within the turbine building, and a less detailed analysis for flooding within the auxiliary building. The analysis of the flooding events within the turbine building was carried out several times: three iterations were performed to increase the detail and realism of the analysis, and a final, fourth iteration was done to include several plant modifications made by Duke Power Company. The BNL review addresses this final analysis of the "modified" plant. The flooding analysis consisted of the following steps:

- 1. Identification of important flood sources and critical flooding areas. Critical flooding areas are those where a flood could result in a plant trip and could cause failure of equipment that might be needed for core cooling after the plant trip.
- Definition of a set of discrete flood-initiating events for each flood source and critical flooding area, to best represent the spectrum of possible flood rates for the purposes of constructing and guantifying core damage sequences.
- Estimation of the frequency of each flood-initiating event, and their grouping into several flood categories.
- Construction of core damage sequences for each flood-initiating category, and plant systems modeling.
- Quantification of the sequences, core damage bins, and containmentsafeguard states.

In this report, the first three steps for flooding events within the turbine building are discussed in Section 2.2, and the last two steps are presented in Section 2.3; Section 2.3 also presents a brief discussion on the flooding events within the auxiliary building. Section 2.4 presents a summary of the results obtained in this review.

2.1.2 Flood-Related Conditions and General Outline of Systems With Flooding Potential

The potential for flooding at Oconee stems from the fact that the turbine building is at a lower elevation than Keowee Lake which provides the suction to the condenser circulating water (CCW) system.

The Oconee plant is built between the intake and discharge canals of Lake Keowee at a level approximately 25 ft below the lake's level. During normal operation, the CCW pumps are needed only to overcome friction losses in the piping, and it was found that the flow out of a break in the CCW system would continue at nearly the regular rate whether or not the pumps continue to operate.

The turbine building basement at Oconee contains equipment needed both for normal operation and for the response to abnormal conditions. In addition to the equipment in the turbine building, the auxiliary building contains safety equipment at levels below the turbine building basement. The two buildings are connected through doors that are watertight to a level of 6 ft ("modified" plant). Above that level it is considered that water from the turbine building will rapidly spill into the auxiliary building. In a meeting² with Duke Power Company it was stated that another modification is going to be made to seal the interface between the two building to a level of 20 ft. This is considered in neither the OPRA nor the BNL review.

The arrangement of the CCW piping in the turbing building which has the potential to be a source for flooding is shown in Figure 2.1. There are four CCW pumps taking suction from the intake canal. The pumps feed into a 186in.-diam pipe that is embedded in the turbine basement floor, and branches into six 78-in.-diam pipes from the floor in the three sections of the main condenser. A similar set of six pipe sections are on the discharge side of the condenser. They drop down back into the floor and join each other below the floor level into two 132-in.-diam pipes. These two pipes rise back to the Lake Keowee level and discharge into the outlet canal on the opposite side of the plant. This arrangement is repeated for each of the three Oconee units (i.e., there are eighteen 78-in. inlet and outlet pipes from the condensers).

In addition to the CCW, other flooding sources are present in the turbine building basement:

- (a) Condensate Coolers Systems
- (b) Recirculating cooling water (RCW) system
- (c) Low pressure service water (LPSW) system
- (d) High pressure service water (HPSW) system
- (e) Unwatering system.

System (a) and parts of systems (b) and (e) are also seen in Figure 2.1. For more details on the systems' description, the OPRA should be consulted.

2.2. Turbine Building Flood-Initiating Events

The following subsection covers the review of the first three steps listed in Section 2.1.1. In support of these three steps, OPRA has conducted a detailed review of the actual flooding events in the data base for the U.S. nuclear plants. It found approximately 60 events (up to 1981) with only 4 (3 in turbine building) resulting in flow rates in excess of 12,000 gpm. The operating experience was not used directly* in quantifying the frequencies of the flood-initiating events. The frequencies are derived from piping and valve breaks considering all the relevant piping located in the turbine building. They are also derived from maintenance and incorrect assembly of systems after maintenance. Thus, much more detailed plant specific analysis is conducted in the derivation of the flood iniciating event frequencies (see discussions in Section 2.2.3). The results of the analysis of flooding experience are used for background information mainly in postulating failure modes for the identification and selection of important flood sources. Several examples of the use of the operating experience in introducing failure modes into the flooding analysis are:

- a. Valve closed inadvertently and water hammer causing the rupture of an expansion joint.**
- b. Expansion joints failure causing extensive leakage.
- c. Pump casing rupture due to pump startup while it was apparently rotating backward (a TMI event).
- d. Air-operated isolation valve opens on air interruption or due to operator error while a system is open for maintenance.
- e. Oconee flood that occurred in 1976 caused by a pneumatic isolation valve failing open after an inverter failure, while the CCW water boxes' manways were open. This flood was recovered after 45 minutes by fixing the inverter and reclosing the valve. It reached a level of two feet and caused failure of equipment associated with the main and emergency feedwater systems. Modifications were done to prevent a recurrence of this event (see OPRA, page 9-138).

2.2.1 Identification of Important Flood Sources and Critical Flooding Areas

Identification of the flood-initiating events was guided by two criteria: (1) a flood event must cause a trip or transient, and (2) the specific flood sources should be able to fail equipment needed for core cooling. The first criterion is satisfied by any flood of a few thousand gpm because it will actuate the 0.5-ft-level flooding alarms that require the operator to manually trip the unit. The second criterion is more demanding and can be met by floods having a flow rate of 12,000 gpm or more (see Section 2.2.2). Thus, the OPRA searched for all potential flooding sources having flow rates greater

- *An early iteration of the Oconee flooding analysis used the operating experience to derive the Oconee flooding frequency. The approach is described in OPRA in general terms to provide another verification for the reasonableness of the results of the fault tree approach used in the later two iterations of the study. The value for the flooding frequency from the two different iterations of the study compare well. The fault tree approach is more detailed and refers more specifically to specific equipment of the Oconee plant which can be a flooding source, so that it helps to identify major contributors.
- **Note that a recent event of flooding due to expansion joint rupture following a water hammer occurred at La Crosse plant (1985); the Quad Cities (1972) was the only event of this kind known previously.

than 12,000 gpm. The CCW piping in the turbine building basement and all systems connected to it such as RCW, Condensate Coolers, LPSW, and HPSW were reviewed to determine the maximum potential flow rates that could result from a break in their integrity.

The critical equipment and its height above the turbine building basement floor were determined by a review of drawings and a walk through, and this resulted in the definition of three characteristic critical levels as shown in Table 2.1. This table includes all equipment considered in the response to transients or LOCAs at Oconee that can be affected by flooding. These levels were verified during a plant walk through by the BNL reviewers.

The turbine building drain is a large 6-ft² hole located in the building wall leading through separate piping to the tail-race which is 100 feet lower than the turbine building floor. This allows for a significant discharge of flood water out of the building, depending on the flood level above the floor. The possibility of reaching a critical level is calculated in Appendix L of the OPRA report, ¹ taking into account both the break discharge flow rates and the drainage rate. BNL accepted the Appendix L results shown in Figure L.7 of the OPRA and made two linear approximations to the figure which enabled the calculation of the time required to reach any level of flood for any flood flow rate. The summary of these calculations is given in Table 2.2. It is equivalent to the flow rates given in Figure L.7 of the OPRA Appendix L, but provides a more convenient tool for calculating operator response times and for the grouping of flood-initiating events into flood-initiating categories.

2.2.2 Definition of a Set of Discrete Flood-Initiating Events

The flood sources listed in Section 2.1.2 were reviewed in detail, and the following information was collected for each system which is a potential flooding source:

- a. Pipe sections and length, valves including their size and type, number of welds, and expansion joints.
- b. The elevation in the turbine building of each of the above.
- c. Maintenance frequency and isolation procedure for condenser, condensate coolers, valves, expansion joints, and LPSW.

In addition to the above data necessary for the calculations of the flood rates, the potential for isolation in each case was also considered:

- 1. Isolation of the pump discharge valves in the Keowee Lake intake.
- 2. Isolation of the pneumatic outlet valves.
- 3. Any single valve in the above failing to close.

Table 9.31 of OPRA is a summary of the flood-initiating events considered and their flow rates. However, the isolation potential given in that table is for the unmodified plant. The corresponding BNL review table is given as Table 2.3; it is based on the CCW piping characteristics given in Table 9.33 of the OPRA, which was used without any change. Its review was beyond the scope or information available to BNL.

In general, the OPRA calculations for the unmodified plant were found to be correct. Some comments acquired in the recalculation by BNL are:

- a. Agreement with OPRA break frequency for the inlet and outlet manual butterfly valves was obtained by BNL only when they were assumed to have one half the break area stated in the OPRA, i.e., $\pi R^2 H = \pi (78)^2 + 14/144$ ft².
- b. Three manways on the floor on the inlet side were not included in OPRA. They have a lower frequency of use. BNL included them and considered their lower frequency of use.
- c. The flow rate of the unisolable manways on the outlet is too low (23,000 gpm in OPRA vs 56,000 in BNL--apparently the wrong elevation was considered).
- d. The CCW crossover valves' diameter is 42 inches in most of the pipe sections above the floor.
- e. Page 9-153 case (3): friction losses (k) should be 3.5 instead of 1.4.
- f. The use of $(1 + K_{ex})$ in the equation for calculating the break flow velocity is suspected to double count exit losses. However, not all the friction losses in pipe contractions are included, which have a compensating effect.

BNL has also calculated a flood rate from the rupture of the expansion joints which is higher than given in OPRA Table 9.31 (120,000 gpm* instead of 76,000 gpm). In a meeting with Duke Power, it was stated that the expansion joint rupture flow rates, because of their importance, were calculated separately by a more elaborate piping analysis computer code; the results of this analysis were shown to BNL reviewers and NRC staff present. Thus, BNL considered that the OPRA results are more realistic than the approximate equations used by BNL to perform its review. The expansion joint flow rates are therefore used as they appear in the OPRA.

The BNL recalculations of the flow rates included the modifications considered in the "modified plant." These are:

- a. CCW crossover values 3CCW-40 and 3CCW-42 are normally closed, isolating the CCW flow between the units (i.e., each unit has its own CCW flow to and from the Lake).
- b. Isolation of one train of the condensate coolers in all three units. On reactor trip, the unit 1 control valve on the unisolated condensate coolers closes. Units 2 and 3 maintain back flow through one of their condensate coolers after reactor trip.

*Oconee FSAR also calculated a 115,000-gpm flow rate.

- c. Following the 6-in.-level flood alarms, the operator manually trips the CCW pumps which, on their closure, close the intake valves at CCW pump discharge.
- d. CCW pump trip causes the pneumatic discharge valves to close. Relay CWSRRVN failure will fail the closure of all six pneumatic valves.

2.2.3 Estimation of Flood-Initiating Events

2.2.3.1 Approach to the Flood Initiators Frequency Estimation

The OPRA estimates the frequency of the flooding-initiating events on the basis of pipes, valves, and expansion joints rupture rates and from evaluation of potential flooding during maintenance, as the result of either human error or isolation equipment malfunction.

a. Pipe Rupture

Flood frequencies due to pipe rupture were estimated by the methodology proposed by Thomas.⁴ It requires detailed information on the length of pipe sections and the number of welds in each section. The methodology yields an estimate of the frequency of catastrophic ruptures. OPRA assumed that the rupture sizes will have frequencies distributed according to their size so that a small-size rupture will occur more frequently than the maximum-size double-ended guillotine break. The distribution assumed is:

Probability of a maximum-size dcuble-ended guillotine break: 0.1

Probability of a large rupture: 0.3

Probability of a medium rupture: 0.6.

The above approach yields higher pipe-break frequencies than could be obtained from the use of the mean rupture rates given by the RSS⁵ for pipes larger than 3 inches in diameter. The pipe rupture rates used by Thomas are based on an appraisal of the data in References 5 to 8.

The reviewers do agree that, overall, the Thomas³ methodology as modified by OPRA to include the break-size-frequency distribution represents a realistic model. While the rupture rates derived by Thomas seem to be on the high side, they are used in the OPRA for the piping of the secondary system which can be anticipated to have rupture rates somewhat higher than those of the primary system.

b. Valve Rupture

Hubble and Mille⁹ (report on valve failure rates) was used by the OPRA to obtain the following failure rates for external rupture:

AOVs -- 2.0 x $10^{-7}/hr$ Manual valves -- 1.3 x $10^{-8}/hr$ MOVs -- 1.7 x $10^{-7}/hr$ Check valves -- 5.2 x $10^{-8}/hr$. The above failure rates are in fact valve leakage failure rates, because most LERs from which they are derived are events of very small leakages, many just in excess of the technical specification limits. OPRA refers to 18 LERs given in Reference 9 of which only one could be regarded as a rupture. BNL in a past review¹⁰ had difficulties* in reproducing this value from the LERs reported in Reference 9; however, this review used the same 1/18 factor. Note that this 1/18 factor is about the same as the factor of 6% derived by Thomas⁴ for the ratio of leakage to rupture in piping.

The above failure rates were then modified by OPRA using a distribution of 10%, 30%, and 60% for the maximum, large, and medium ruptures, respectively, in the same way it was used for the pipe rupture.

BNL compared these results with the RSS⁵ mean value of 3 x 10⁻⁸/hr for valve rupture used equally for all kinds of valves. It is the reviewers' opinion that the 1/18 factor could also be used with the RSS⁵ valve failure rate, and in an earlier study,¹⁰ BNL used it for valve rupture failure rates. However, when the distribution of valve break size and frequency is also applied, the probability of catastrophic rupture of a manual valve in OPRA¹ becomes smaller by a factor of 10 compared to the value derived if the RSS⁵ rupture data are used. The LER data⁹ for valves indicate that manual valves have a lower external leakage frequency than AOVs or MOVs by a factor of more than 10. This does not necessarily imply that the external valve rupture frequency of these valves is smaller by such a large factor. Similarly, it could also be argued that the rupture frequency of the AOVs used in the OPRA is on the high side.

BNL, nevertheless, used the same rupture rates as those used in the OPRA, because the effect on Oconee core damage frequency would not be large if higher values for manual valves rupture were used. The above discussion is intended to point out that the contribution from manual valves may have been underestimated, but more work is needed to confirm such a conclusion.

c. Expansion-Joint Rupture

OPRA has performed a special study on the rupture rate of expansion joints. Two failure modes were considered in OPRA:

- Random failure rates. These were derived from the NPE¹¹ data base (0.01 to 0.02 per plant-year); this data base is known to include small leaks as well as other failure modes. Also, data from Duke Power fossil plants experience were considered.
- A water hammer causing the expansion joint to rupture. A valve transfer closed failure mode was considered to cause the water hammer that has the potential to rupture a close-by expansion joint.

^{*}There are 130 LERs in Reference 9 under the title of "External Leakage/Rupture": however, none is an external valve rupture of significance. More time than was available for the BNL review is required to evaluate the 1/18 factor in detail.

The expansion joint random failure rate was evaluated to be 2.5×10^{-4} per expansion joint per year. It was derived by using the NPE¹¹ data as a prior, and the Oconee evidence of one event in over 3,000 expansion joint years was used to obtain the above posterior.

For the expansion joint random failure, the same distribution of 10%, 30%, and 60% used in pipes and valves was not used, because sufficient data to support such a distribution were not available to OPRA.

The frequency of water hammers on the outlet pipe given a pneumatic valve transferring closed was calculated as

 $(18 \text{ valves}) \times (8.9 \times 10^{-8} \text{ hr}^{-1}) * 8760 \text{ hr} = 4.2 \times 10^{-3} \text{ yr}^{-1}$

and 30% of these cases were judged to cause a severe rupture of the close-by expansion joint. In the last 10 years two cases of expansion joints rupture by water hammer were reported (Quad Cities and LaCrosse). The above value is consistent with this experience.

This OPRA evaluation results in a frequency of 0.01 per year for a maximum rupture of an expansion joint in one of the condenser inlet or outlet pipes. These failures account for one third of the total flooding frequency calculated for Oconee. BNL has accepted this approach and considers it to be reasonable and consistent with the data available.

d. Maintenance Events

Several maintenance acts are considered in OPRA:

- Water boxes removed for retubing (1/40 years) and either operator reinstallation error or isolation valve opening inadvertently.
- Valves removed for maintenance and either valve improperly installed or isolation valve inadvertently opens.
- Manways removed during refueling or shutdown, and manways are improperly reinstalled (there are about 100 manways in the turbine building CCW system).
- Expansion joint is out for replacement (1.8 per year) and valves open inadvertently.
- 5. Condensate coolers opened during shutdown and improperly reinstalled.
- 6. Uther similar equipment maintenance.

The cases of improper reinstallation were judged to have three types of errors of different size and probability. It was assumed that 10% of the installation errors would be maximal, 30% large, and 60% small. The basic probability of installation error was taken as 3×10^{-3} , the same as for procedural error of omission.

BNL accepted the above approach for maintenance-induced floods. OPRA has included another factor of 0.1 for recovery of maintenance-induced floods

starting on the inlet side of the condenser. BNL did not find a significant reason for treating the inlet and outlet sides of the condenser in a different way (by a factor of 10) and used the same value for improper installation errors with no recovery for both. This has increased the contribution from inlet side floods, while the outlet side floods contribution from maintenance remained the same as in the OPRA analysis.

It can be concluded from the above discussion that BNL agrees in most cases with the frequencies calculated for the individual flood-initiating events in OPRA. The flooding-initiator frequencies for OPRA and for this review are given in Table 2.4.

2.2.3.2 Categorization of Flood-Initiating Events in Broad Groups

The OPRA has used three broad groups for the flood analysis; in two of these groups it distinguishes between breaks occurring on the intake side of the condenser and those occurring on the outlet side. In addition it distinguishes between isolable and nonisolable breaks. Thus the resulting grouping in the OPRA becomes:

- FVLI: Very large flood and isolable. Flood rate ~160,000 to 450,000 gpm.
- FVLN: Very large flood and nonisolable. Flood rate ~160,000 to 450,000 gpm.
- FLII: Large flood and isolable. Inlet side flood rate ~60,000 to 160,000 gpm.
- FLIO: Large flood and isolable. Outlet side flood rate ~60,000 to 160,000 gpm.
- FLN: Large flood and nonisolable. Flow rate between 60,000 to 160,000 gpm.
- FMII: Medium flood and isolable. Inlet side flow rate 12,000 to 60,000 gpm.
- FMIO: Medium flood and isolable. Outlet side flow rate 12,000 to 60,000 gpm.
- FMN: Medium flood and nonisolable. Flow rate 12,000 to 60,000 gpm.

BNL made a more detailed grouping than the above for the following reasons:

- As seen from Table 2.3 for the column of "particular valve to break fails," different groups of flooding will result for an inlet and an outlet for very large floods. Therefore, BNL also distinguished between the inlet and outlet for the very large floods.
- 2. The FVLI group covers a large range of flood rates for which different operator response times are anticipated, as can be seen from Table 2.2 (e.g., 2-ft level reached in 11 min for 170,000-gpm flood and in 4 min for 450,000-gpm flood). Therefore, BNL made a breakdown of the very large flood category into two subcategories: (1) very large No. 1 (350,000 to 450,000 gpm) and (2) very large flood No. 2 (170,000 to 349,000 gpm).
- 3. Similarly, the large flood group was divided into two: (1) large flood No. 1 (120,000 to 169,000 gpm) and (2) large flood No. 2

(60,000 to 119,000 gpm). The main reason for this breakdown was that a large flood No. 1 has the potential for reaching the 6-ft level in one to two hours, whereas the large flood No. 2 cannot reach this critical level. OPRA did not consider this division, primarily because the dominant flood-initiating event in their large flood category is the failure of the expansion joint which has a flow rate of about 75,000 gpm. However, some other flood-initiating events are included and give rise to some additional contributions to the core damage frequency, which were later found to be guite small.

The summary of the OPRA and BNL grouping is shown in Table 2.4. The values for OPRA are obtained from fault trees (OPRA Figures 9.63 to 9.70), The BNL frequencies for each group are evaluated in Appendix A which provides the list of initiators, their flow rates, and the frequency of each group.

In deriving the frequencies of OPRA and BNL, additional considerations were also taken into account. The OPRA has divided the pipe and valve ruptures and improper installation errors into maximum, large, and small and assigned different frequencies to them as discussed before. OPRA defined large rupture or installation error as having a break size that leads to a flood one category less severe than the one obtained for the maximum size and defined small (or medium) rupture as two categories less severe. As a consequence of this breakdown, both a 78-in. pipe rupture (450,000 gpm maximum flow rate) and a 78-in. butterfly valve rupture (175,000 gpm maximum flow rate) were transferred to the same group of large floods when the 30% case was considered, and similarly to the medium flood for the 60% case. However, the large flood from a 78-in. pipe rupture can still reach the 6-ft critical-level and the 78-in. butterfly valve large rupture cannot. To allow for this different effect, this review considered the flow rate reduction factors proportional to the break size area; a value of 0.3 as a reduction factor from the maximum flood rate to the large and from the large flood rate to the small. Appendix A of this report shows the effect of this different breakdown. Overall, as will be seen later, these changes did not have a large effect on the resulting core damage frequency because the longer time available to the operator to take action in the case of the floods with lower flow rates compensates for the increase due to the transfer to a higher flow rate category.

2.2.4 Consideration of Flood Isolation Measures

The OPRA flood study considered that the operator took prompt action to isolate the flood. This is performed by a manual action to trip all CCW pumps. The flooding emergency procedure ¹² requires the operator to take this action immediately after receiving the flooding alarm from the 0.5-ft flooding level sensors. Upon the pumps' trip, all the pump discharge valves (intake valves) and the pneumatic condenser outlet valves are automatically shut in about three minutes. Following successful isolation of all the valves, only a flood rate of about 20,000 gpm or less can continue, because of the designed backflow through the condensate coolers. This backflow exists only in Oconee units 2 and 3. In Oconee unit 1, only a small backflow rate of less than 12,000 gpm continues from the RCW lines backflow, because the condensate coolers are automatically isolated when the unit is tripped. This backflow is provided to supply suction to the LPSW and HPSW pumps in unit 2 and/or unit 3. A LPSW pump and a HPSW pump when operating will use about one half the

20,000 gpm flow rate. Thus, if the break causing the flood is in the outlet (isolable) or inlet piping, the rate of discharge through the break would be smaller than 12,000 gpm if a LPSW pump (up to 15,000 gpm) is in operation. However, if the break occurs in the condensate coolers path, the CCW crossover, or the LPSW system, then LPSW suction would be lost and the flood rate would continue at about 20,000 gpm, after isolation of the intake and outlet valves.

The above considerations were modeled in both the OPRA and the BNL review studies. In addition, several valve and relay failures, or delayed operator action to trip the CCW pumps can fail the isolation of the flood to the flow rate corresponding to the smaller backflow rate. The failures considered in OPRA are:

- a. Failure of any of the four intake valves to close (CCWMVC).
- b. Failure of any of the six pneumatic outlet valves to close (CCWAVC).
- c. Failure of the particular pneumatic outlet valve leading to the break (CCWAVIC in OPRA and CC20AVC in BNL).
- d. Failure of the operator to trip the CCW pumps in time (e.g., CCWI15H).

BNL used the same isolation failure considerations as above but distinguished two cases for the failure of the outlet valves:

- (b1) Failure of any of the six pneumatic outlet valves to close (CCWAVC).
- (b2) Failure of all pneumatic outlet valves because of a common relay failure (CWSRRVN).

For case (b2) the break flow rates are higher than for a (b1), as can be seen in Table 2.3.

The effects of each of the above isolation measure failures are given in Tables 2.5 and 2.6. Table 2.5 shows the fraction of the flood frequencies that affect the LPSW suction and therefore cause direct failure of this system. Note that when the break occurs in the condensate coolers (rather than in the LPSW system itself), a recovery action to supply LPSW to unit 3 from the other unaffected unit is possible. Table 2.6 summarizes the categorization of the flood rates following the failure of any of the isolation measures. It is seen that even in the case of partial isolation failure the flood rate can be smaller. Any case of transfer to another category of flood rate has a reduced potential for damaging safety systems.

Note that floods denoted as "unisolable" have their flood rate continuing until additional isolable actions are taken beyond those considered in this section.

The data used to quantify an isolation measure failure are taken from the OPRA Table 9-40, which is consistent with the basic data used in the OPRA.

2.2.5 Some Particular Considerations Reviewed by BNL

During the BNL review of the OPRA flooding study, several elements were reviewed in particular:

a. OPRA refers in their data analysis (see Table 5.3, page 5.20) to the failure rates of MOVs. For Oconee, the value of 6.4 x 10⁻³/d is used for all MOVs, except for the condenser circulating water (CCW) system; in this case (CCW system) a value of 0.1/d is given. If the latter were to be used in the flooding study, the intake isolation failure probability would increase significantly and a higher frequency of core damage would result. This seeming discrepancy was discussed with Duke Power² and it was shown that the 0.1/d is based on the MOVs of the CCW emergency discharge lines which do not affect the flooding study. Duke Power stated that the experience with the CCW intake valves does not significantly deviate from the general MOV experience at Oconee.

Therefore, the 6.4 x $10^{-3}/d$ is appropriate and was used by both the OPRA and this review.

b. Appendix A.3 of OPRA discusses the HVAC system. It states that the LPI motors require HVAC room cooling to prevent motor overheating. This has some effect (-5×10^{-6}) on core damage frequency for sequence No. 6 (see next section). This dependence was not considered in the OPRA flooding study. In a meeting with Duke Power,² it was clarified that calculations by Westinghouse for DPC show that without room cooling from the HVAC system it would take several days for overheating of the LPI pumps. BNL did not receive this calculation, but accepted this assumption.

2.3. Core Damage Sequences and System Modeling

2.3.1 The Accident Sequences

The accident sequences were developed in a manner similar to that in the internal events analysis.¹ A functional event tree (see Figure 2.2) with supporting logic (top level fault trees) for each function appearing on the functional event tree, and some simplified system fault trees were constructed for the turbine-building flooding analysis. The fault trees provide the major failure modes of the equipment used to mitigate the progress of the core damage sequences. They include separate subtrees for the possible failure of the system because the flood will exceed a certain critical level. The system fault trees include a transfer-in from these subtrees whenever a certain critical level can fail the equipment. For example, the failure of the service water system includes the event "flood exceeds critical level 2 (4 ft)," and this event is developed as a top event of a subtree to which all flood-initiating categories are input. In addition, the possible failures of rlood-isolation measures or operator failing to initiate the isolation measures in time, are also included.

Eight event tree sequences initiated by a turbine building flooding are considered in the OPRA and are shown in Figure 2.2 (OPRA Figure 9-77); this review did agree with the event tree used in the OPRA.

2.3.1.1 Success Criteria

The success criteria used in defining the supporting logic for each function on the event tree (e.g., Qs, B, X, etc.) are similar to those used for the internal event analysis. However, the SSF system, consisting of the ASW and the makeup subsystems, was explicitly modeled into the supporting logic of the flooding event tree; it is used only as a recovery action in the internal event analysis.

The important success criteria used in the OPRA flooding analysis are:

- 1. RCP Seal Protection:
 - Either one HPI pump or SSF makeup required.
 - Component Cooling is assumed to be unavailable because of containment isolation following the Emergency Safeguard (ES) signal. The flooding emergency procedures direct the operators to actuate the ES channels 1 and 2 immediately after the flooding alarm.
 - If the RCPs are not tripped (another procedural action), seal failure will occur in one minute if not protected by seal injection. Otherwise, seal leakage will occur in about 30 minutes if seal injection fails.
- 2. RCP Pressure Relief:
 - In the case of a very large flood (represented in the OPRA by a flow rate equal to 300,000 gpm), it is assumed in the OPRA that because of the fast loss of MFW and EFW systems, the pressurizer PORV does not have sufficient relief capacity in about 80% of the cases. Therefore, the opening of one of the two SRVs is required. In all other cases the opening of PORV is sufficient to avoid a challenge to the SRVs.
 - Since the BNL review subdivides the large floods into two groups, i.e., very large flood 1 (flow rate >350,000 gpm) and very large flood 2 (flow rate between 170,000 and 349,000 gpm), the following success criteria were used:
 - one out of two SRVs for the "very large flood 1," and for 50% of the cases of "very large flood 2," and
 - . PORV opening sufficient for all other cases.
- 3. RCS Integrity Loss:
 - Any stuck open SRV, or
 - seal failure.
- 4. RCS Inventory makeup:
 - One HPI pump or the SSF makeup pump.

- 5. RCS Heat Removal by Steam Generators:
 - One out of three EFW pumps or feedwater from the SSF Auxiliary Service Water (ASW) system. Note that the MFW is lost for all floods, and the EFW is also lost in practically all floodinginitiating events considered. This fact implies that the SSF ASW is a must (its actuation is a procedural action).
- 6. High Pressure Injection (HPI):
 - One HPI pump is sufficient for core cooling injection, or feed and bleed.
- 7. RCS Makeup Supply Maintained:
 - BWST must be available. Two cases are considered:
 - If LPSW is available, then RBSS is not required and BWST suffices for 12 hours.
 - If LPSW is unavailable, then RBSS is required by procedure and the BWST would be depleted in two hours.
- 8. Long-Term Cooling:
 - SSF ASW and RCS inventory makeup when RCS integrity is maintained.
 - High pressure recirculation (HPR), and either LPSW or SSF ASW when RCS integrity is breached or when in the feed and bleed mode.

To summarize, this review has accepted the success criteria given above, and the only modification was in the criterion (2); this modification is also given above.

2.3.1.2 The Main Sequences

The eight main sequences are shown in Figure 2.2. They start with one of the flood initiators. The first function considered is the seal injection. As the component cooling is assumed lost immediately following operator action to actuate the ESF, this function depends on whether HPI or SSF makeup is available. Thus sequence No. 8 is a seal LOCA sequence.

The next function affecting the progress of the accident is the RCS integrity. Thus, sequences 5, 6, and 7 are small-LOCA (SRV stuck open) sequences. If injection is not available, sequence 7 is obtained. If injection is successful but recirculation fails because the BWST is depleted by RBSS or other failures, sequences 5 and 6 are obtained.

If RCS integrity is maintained, the availability of steam generator (SG) cooling determines whether makeup or feed and bleed is required. If SG cooling is available, the RCS inventory makeup function is sufficient (sequence No. 1). If SG cooling fails, then feed and bleed is required. Failure of feed and bleed leads to sequence 4. Sequences 2 and 3 are the failure of long-term cooling, distinguishing between the early and late needs for HPR, which depends on the availability of LPSW for the RBCS (failure of LPSW will require the RBSS and as a consequence the BWST is depleted faster). Again, in sequence 3, there are two contributions in the X36 functions: (1) early failure of ASW and (2) late failure of ASW. Even though both are called in the OPRA "long-term cooling," the first contributes to bin III and the latter (failure after approximately 12 hours or more) contributes to bin IV.

The "core-damage bins," referred to above, are the same as those used in the internal events analysis and a short description follows:

- Bin I: Early core damage for sequences with loss of RCS integrity (LOCA).
- Bin II: Late core damage for the above sequences.
- Bin III: Early core damage for sequences in which RCS integrity is maintained.
- Bin IV: Late core damage for the above sequences.

BNL did not change the main accident sequences. However, a split between the short-term (-2 hours) and late (-12 hours) components of X36 and X57 (long-term cooling) is more realistic; this split was used in the BNL review. As a consequence of this split, sequence No. 6 contributes to bins I and II, and sequence No. 3 contributes to bins III and IV; the contributions to bins I and III are dominant, as presented in Appendix B. Note that the eight sequences in Figure 2.2 were used in the BNL quantification for each sequence. After the minimal cut sets in each segment were obtained with the SETS code, the splits of the sequences 3 and 6 were performed.

2.3.1.3 Supporting Logic and Fault Trees

For each function of the flooding event tree a supporting logic was prepared in the OPRA and reviewed by BNL. The following comments pertain to the supporting logic given in Figures 9-78 to 9-83 in OPRA.

- a. QS1: The failure probability of the operator in tripping the RCPs following a flood (HPRCPH) was made dependent on the flood category in the BNL review. For very large floods, the time available to trip the RCP is short (15 minutes) and a higher probability of failure was used.
- b. QV1: The event "PORV insufficient" (RC66RVOF) and the operator action to open the block valve (RC4MVH) were made dependent on the flood category, according to the time available until the EFW is lost and the SRVs are challenged as follows:

VFL1 -- RC66RV0F1 = 0.8 and RC4MVH1 = 0.5 VFL2 -- RC66RV0F2 = 0.5 and RC4MVH2 = 0.3 FL -- RC66RV0F3 = 0 and RC4MVH3 = 0.2 FM -- RC66RV0F3 = 0 and RC4MVH4 = 0.1

c. B1: No modification was made in this review.

d. U1: No modification was made in this review.

- e. Y: This function can be shown to correspond to the event of a loss of the LPSW system given by the top event SW01.
- f. X: The intermediate events X11, X34, X36, X54, and X57 are directly used from the supporting logics. No change was made to the subtrees, but the events corresponding to lowering Lake Keowee and recovering the long-term cooling to SSF were explicitly added to the subtrees. In addition, the term HPI in X11 means "Interruption of the RCS makeup due to the failure of the HPI system."

In the system fault trees the following changes were made by BNL:

- a. HP1 -- Failure of the HPSW elevated tank was added into a new AND gate with gate HP3A. Event SWHPSWF was changed to reflect HPSW hardware failure only.
- b. EF1 -- No change was made.
- c. SWO1 -- No changes, apart from FLII which was changed to FL2II, the BNL corresponding flood category.

The OPRA used fault trees to model the various possibilities for a flood to reach the critical flood levels; these fault trees are given in Figures 9.91 and 9.92 of the OPRA. In this review, the same method was used; however, BNL added more flood categories and used a more detailed model to account for the various flood rates in cases of partial isolation. This modification, which is responsible for most of the differences between the BNL fault trees (Appendix C) and those of OPRA Figures 9.9.1 and 9.9.2, was performed by using the matrix given in Table 2.6 and the various flood rates given in Table 2.3. An example of the use of Table 2.6 is given here: if a large flood 1 occurs in the inlet of the condenser hotwell (FL111) and any one of the intake valves fails to close (CCWMVC), this flood will have a flow rate equivalent to a large flood 2 (FL2); if the failure is in the outlet valves (CCWAVC), the resulting flood rate will be equivalent to a medium flood (FM).

Note that the trees do not include specifically the contribution of very large and large floods to the medium floods, after successful flood isolation. This is based on the assumption that only part of the backflow through the condensate coolers will be discharged through the break, because credit was given to continued LPSW and HPSW operation, which leads to the consumption of a part of the backflow rate before the rest is discharged through the break. Even if sequences including this "successful isolation" were included, assuming failure of the LPSW, the effect on core damage frequency would not be large (<10% of total CDF).

2.3.2 Quantification of the Core Damage Sequences

The quantification of the sequences, as discussed before, was performed with the SETS¹³ code. The results are presented in Appendix B where the most important sequences in each core damage bin are described.

An additional event which appears in several sequences, the failure to maintain long-term suction to the SSF ASW, will be discussed next, before the presentation of the summary of the results.

2.3.2.1 Long-Term Assurance of ASW Suction

The sequences contributing to bins II and IV include a term for failure to maintain a long-term suction for the SSF ASW. The OPRA has analyzed this event separately, using the event tree shown in Figure 2.3. Table 9.41 of the OPRA provides the data used in their quantification. BNL reviewed this analysis and considers it to be a good representation of the main actions necessary to isolate the flood in the long term, and also to maintain long-term suction to the SSF ASW. A discussion of the top events in Figure 2.3 follows:

- a. Operators will try to locate the flood early in the sequence. This, in fact, is called by the flooding emergency procedure.¹² The values used for the quantification of the probability of successfully locating a flood are reasonable for the time frame considered (more than 12 hours). The probabilities used in this review, which are basically the same as those used in the OPRA, are given in Table 2.7.
- b. In general, the same chance exists for having a flood in any of the three units. However, ASW takes suction from unit 2. Thus, one can isolate the backflow through the condensate coolers in unit 3 (in unit 1, it is isolated automatically after a trip), and obtain the needed LPSW and ASW suctions from the unit 2 crossover pipe. F13 separates out the fraction of floodings occurring in unit 2 from that of units 1 and 3, in order to account for the additional flood isolation possibilities in units 1 and 3.
- c. The unisolable (N type) floods cannot be isolated in any of the three units; this is reflected in Table 2.7 where a value of 1.0 is used for all floods except FMN. The FMN flood includes some smaller floods with flow rates, which apparently OPRA considered to be isolable without need to lower the lake, and used a value of 0.8 in this case.

It is assumed in OPRA that isolation of breaks in the outlet side of the condenser (type IO) can be made without interrupting flow to the crossover pipe where ASW and LPSW take suction. The manual condenser inlet valves are assumed by BNL to be a possible isolation measure. However, if the flood is on the inlet side of the condenser, the inlet isolation valve may be affected, or access to it would be difficult. This may explain the value of 1.0 used by OPRA in this case. Thus, isolable floods (inlet or outlet sides) in units 1 and 3 can be isolated by manually terminating the backflow; in unit 2 the outlet side floods can be isolated without need to lower the lake.

- d. For CM12, defined as core damage in unit 1 or 2 affecting operator actions in unit 3, BNL calculated its own fractional conditional probabilities as shown in Table 2.8. The main differences in the CM12 probability calculated in this review (Table 2.7) and those in the OPRA (Table 9.41) are:
 - FVLN: The value of 0.18 in the OPRA is much higher than the BNL values (0.09 and 0.07).

- FVLI: The value used in the OPRA (2.2E-4) is much lower than the ones calculated in this review (8.0E-3, 6.0E-3).
- FMN: The BNL value is smaller. In the review, only early core damage is considered for CM12, because the late core comes mostly from lowering the lake.
- e. For FCI/CM12 and FCC2, a value of 5×10^{-3} was apparently used in OPRA. BNL considered that the actions required to replenish the refueling pool (SFMPPSH = 0.01) and the elevated tank for HPSW (REHSTK = 0.01) to be similar to the actions required to maintain ASW suction such as FCC2 and FCI. Thus, a probability equal to 0.01 was used for FCI/CM12 and FCC2. This change resulted in an increase in some of the late core damage sequences in the BNL review (in particular the FMN sequence No. 18 in Appendix B).

The BNL results for the probability of failure to maintain suction to long-term ASW is shown in the last column of Table 2.7. Note that in some sequences in the OPRA, the value used for the probability of maintaining long-term ASW suction is wrongly taken from other flood initiators (e.g., sequence 3, page 9-278, or sequence 5, page 9-281).

2.3.2.2 The Results of the Core Damage Sequences Quantification in the BNL Review

The results from Appendix B are summarized in Table 2.8. This table presents the core damage for each flood-initiating event category and the frequencies for each flood category by bin, and the total core damage frequencies. It is seen that FVL2N, rupture of one of the condenser outlet pneumatic valves, is a main contributor to the early core damage frequency (LOCA type). FMN, corresponding to a medium rupture of the above valve and installation errors in one of the unisolable manways, is a main contributor to the late core damage frequency. Large floods are the main contributors to the early core damage in bin III (non-LOCA sequences).

Table 2.9 is a summary of the BNL results and a comparison with the OPRA. It is seen that the BNL review results in about the same core damage frequency as that in the OPRA. Even the breakdown into the four bins is quite similar if the OPRA core damage frequency for the very large flood (1.9×10^{-5}) , allocated to bin III, is moved to bin IV, which is more appropriate and was done in this review.

The only differences are found in the specific flood categories, mainly the large floods in which BNL grouping has made some differences.

In evaluating the detailed results, the following comments can be made:

a. The BNL assumption, according to which a lower probability of SRV challenge can be used for FVL2 than for FVL1, has reduced the main sequence by 10%. BNL believes its assumption is realistic because in OPRA a flow rate of 300,000 gpm was used to characterize a flood initiator which is dominated by a 170,000-gpm flow rate.

- b. The OPRA way of combining both inlet and outlet isolable floods for the very large category (FVL1) is appropriate. The only difference between the inlet and outlet floods in this case is in the long-term core damage sequences. However, their main contribution is to the early core damage, so that this BNL grouping made no difference in core damage frequency.
- c. The introduction by BNL of the large flood group 1 (FL1), which can reach critical level 3, has led to a moderate increase in early core damage in bin I. This is the reason that BNL calculated a higher core damage frequency of large floods in bin I. However, this effect is small compared to the main contribution coming from the very large floods. Thus, the less detailed grouping made by OPRA seems to be adequate.
- d. The treatment of the Class II in OPRA is somewhat on the conservative side, because for the FLII, FMII, and FMIO, the OPRA did not consider the probability of a successful isolation of the inlet and outlet valves. Part of the reduction in the BNL value for this class stems also from the flood grouping which transferred some of the flood contributors from FL2N to FMN in the BNL reevaluation.
- e. The same reasons stated in (d) above pertain to the small reduction in the large flood contribution to bin III seen in the BNL results.
- f. As discussed before, the CM12 and the assumption on the failure probability of the operator actions to assure long-term suction are higher in the BNL reevaluation, leading to an increase in the FMN contribution in bin IV.

2.3.3 Analysis of Additional Flood Sequences

2.3.3.1 Other Turbine Building Flood Sequences

The OPRA analysis concentrated on the flood sequences in the turbine building that originated in the CCW system. Only the parts of LPSW and HPSW that directly connect to the CCW have been considered in that analysis. The piping and valves in those systems themselves were not considered in the main study, but a scoping estimate of their contribution was made as described in the OPRA, Sections 9.5.3.5, 9.5.3.6, and 9.5.6.1. OPRA also found that EFW floods cannot reach critical level 1 and that LPSW and HPSW floods have a small contribution that was neglected.

The contribution of the LPSW can be easily derived, because all sequences corresponding to breaks in the LPSW are developed (see Appendix B) and only the frequency of the flooding initiators in the LPSW needs to be appropriately increased by the additional flooding frequency stemming from the system piping downstream of the pumps, i.e., the portion not considered in the main study.

The frequency of the LPSW breaks is derived on the basis of operating experience of four events in 436 turbine-building-years resulting in 9×10^{-3} per year. Because most plants have two service water systems, this frequency was split equally between the HPSW and LPSW systems. However, in all four cases the floods were small, i.e., floods larger than 12,000 gpm were not

experienced up to 1981. Thus, the frequency of the flood was estimated to be 10% of the total, i.e., LPSWFLOOD = 4.5×10^{-6} . On the basis of the LPSW piping diameters OPRA estimates that flood rates less than 45,000 gpm can develop which all contribute to the medium flood category. BNL assumed, in addition, that these floods occur in the inlet side of the condenser, i.e., they can be isolated from the CCW discharge lines; thus, these floods should be added to the FMII category (not to the FMN category as used in the OPRA).

Then, BNL accounted for the floods by adding this initiator frequency to that of the floods from the suction side of LPSW of unit 3, i.e., this frequency was added to the sequences including events FMII*M11F in Appendix B. Therefore, the mean value for $M11F_{NEW}$ is obtained by

FMII.M11F_{NEW} = FMII (8.0x10⁻³) * M11F_{OLD}(0.034) + LPSW_{FL00D}(4.5x10⁻⁴) = $7.2x10^{-4}$,

and

 $M11F_{NEW} = 0.09.$

Using this new value for MllF, i.e., $MllF_{NEW}$, the additional LPSW contribution to core damage becomes:

Medium	Floods	**	bin	I	-	3.5x10-7	yr-1
			bin bin	III IV		3.4×10-6 1.1×10-6	yr=1 yr=1
			Tota	1	=	4.8×10-6	yr=1

The OPRA assumed for the HPSW that the flow rates cannot exceed 20,000 gpm if a break in the piping of the system occurs. Thus, only 5% of the frequency was assumed to be in this break size resulting in a frequency of 2.3x10⁻⁺ per year for an HPSW-originated flood. However, this flood does not exceed critical level 1, and LPSW is available from unit 3 with some backup from unit 2 (part of the HPSW breaks may affect the LPSW in unit 2). Therefore, an HPSW flood does not create a challenge to unit 3 which is significantly more severe than the loss of all feedwater which was considered in the FMII category; the initiator frequency in this class, FMII, is more than one order of magnitude larger than the HPSW floodings (8x10⁻³ per year compared to 2.3x10⁻⁺ for HPSW). Its contribution is therefore covered by the FMII sequences.

2.3.3.2 Auxiliary Building Flooding

A less detailed analysis of the auxiliary building floods was performed in the OPRA. The flood-initiating events were identified by using engineering judgment to locate critical areas of the auxiliary building and then determining initiating events that could affect the critical areas. Several critical areas were identified, of which it was judged that the equipment room and the HPI pump room would have the largest potential for producing important accident sequences in the auxiliary building. The HPI pumps are located in the basement of the auxiliary building and share the room with low- and high-activity waste tanks. The floor drains in the auxiliary building lead to these tanks. Thus, flooding at higher elevations is likely to cause the tanks to overflow and flood the HPI pump room. A flood height of several feet of water is required to flood the HPI pump motors.

The LPI pumps are located in a room next to the HPI. Flooding of both rooms is not independent because direct paths are available for the water to propagate between the rooms. Thus OPRA concludes that any major flood in the auxiliary building could flood both the HPI and LPI pumps.

The equipment room is located high up in the auxiliary building and can be affected only by local flood sources which are limited. However, it includes redundant cabling required for safe shutdown such as the 600-V and 208-V switchgear for all ES loads (fail the redundant safety equipment). Its effect on CDF is smaller than that of a flood in the HPI room, mainly because the frequency of flooding of this room is more than two orders of magnitude smaller. Note that in the HPI room case any large flood in the auxiliary building will affect the HPI room, while only local floods can affect the equipment room.

The frequency of the flooding in the auxiliary building is based on two events, one in Browns Ferry 3 (4/78) and one in TMI-2 (3/79) in which a flood of several tens of thousands gallons occurred. When this is evaluated from two events in 375 auxiliary building years (up to 1981), a mean frequency of 5.3×10^{-3} /yr was obtained in OPRA. Then, a judgment was made that only 20% of this frequency is applicable to the HPI pump room situation in Oconee. The HPI room flooding frequency is therefore 1.1×10^{-3} /yr, which is considered reasonable by BNL.

From this point, the OPRA does not provide any additional details on alarms, operator procedures, timing, and accident sequences. It is only stated that from a review of the sequences for the modified plant (Appendix D.4.5 of OPRA). It is estimated that the flooding of the HPI pump room, given the above frequency of 1.1×10^{-3} /yr, would contribute less than 15% to the total core damage frequency due to internal flooding. There is not sufficient information available in the OPRA to review this estimate.

2.4 Summary

BNL has conducted a thorough review of the OPRA turtine building flooding study. The review was based on the reevaluation of the entire study. The following steps were treated, and modifications were made to them as found appropriate:

- Identification of the flooding sources.
- Review of the critical flood levels and equipment failure (by a walk-through).
- Grouping of the flood initiators into flood categories related to the critical flood levels.
- Inclusion of the modified plant additional features.
- Evaluation of the success criteria and the flooding main event tree.
- Review of all the supporting logic and the simplified fault trees as well as reconstruction of subtrees for the flooding critical levels.
- Review of the data.
- Review of the model for maintaining long-term ASW suction.
- Reguantification of the accident sequences with the SETS code.
- Review of the core damage sequences on a sequence by sequence basis.

In some of the above steps, modifications were made by BNL; however, as the results indicate, their overall effect on the total core damage frequency is very small. This is because the OPRA model has appropriately accounted for all the above steps.

The main results of the review efforts were discussed earlier in Section 2.3.2 and are summarized here:

a. The total core damage frequency for floods from sources in the plant calculated in this review is equal to 9.7E-5/yr, i.e.:

Turbine building CCW floods: 8.0E-5/yr (OPRA = 8.8E-5/yr).

Other floods in turbine building: 4.8E-6/yr.

Auxiliary building floods: 1.2E-5/yr (OPRA = 1.3E-5/yr).

Note that the OPRA has done some analysis for the other floods in the turbine building and for the auxiliary building floods. However, in their final results, they fail to present the core damage frequency contribution from these sources. In this review, they were added to the turbine building CCW floods, the major contributor to core damage frequency from internal floods.

- b. The major difference between the BNL review and the OPRA appears in bins III and IV. This difference is due to the fact that, in the opinion of the reviewers, the OPRA has assigned a sequence with frequency equal to 1.9E-5/yr to bin III (early core damage) when it should have been assigned to bin IV (late core damage).
- c. A main contributor to the medium floods in the BNL reevaluation is failure of proper reinstallation of equipment or manways following maintenance actions. In the OPRA, the main contribution to this category comes from medium rupture of the condenser outlet valves.
- d. A scoping assessment of the uncertainties about the frequency of core damage for internal flooding was performed and the results are presented in Table 2.11.

2.5 References

- Oconee PRA "A Probabilistic Risk Assessment of Oconee Unit 3," NSAC/60, June 1984.
- Meeting at DPC headquarters (Participants: DPC, NRC, BNL), June 14, 1985.
- 3. "Oconee Final Safety Analysis Report," Duke Power Company.
- Thomas, H. M., "Pipe and Vessel Failure Probability," Reliability Engineering, Vol. 2, pp. 83-124.
- "Reactor Safety Study An Assessment of Accident Risks in U.S. Commercial Nuclear Power Plants," WASH-1400 (NUREG-25/014), 1978.
- Bason, S. L. and Burns, E. T., "Characteristics of Pipe System Failures in Light Water Reactors," EPRI NP-438, August 1977.
- Anderson, O., "Failure Characteristics of Oil and Gas Pipelines," Second National Conference on Reliability, Birmingham, 1979.
- Bush, S. H., "Reliability of Piping in Light Water Reactors," IAEA Symposium, Vienna, 1977.
- Hubble, W. H. and Miller, C. F., "Data Summaries of Licensee Event Reports of Valves at U.S. Commercial Nuclear Power Plants," NUREG/CR-1363, June 1980.
- Ilberg, D. et al., "A Review of the Shoreham Nuclear Power Station PRA," NUREG/CR-4050, June 1985.
- "Nuclear Power Experience," NPE, Petroleum Information Corporation, Denver, Colorado, August 1981 issue.
- Worrell, R. B. and Stack, D. W., "A SETS User's Manual for the Fault Tree Analyst," NUREG/CR-0465, November 1978.







Uncontrollable

(OPRA Figure 9-77)

2-25



Figure 2.3 Event tree for failure to maintain long-term SSF ASW suction after a flood.

-OK

-(5) L·CM12·FCICM

0.5

(OPRA Figure 9-93)

Table 2.1 Key Events Occurring at Four Critical Elevations in the Turbine Building (Modified Plant)

Initial Level (approximately 0.5 ft) -- Elevation 775.5

```
Flood alarm
Hotwell pumps fail
Electrical loads powered from MCCs on elevation 775 of the turbine building are lost
```

First Critical Level (approximately 2 ft) -- Elevation 777

- Main feedwater pumps fail - Condensate booster pumps fail - Motor-driven EFW pumps fail - Steam-driven EFW pump fails - RCW pumps fail - Instrument-air compressors fail - Chilled-water pumps fail - HPSW jockey pump fails Second Critical Level (approximately 4 ft) -- Elevation 779 - LPSW pumps fail - HPSW pumps fail Third Critical Level (approximately 6 ft) -- Elevation 781 - Water spills into the auxiliary building and when the flood level reaches ventilation ducts, flooding of the following pumps occurs: - HPI pumps - LPI pumps - Reactor Building (RB) spray pumps

				Floo	d Level	(ft)			
(gpm)	0.5	1	2	3	4	5	6	7	8
12,000 25,000 40,000	30 15 10	120 50 25	* 100 50	120	8				
60,000 75,000 100,000	8 5 3.8	17 12 9	35 30 20	55 45 30	160 90 55		-00		
120,000 150,000	3.5	7.5	16 12	25 20	40 28	55 40	90 55	70 [∞]	
170,000 200,000 250,000 300,000	2.6 2.2 1.6 1.4	5.5 5 3.5 3	11 10 7 6.2	17 14 11 9	24 20 16 13	33 27 21 17	45 35 26 21	60 45 33 26	90 60 40 32
350,000 375,000 450,000	1.2 1.1 1.0	2.5 2.3 2.0	5.3 4.8 4.0	8 7 6	11 10 8.5	15 13 11	18 16 13.5	22 20 16	27 24 19

Table 2.2 Time to Reach Critical Levels as a Function of Flood Flow Rate (in minutes)

122	
- 60	
- 25	
- 22	
- 147	
20	
- 22	
1.00	
1.85	
- 52	
- 25	
- 100	
- 102	
1.60	
-	
- 21	
- 5	
- 58	
-	
- 64	
- 71	
- 100	
- 24	
- 50	
- 25	
100	
20	
34	
- 25	
1996	
1.1	
100	
100	
- 24	
- 52	
- 16	
- 25	
- 164	
- 26	
195	
14	
10	
191	
S at	
ds at	
ods at	
te spor	
te spoo	
Loods at	
Floods at	
Floods at	
Floods at	
g Floods at	
the Floods at	
the Floods at	
Ing Floods at	
ding Floods at	
te shoots at	
Iding Floods at	
ilding Floods at	
utiding Floods at	
suilding Floods at	
Building Floods at	
-Building Floods at	
e-Building Floods at	
se-Building Floods at	
ne-Building Floods at	
tee-Building Floods at	
bine-Building Floods at	
three-Building Floods at	
utite-Suilding Floods at	
urbine-Building Floods at	
Turbine-Building Floods at	
Turbine-Building Floods at	
Turbine-Building Floods at	
f Turbine-Building Floods at	
of Turbine-Building Floods at	
of Turbine-Building Floods at	
. of Turbine-Building Floods at	
s of Turbine-Building Floods at	
es of Turbine-Building Floods at	
ces of Turbine-Building Floods at	
ces of Turbine-Building Floods at	
rces of Turbine-Building Floods at	
arces of Turbine-Building Floods at	
urces of Turbine-Building Floods at	
ources of Turbine-Building Floods at	
Sources of Turbine-Building Floods at	
Sources of Turbine-Building Floods at	
Sources of Turbine-Building Floods at	
1 Sources of Turbine-Building Floods at	
3 Sources of Turbine-Building Floods at	
.3 Sources of Lurbine-Building Floods at	
2.3 Sources of Turbine-Suilding Floods at	
2.3 Sources of Turbine-Building Floods at	
2.3 Sources of Turbine-Building Floods at	
e 2,3 Sources of Turbine-Building Floods at	

Flood Source	Elevation (ft)	Break Stre D	Before Mod BNE	Ification OPEA	Nodified Plant	Any One of Presentic Valves Fails to Close	Relay failures: All Thesmatic Valves Fail	Particular Valve to Break Fails	Any One of Intake Valves Fails to Clos
CCM Condenser Inlet: Fipe sections [13] Monual hotterfly walkes [18] Rubber expansion joints [18]	780 780 780	78" x14 78" x14 78" x0. 5	465,000 ¹ 180,000 ² 76,000 ³	\$50,000 160,000 76,000	19,000 19,000 19,000	120,000 104,000 60,000	350,000 ¹ 172,000 72,000	120,000 104,000 60,000	313,000 ² 176,000 72,000
Manways (54) Manways (3)	790	24.v 24.v	35,000	23,000	13,000	32,000	35,000 55,000	32,000	35,000
CCM Condenser Butlet: Pipe sections [18] Dependic Nutherfly walves [19] Mubber expansion joints [18] Munways isolable [5] Manways unisolable [6]	280 275 281 281 281	78° 78°\$14 78°\$14 24°	444,000 175,000 56,300 ¹ 33,000 50,000	437,000 175,000 56,000 23,000 23,000	19,000 21,000 15,000 13,000 13,000	90,000 80,000 38,000 30,000 48,000	440,000 175,000 56,000 56,000 59,000 59,000	330,000 175,000 35,000 33,000 59,000	120,000 104,000 45,000 32,000 50,000
CCM Emergency Discharge: Pipe sections [including 4 metal expansion joints] Pipe sections Pipe sections Manual butterfly valwes	790 790 790 797	12" 12" 23" 12" 24"	50,000 32,000 20,000 2,000	<10, 900 20, 900 13, 000 <10, 000	13,090	000, 61	33,400 29,400	2*,000 18,000	33,000 20,000
CCW (rotsoer: Fipe sections (Batt 1 + 2) Manual buttarle values (1)	275	42.° 36° 16°*212	140,000 101,000* a1 000*	157,000	23,000	000,001	140,000	100,000	140,000
Pipe section (Unit 1 + 2) LPSM suction	612	36	103,000*	144,000	23,000	80,000	103,000	80,000	101,000
riperstrond (ant a) (ESM suction MOV (ESM suction (5) NESM pump suction valves	642 542	30" ×12 20"	8.3,000* 68,000* 36,000	109,000	23,000 22,000 19,000	65, 000 51, 000 29, 000	72,000 57,000 30,000	65,000 53,000 29,000	7.2,000 57,000 30,000
Unwatering Connections: Fiping sections [failet) Manual butterfly values [inlet) Piping sections (outlet) Manual butterfly values [outlet)	5424 5424 5424	$\frac{18^{\circ}}{18^{\circ}} \times 4_{\circ} S$ $\frac{18^{\circ}}{18^{\circ}} \times 4_{\circ} S$	21,000 14,000 20,000 10,000	27,000 14,000 20,000 10,000	21,000 11,000 20,000 10,000	27,400 14,000 20,000 10,000	21,000 11,000 20,000 10,000	27,000 14,000 20,000 10,000	27,000 14,000 20,000 10,000

	3		
	2		
	2		
i	ć		
		k	
1	L.	÷	
	2	ŝ	

Flood Source	[ft]	Break Stre D	Before Not BNI	if ication 0/8A	Modified Plant	Any One of Pneumatic Valyes Fails to Close	Relay Failures: All Pheumatic Valves Fail	Particular Valve to Break Fails	Any one of Intake Valves Fails to Close
Condensate Confers Inlet: Piping sections Manual hutterfly valves Auther espansion joints ⁹	780 730 730	30" 30"×12 30"×4, 5	90,000 55,000 41,000	83,000 67,000 75,000	22,000 19,000 18,000	77,000 50°, 40 41,000	85,000 54,000 64,000	77,000 50,000 41,000	85,000 54,200 64,708
Conternste Coolers Hutlet: Piping sections Manual butterfly valves (12) Manual butterfly valves (12) Control valves (6)	7.80 7.80 7.80 7.85 7.85 7.95	30" 167 30"×412 18"×4, 5 16"×4	90,090 25,000 55,000 14,000 11,000	70,,000 20,,000 56,,000 13,000 10,,000	65,000 18,000 55,000 16,000 11,000	84,000 22,000 52,000 14,000 11,000	85,000 22,000 55,000 14,000 11,000	84,530 22,000 55,000 55,000 14,000 11,000	85,000 22,000 55,000 14,000 11,000
BCW Connections: [nlet piping [from CCW cressover 1 Piping sections Piping sections Monual butterfly values [2] Monual butterfly values [2]	te PCM cooler 780 780 780 780 780	11: 18* 12**4.5 12**4.5	34,650 20,630 ³ 14,000 ⁵ <10,000 <10,000	31,200 14,090 14,090 16,090 <10,090	10,000 10,000 14,000	34,000 20,000 14,000	34,300 22,000 14,000	34,900 20,000 14,000	34,000 20,000 14,000
Outlet piping (from RCW conlers to Piping sections Piping sections Piping sections Piping sections Manual burterfly universed	2004 dischart 775 775 780 780 780 780	201: 36° 110° 110° 110° 110° 110° 110° 110°	115,000 80,000 46,000 21,000 410,000	103,000 72,000 46,000 20,000 20,000 20,000	105,000 665,000 39,000 13,000	115,080 83,000 45,000 21,000	115,000 80,000 86,000 71,000 71,000	115,000 80,000 46,000 21,000	115,000 80,000 46,000 21,000
Manual butterfly valves (4) Manual butterfly valves in RCM discharge RCM discharge	ie e	36"x12"	11, 000 63, 000 29, 000	000*69 88*,000	77,000 60,000 20,000	71,5000 60,000 20,000	77,000 60,000 20,000	77,600 60,000 20,000	77,400 60,000 20,000
Notes:									

3858

Break flow is calculated from two sides of break (this mote is marked in some cases only). Sheak flow is calculated only as coming from one side (this note is marked in some cases only). Sile of flow rate taken from OPNA, because it was calculated Duke Puwer Company using a pipiny analysis computer code. SNL included the modifications in the COM crossover in which values given in Table 9.31 are for the case of the .medified plant.

	OPRA		BNL
OPRA	Frequency	BNL	Frequency
Categorization	(per year)	Categorization	(per year)
Very large flood		Very large flood 1	
Flow rate > 160,000 gpm		Flow rate > 350,000 gpm	
Represented by 300,000 gpm		NonisolableFVLIN	1.0k 10-5
Nonisolable FVLN	1.9×10-4	Isolable InletFVL111	1.9x 10""
Isolable FVLI	3.8 10-4	isolable outletFVL110	1.9×10-4
		Very large flood 2	
		Flow rate > 170,000 gpm	
		NonisolableFVL2N	1.8×10-4
		Isolable InletFYL211	1.8x10 ⁻⁵
Large flood		Large flood 1	
Flow rate 60,000 to		Flow rate 120,000 to 169,000	
159,000 gpm		Nonisolable FLIN	3.8×10" >
Represented by 75,000 gpm		Isolable inlet FL111	5.3×10-4
		isolable inter FL110	5.5x 10-"
Nonisolable FLN	8.9×10-4	Large flood 2	
Isolable Inlet FL111	5.4x 10"3	Flow rate 60,000 to 119,000	
Isolable Inlet FL110	5.7×10-3	Nonisolable FL2N	4.1x10-4
		isolable inlet FL211	5.4×10-3
		isolable outlet FL210	5. 1x 10 ⁻³
Medium flood		Medlum flood	************
Flow rate 12,000 to 60,000		Flow rate of 12,000 to 60,000	
Represented by 30,000 gpm	and the state of the		
Nonisolable FMN	1.7×10-3	Nonisolable FMN	2.7 10-3
Isolable Iniet FMII	7.1x10-3	Isolable Inlet FMil	8.0x 10-3
Isolable outlet FMIO	7.2-10-3	Isolable outlet FMIO	1. hx 10 ⁻²
Total Flood Frequency	2.9× 10-2		3.4 10-2

Table 2.4 Categorization and Frequencies of Flood Initiators in OPRA and BNL Review

		Conditional	Probability	
	Condensat	e Coolers	LP	SW
Category	OPRA	BNL	OPRA	BNL
FL2II	L311F= 0.0054	0.005	L21F= 0.013	0.013
FMII	M321F = 0.012	0.042*	M11F= 0.031	0.034
FMN	M361F= 0.061	0.043		0.006

Table 2.5 The Conditional Probabilities That a Flood Results From the Unit 3 LPSW or Condensate Cooler System

*The additional contribution in BNL comes from the condensate coolers expansion joints.

Flood Category	Calculated Frequency (per year)	Intake Valve (COWMVC)	Any Outlet Vaive (CCWAVC)	Relay to Ali Outlet Valves (CWSRRVN)	Particular Outlet Valve (CC20AVC)	isciation of Inlet And Outlet Valves
FVL IN	1.0 × 10-5	· ·		+		+
FVLILL	1.8 × 10-4		FL12		NA	FM
FYL110	1.9 × 10-4	FL1	FL2	•	FVL2	FM
FVL2N	1.8 × 10-4		•	+	+	
FVL211	1.8 x 10 ⁻⁴	•	FL2	٠	NA	Isolated
FLIN	3.8 × 10 ⁻⁵		+			+
FLILI	5.3 × 10-4	FL2	FM	FL2	NA.	isolated
FL110	5.5 x 10-4	FM	FM	•	FL2	Isolated
FL2N	4.1 × 10 ⁻⁴			•		
FL211	5.4 × 10-3				NA	Isolated
FL210	5.1 x 10 ⁻³	FM	FM	•	•	Isolated
SMN	2.7 × 10-3	+	+	+		+
FMII	8.0 × 10-3				NA	Isolated
FMIO	1.1 x 10-2	+	+			Isolated

Table 2.6 BNL Selection 1 of Flooding Categories for the Flood-Isolation Failure Possibilities

¹Based on flow rate from Table 2.3 and dominant flood-initiating events from Table A.1.

2FVL1 -- Very large flood 350,000 to 450,000 gpm.

FVL2 -- Very large flood 170,000 to 349,000 gpm.

FL1 -- Large flood 120,000 to 169,000 gpm.

FL2 -- Large flood 60,000 to 119,000.

FM -- Medlum flood 12,000 to 60,000.

+No change in flood category.

Flood- Initiating	Flood Source Located	Flood Is M Isolate	Not Locally ed (FI)	Failure to Sustain Long-Term ASW
Event	(L)	F1/F13	F1/F13	Suction (ASWLIF)
FVL1N	0.2	1.0	1.0	0.052
FVL2N	0.2	1.0	1.0	0.042
FVLIII	0.01	ε	1.0	0.006
FVL1I0	0.01	ε	ε	ε
FVL211	0.01	ε	1.0	0.006
FL1N	0.05	1.0	1.0	0.019
FLIII	0.01	ε	1.0	0.0035
FL1I0	0.01	ε	ε	ε
FL2N	0.05	1.0	1.0	0.011
FL2II	0.01	ε	0.8	2.0027
F'.210	0.01	ε	ε	3
FMN	0.01	0.8	0.8	0.008
FMII	0.01	ε	0.8	0.0026
FMIO	0.01	ε	ε	ε

Table 2.7 Values for the Failure to Maintain Long-term SSF ASW Suction After Flood^a

aSee Figure 2.6 for event tree where these events are used. bOther events in event tree in Figure 2.6: FCC2 = 0.01 in all cases. FCI = 0.01, given no CM12. If CM12 occurred, FCI = 0.5.

		* ********				Fractional	Total	Contribution
Flood	Frequency	BIN	BIN	BIN	BIN	Conditional Probability	Conditional Probability	of Flood Initiator to Core-Damage
Initiators	(yr ⁻¹)	1	11	111	14	CM12*	of Core Damage	Frequency (yr ⁻¹)
FVLIN	1.0-5	0.14	-	-	0.10	0.09	0.24	2.4-6
FVL2N	1.8-4	0.11	-	-	0.08	0.07	0.19	3.4-5
FVLIII	1,8-4	0.011	4.0-4	7.2-4	1.6-3	0.008	0.014	2.5-6
FVL110	1.9-4	0.010	1.3-4	7.4-4	1.2-3	0.008	0.012	2.3-6
FVL211	1.8-5	8.3-3	1.5-5	5.6-4	1.0-3	0.006	0.010	1.8~7
FLIN	3.8-5	0.040	-	-	0.040	0.027	0,080	3.0-6
FLIII	5.3-4	4.4-5	66	5.4-4	2.3-4	3.0-4	8,3-4	4.3-7
FL110	5.5-4	1.5-3	16	2.5-5	1.0-4	7.5-4	1.6-3	8,8-7
FL2N	4.1-4	1.4-3	3.8-4	8.0-3	6.8-3	0.005	0.016	6.6-6
FL211	5.4-3	1.3-4	2.0-5	8.0-4	2.4-4	5.0-4	1.2-3	6.5-6
FL210	5.1-3	4.0-5	8.0-6	2.6-4	3.2-5	1.5-4	3.3-4	1.7-6
FMN	2.7-3	3.7-5	1.0-4	1.7-4	4.0-3	1.0-4	4.4-3	1.1-5
FMII	8.0-3	6.0-5	5.0-6	3.9-4	2.0-4	2.0-4	6.6-4	5.3-6
FMIO	1.1-2	-	2.0-6	1.0-4	2.0-6	5.0-5	1.0-4	1.1-6
TOTAL	3.4-2						2.4-3	7.8-5
CONTRACTOR OF THE OWNER WAS ADDRESSED.	the second se	and the second s	and the second sec	the second se	the second se			

Table 2.8 Core Damage Conditional Probabilities of the Various Flood Categories

*CM12 is a conditional probability that given early core damage in unit 1 (unit 2), unit 3 would be in a non-core-damage state.

Core Damage	Flood Initiator	OPRA Results	BNL Results
I	Very Large Flood Large Flood Medium Flood Total Bin I	2.9E-5 2.6E-6 6.6E-7 3.2E-5	2.5E-5 3.8E-6 5.8E-7 3.0E-5
II	Very Large Flood Large Flood Medium Flood Total Bin II	8.2E-7 8.0E-7 1.6E-6	1.0E-7 2.8E-7 <u>3.2E-7</u> 7.0E-7
III	Very Large Flood Large Flood Medium Flood Total Bin III	1.9E-5* 1.3E-5 <u>4.8E-6</u> 3.7E-5	2.8E-7 9.2E-6 <u>4.7E-6</u> 1.4E-5
IV	Very Large Flood Large Flood Medium Flood Total Bin IV	9.7E-6 7.7E-6 1.7E-5	1.6E-5 6.1E-6 <u>1.3E-5</u> 3.5E-5
TOTAL		8.8E-5	8.0E-5

Table 2.9 Summary of Core Damage Frequencies for Turbine Building CCW Floods

*Binning error -- should be in Bin IV.

	Turbine	Building	Aurilian Duilding
	CCW Floods	Other Floods	Floods*
Bin I	3.0E-5	3.5E-7	4.5E-6
Bin II	7.0E-7	-	1.1E-7
Bin III	1.4E-5	3.4E-6	2.1E-6
Bin IV	3,5E-5	1.1E-6	5.3E-6
TOTAL	8.0E-5	4.8E-6	1.2E-5

Table 2.10 BNL Review Summary of Core Damage Frequencies for Floods from Sources Within the Plant Total CD Frequencies = 9.7E-5

*Assumed to be 15% of the core damage frequency for CCW floods (see OPRA, page 9-285).

Bin	X ₀₅	X 50	Mean	X 95
I	3.2E-6	1.6E-5	3.4E-5	1.2E-4
11	3.2E-8	2.6E-7	7.5E-7	2.5E-6
111	1.2E-6	7.2E-6	1.7E-5	5.9E-5
IV	2.1E-6	1.5E-5	3.9E-5	1.5E-4
Total CD	1.3E-5	5.2E-5	9.1E-5	2.8E-4

Table 2.11 BNL Review Core Damage Frequency Distribution for Floods from Sources Within the Plant

3. REVIEW OF SEISMIC EVENTS

3.1 Introduction

The Oconee Probabilistic Fisk Assessment (OPRA) provides a detailed analysis of the effects of seismic events as described in Section 9.1 and Appendix J of the OPRA. In the OPRA, the seismic contribution to the core damage frequency was evaluated in the following four steps:

- The Oconee site was evaluated to obtain the seismic hazard in terms of the frequency of occurrence of ground accelerations.
- The capacities of important plant structures and equipment to withstand earthquakes were evaluated to determine conditional probabilities of failure as a function of ground acceleration.
- 3. The event tree and fault tree models developed for the internal initiating events were modified to reflect plant response to seismic events. These modified logic models were then solved to obtain Boolean expressions for the seismic event sequences of interest.
- The Boolean expressions were quantified by convolving the probabilistic site seismicity and the fragilities for the plant structures and equipment obtained in steps 1 and 2.

Jack R. Benjamin and Associates, Inc. (JBA) was retained by BNL to perform a preliminary review of the OPRA for the effects of seismic events. JBA was the main contributor to this section, with participation of the:

- BNL/Risk Evaluation Group for the review of event tree and fault tree models with the generation of the Boolean expressions for the seismic sequences, and
- BNL/Structural Analysis Division for a qualitative review of Appendix J--Annex J3 of the OPRA: "Seismic Fragility Curves for Jocassee Dams and Oconee Dikes," prepared by Danielle Veneziano, June 1981.

The JBA review focused on the following sections of the OPRA:

- o Section 9.1: "Analysis of Seismic Events"
- o Appendix J Annex J1:

"Seismic Ground Motion Hazard, Oconee Nuclear Power Plant Site, Oconee, South Carolina, prepared by Law Engineering Testing Company, May, 1981.

o Appendix J - Annex 12:

"Conditional Probabilities of Seismic Induced Failures for Structures and Components for Oconee Generating Station Unit 3," prepared by Structural Mechanics Associates, September 1981. It is our understanding that the seismic fragility analyses for the PRAs published to date by Structural Mechanics Associates were performed in the following order:

Zion Indian Point, units 2 and 3 Oconee, unit 3 Midland Limerick Seabrook Millstone, unit 3

Thus the OPRA is older than the PRAs conducted for Seabrook and Millstone. This is particularly pertinent to the approach adopted for the interface between the seismic hazard and fragility data as discussed in Section 3.2. In terms of the general approach, the OPRA is similar to the PRAs conducted for Zion¹ and Indian Point.² These three PRAs differ from the PRAs listed above in the manner in which the hazard curves are defined, and the way in which duration factors, ductility factors, etc. are incorporated. This is discussed further in Section 3.2.

The review of the OPRA focused on the critical issues which may significantly affect the results. In contrast to the review of the Zion and Indian Point PRAs, which consisted of an in-depth evaluation of each section and subsection of the PRA report,^{3,4} this review focused primarily on critical areas which may affect the mean frequency of core melt. The reader is directed to References 3 and 4 which give specific comments on the report sections of the Zion and Indian seismic fragility analyses, respectively. Many of those comments also generally apply to the OPRA since the fragility reports for all three PRAs were performed at approximately the same time and are very similar.

In the review of the OPRA, an attempt was made to look for both conservative and unconservative assumptions which could significantly affect the results. To help the reader, an effort is made to indicate, where possible, the ultimate impact of the issues which have been raised. Comments are primarily directed to the mean frequency of core melt or to the individual sequences which contribute significantly to core melt. The following scale has been adopted to quantify comments made in the review of the OPRA:

Comment	Effect on Mean Frequency of Core Melt				
Small	Factor < 2				
Moderate	2 < Factor < 10				
Large	Factor > 10				

The methodology used in the OPRA for seismic effects is appropriate and adequate to obtain a rational measure of the probability distribution of the frequency of core melt. The results from the OPRA are useful in a relative sense and should not be viewed as absolute numbers. The procedure used to quantify seismic risk is based on simple probabilistic models which use some data, but currently rely heavily on engineering judgment. The analysis does not include a comprehensive consideration of design and construction errors and, hence, may be biased (note that errors may be either conservative or unconservative). Because of the newness of these types of analyses and the limitations pointed out above, the results are useful only in making relative comparisons. Although more sophisticated analytical models exist, the limitation of available data dictates that the simple models used in the OPRA are in a practical sense at the level of the state of the art; note that some improvements have been made in more recent seismic PRAs as discussed below. Concerns about the basic data used in the seismic PRA analysis are discussed in the following sections.

3.2 Seismic Hazard

3.2.1 Review Approach

A critical review was conducted of Annex J1 of the OPRA which describes the methodology and assessment of the seismic ground motion hazard at the Oconee site. Section 9.1 of the OPRA summarizes the seismic risk methodology and the results of the probabilistic seismic hazard analysis. To assist in the review, the services of a consultant, Professor Pradeep Talwani, were retained by JBA to review Annex J1 from the seismologist's viewpoint. Professor Talwani's report is provided in Appendix D to this review, while important points are incorporated in the body of this report.

As part of the review, the interim results of USNRC research performed by Lawrence Livermore National Laboratory (LLNL), the Seismic Hazard Characterization Program (SHCP), ⁵ are used. Although specific probabilistic estimates for the Oconee site are not available, seismic source zone characterizations and seismicity parameter estimates are provided by the experts who participated in the SHCP. These data are used for comparison with the Oconee seismic hazard analysis. In addition, the results of the U.S. Geological Survey (USGS) seismic hazard calculations for the contiguous U.S. also provide a basis for comparison.

The review of the seismic hazard analysis in the OPRA concentrated on a number of issues. To begin, the adequacy and appropriateness of the analysis approach to estimate the probability distribution on the frequency of ground motion is considered in Section 3.2.2. Individual elements of the seismic hazard analysis, including seismotectonic zones, seismicity parameters, and the ground motion characterization, are reviewed in Sections 3.2.3 to 3.2.5, respectively. Review comments on the hazard/fragility interface approach are discussed in Section 3.2.6. In Section 3.2.7 the results of the Oconee hazard analysis are briefly compared with those of the USGS. As part of our review, we performed seismic hazard calculations to verify the hazard estimates reported in the OPRA and to consider the effect of our review comments on the results. The insights and findings from these calculations are reported in Section 3.2.8. Concluding remarks are given in Section 3.2.9.

As pointed out in Section 3.1, the OPRA is an older PRA; the seismic hazard analysis was performed in late 1980 and completed in May 1981. Since that time there have been significant developments in the methods used to conduct probabilistic seismic hazard studies.⁵,⁷ In addition, there is a greater availability of scientific data and hypotheses on the mechanisms that cause earthquakes in the eastern U.S.,⁵,⁷⁻⁹ and a greater body of experience in performing these studies.¹⁰⁻¹² Similarly, there has been further work in

ground motion attenuation models. In comparison, the general approach used in the Oconee seismic hazard analysis is similar to that used in other seismic PRAs listed in Section 3.1, in particular, the Zion and Indian Point studies.^{1,2}

The review of the Oconee seismic hazard assessment focused on important elements of the analysis that may significantly affect the results. It is particularly important to determine that accurate and up-to-date information was used in the study (available at the time the study was performed), and to evaluate the adequacy and appropriateness of the seismic hazard results.

3.2.2 Seismic Hazard Methodology

be seismic hazard analysis methodology used in the OPRA follows well-established procedures for evaluating the frequency of exceedance of ground shaking.¹³⁻¹⁵ The basic steps in the analysis are:

- Collection of historical seismicity data, geophysical, geologic, and tectonic information.
- Establishment of seismic source zones based on available data and expert input regarding causative mechanisms of earthquakes.
- Development of seismicity parameters that describe the spatial and temporal frequency of earthquake occurrences (i.e., maximum magnitude, b-values, activity rates).
- Selection of a method of characterizing ground shaking and corresponding ground motion attenuation models.
- Solicitation of expert opinion regarding alternative approaches to model the occurrence of seismic events and the intensity of ground motion for each of the above steps.
- Solicitation of expert probability assignments that characterize the degree-of-belief in each alternative for each hypothesis.
- Calculation of the frequency of exceedance of ground motion per year for the family of seismic hazard modeling alternatives.
- Aggregation of the results to establish the probability distribution on the frequency of exceedance.

The use of this procedure in the Oconee seismic PRA to evaluate the groundshaking hazard at the plant site is considered appropriate and adequate.

In actual application, the analyst has considerable latitude in defining how each step in the analysis is performed and to what level of detail. For example, the analyst controls the selection of experts, the method of soliciting expert opinion, and the degree of documentation in reporting the study results. Overall, the seismic hazard analysis used in the OPRA is the same methodology used in previous PRAs submitted to the USNRC.^{1,2},¹⁰⁻¹² The adequacy of individual aspects of the analysis is discussed in subsequent sections.

Soliciting Expert Opinion

To a large extent, seismic PRAs and seismic hazard analyses in particular rely a great deal on expert judgment to estimate the value of key parameters and to assess the uncertainty in such estimates. Thus, an integral part of the analysis is associated with soliciting expert input and quantifying subjective judgments. For the most part, an ad hoc approach has been taken in past seismic PRAs in soliciting expert opinions and in establishing subjective probability weights. Among the generally recognized inadequacies associated with this type of approach are the arbitrary assignment of subjective probability weights, failure to identify the sample space of key random variables (i.e., range of possible values), bias, miscalibration, lack of coherence in expert statements, and failure to adequately assess the uncertainty in expert judgments, among other potential problems. We suspect that the OPRA suffers from a number of these problems, although it is difficult to establish this quantitatively. In comparison, however, the approach used in the OPRA is similar to that in previous studies. In this review, the following three aspects of the process of soliciting expert judgments and subjective probabilities are considered:

- Methodology approach used to solicit and combine multiple expert input,
- Application how the methodology was applied, how many experts were used, and
- o Documentation completeness in reporting the results of the study.

Within the context of these broad categories, comments on the approach used in the Oconee seismic hazard analysis are discussed below.

The Oconee study does not provide a specific discussion of the method used to quantify the uncertainty in key aspects of the hazard analysis. For an uncertain parameter in the analysis, alternatives are defined and subjective probability weights are assigned. There is no definitive discussion of the approach taken to identify alternative hypotheses or to assign subjective probabilities. From previous experience in reviewing seismic PRAs, the Oconee study uses an ad hoc approach which, at a minimum, suffers from a lack of a systematic, coherent method of quantifying subjective judgments. At worst, subjective probability assignments estimated in this way could be an inappropriate characterization of the professional uncertainty in the seismic hazard. Although it is true that the state of the art in this area is advancing rapidly, nonetheless, a clear presentation of the approach used should be provided.

In the seismic hazard analysis report, limited documentation is provided regarding the assessment of subjective probabilities assigned to alternative model hypotheses. Specifically, little or no information is available on the following:

- the experts (i.e., seismologists, geologists, etc.) who provided alternative model assumptions and assigned subjective weights for seismic source zones, seismicity parameters, and attenuation models;
- o methodology used to solicit subjective input and the procedure to combine input from a group of experts (if there was more than one);
- o supporting scientific basis for individual model hypotheses.

As a result of the limitations in these areas, documentation is inadequate to support the probability distribution on the frequency of exceedance per year of ground shaking. In effect, the reader is expected to accept the seismic hazard model and modeling uncertainties that are presented on faith.

In past seismic hazard studies conducted for PRAs, 1, 2, 10-12 a limited number of experts (possibly only one) were consulted to evaluate the modeling uncertainty in various phases of the analysis. As a result, the process of identifying credible parameter values or model hypotheses may be self-limiting in the sense that the one or two experts participating in the analysis represent a restricted sample of the range of possible expert opinions. This observation is supported by the fact that a comparison between seismic hazard studies using many experts and those using only one or two shows greater variability in the probability distribution on frequency. Figure 3.1 gives an example of the logarithmic standard deviation of the frequency of exceedance at different peak ground acceleration levels as estimated in various site-specific seismic hazard studies, and in the LLNL Sesimic Hazard Characterization Project.⁵ In our review of recent seismic PRAs, 16-19 a similar concern was expressed that the uncertainty in key parameters in the hazard analysis was not adequately represented, which ultimately limits the assessment of the probability distribution on the frequency of ground motion.

A second issue that is strongly influenced by the number of experts taking part in the analysis is the central tendency of the hazard curves compared to the results of a multi-expert analysis. For example, it might be anticipated that the results of a hazard analysis that uses a limited number of experts (i.e., one or two), could diverge from a study using many experts. That is, any one expert in a small group can differ from the group and affect the results significantly.

In summary, an ad hoc approach was used in the Oconee seismic hazard analysis to identify alternative modeling assumptions and to quantify subjective probability weights denoting degree of belief. In general, the hazard analysis methodology is consistent with other PRAs that have included external events. At the same time, the study fails to thoroughly document essential aspects of the analysis, and thus results of the uncertainty analysis are in many respects unsupported.

3.2.3 Seismotectonic Regions

In Section 2.1 of Annex J' of the OPPA, the seismotectonic regions used in the seismic hazard calculations are defined. In all, a single seismotectonic model for the southeastern U.S., composed of seven source areas, was used. As a general comment, consideration of one seismotectonic hypothesis does not parallel past studies wherein multiple hypotheses are considered. This is due to the differences of opinion between earth scientists as to the causal mechanisms of earthquake occurrences in the East. In fact, variations in seismotectonic regions are generally believed to contribute significantly to the uncertainty in seismic hazard assessments.

The delineation of seismic zones is difficult, since it requires an evaluation of the mechanism of earthquake occurrences in the region surrounding a site. As part of this evaluation, the earth scientist must identify tectonic features or regions that exhibit (or are assumed to have) consistent patterns of seismicity. Since the origin of earthquakes in the eastern U.S. is not known, the seismic hazard analyst, in consultation with geophysical scientists, must interpret uncertain information and quantitatively document the state of knowledge with respect to the causative mechanism of earthquake occurrences. Available information exists partially in the form of physical data (i.e., observed geologic, geophysical properties), observation (i.e., historical seismicity), and scientific hypotheses that attempt to provide a coherent explanation of earthquake occurrences. Because of the limited physical data base and observational record, there is considerable scientific speculation regarding the cause of earthquakes in the East. As a result, the analyst must rely more on the opinions and speculation of experts than on physical data to establish source zones.

The delineation of seismogenic zones is important in the seismic hazard analysis since it establishes the geometric pattern of earthquake occurrences relative to a site. In addition, it also defines the subset of historical earthquakes that serve as the principal basis for estimating seismicity parameters for each seismogenic zone.

In the OPRA hazard analysis report a limited discussion is provided to support the choice of the seismotectonic model adopted which, given the overall importance of the definition of source zones and the subjective nature by which they are generated, is considered to be inadequate. In our view, a complete, comprehensive discussion should be provided to support the seismotectonic model developed and the credibility assigned to it (i.e., the subjective probability weight). The discussion should focus on a number of issues which include (but may not be limited to) the following:

- o regional geologic and tectonic structures
- o assessments of crustal stress patterns
- o geophysical data
- o physiographic data
- o correlation of historical seismicity patterns with geologic and tectonic structures
- summary of the hypothesized earthquake generation process based on the above points.

With respect to the delineation of seismic source zones, there is no evidence that previous studies on eastern U.S. hazard assessments $^{20-22}$ or seismicity data were used directly in the process of delineating source

areas. Given the subjective nature of seismic hazard evaluations in the East and the limited available data, the documentation in the report leads to questions concerning the completeness of the seismotectonic evaluation.

Figure 3.2 shows the seismotectonic model used in the OPRA to evaluate the ground motion hazard. This model was assigned a 1.0 weight, suggesting that there are no other realistic tectonic models that should be incorporated in the analysis or that all other relevant tectonic models are reasonably embodied in the model in Figure 3.2. The former view seems inappropriate, given that other studies available at the time 20-22 offer a number of alternative seismotectonic models. The latter view is a matter of practical concern since the only way to actually evaluate whether the results of other sources are implicitly included is to run the actual cases.

In his review of the seismotectonic source areas, Professor Talwani (see Appendix D) indicates his basic agreement with the seismotectonic model and source areas that have been used. The exception to this is his recommendation that a southern Appalachian region should be modeled as a single source, as suggested by patterns of seismicity. As a result, our concerns are focused less on the scientific merit of the seismotectonic model that was used, but rather on hypotheses that have not been considered.

A comparison of the Oconee seismotectonic model with the 1976 USGS study ²¹ and the initial Lawrence Livermore National Laboratory seismic hazard study ²⁰ was made. Figure 3.3 shows the USGS source areas used in their 1976 study. Relative to the seismic hazard at the plant, USGS source areas 64 and 65 represent an alternative seismotectonic model to that used in the Oconee study. One consequence of this model is the fact that the Charleston zone (No. 65) is considerably closer to the plant site. As a result, this model would allow Modified Mercalli Intensity X events to occur much closer to the plant. It is anticipated this would result in a higher estimate of the hazard at the plant.

Comparisons with the LLNL study²⁰ in which experts were asked to assess seismic sources indicate that a variety of alternative model approaches were proposed that differ significantly from the Oconee seismotectonic model. Figures 3.4 and 3.5 are two examples of models that differ from that used in the Oconee study.

The development of seismic sources in the Oconee hazard analysis generally parallels the physiographic provinces identified in the eastern U.S. As discussed in Appendix D, it would appear that source zones that would be derived from analysis of seismicity patterns were not considered. In particular, Bollinger²² has analyzed the seismicity in the southeastern U.S. Figure 3.6 shows the seismic source areas suggested in Reference 22. The observation from Figure 3.6 is that historical patterns of seismicity suggest different source zone interpretations from those derived from physiographic data.

Further comparison with the most recent LLNL study⁵ in which 11 seismotectonic models for the eastern U.S. were developed lends additional support to a conclusion that considerable variation exists in the opinion of experts concerning the nature and extent of earthquake occurrences in the proximity of the Oconee plant. This leads us to conclude that the seismotectonic evaluation in the Oconee hazard analysis is incomplete, failing to incorporate reasonable alternative seismotectonic models.

We suspect that the Oconee seismic hazard assessment may be unconservative because alternative seism ectonic models were not considered. It is difficult to assess the impact of other seismotectonic models quantitatively without the benefit of actual hazard computations. However, a review of source zones developed by others leads at least in part, to alternative estimates of seismicity parameters, in particular, maximum intensity (or magnitude). Therefore, all other factors being equal (i.e., attenuation models, etc.), consideration of additional seismotectonic models would lead to increased uncertainty in the seismic hazard analysis and a higher mean or best estimate of the frequency of ground shaking. Experience suggests that increasing the maximum event size that can occur at or near a site can have a moderate or greater effect on the results (i.e., a factor of 2 or greater).

3.2.4 Seismicity Parameters

The adequacy of the assessment of seismogenic zones is realized in the quantification of seismicity parameters and ultimately in the predicted spatial distribution of earthquake occurrences near a site. In this review, the latter is difficult to assess without the detailed results of seismic hazard calculations and thus it is addressed here only in a qualitative manner. In this section, the seismicity parameters defined for each seismogenic zone are reviewed. We used the results of the SHCP to make quantitative comparisons with the Oconee results. With this information, the following seismicity parameters were estimated:

- o seismic activity rates
- o Richter b-values
- o maximum intensities.

In determining the seismicity parameters for each seismogenic zone, a number of initial assumptions were made. A lower bound on earthquake intensity of V was assumed. Although no basis was given to support this choice of a lower-bound intensity, it is considered reasonable.

The seismicity parameters used in the Oconee hazard analysis are given in Table 3.1.

Seismogenic Zones

The adequacy of the seismogenic zones defined in the OPRA was reviewed in the preceding section. For each zone, a subset of earthquakes from the historical catalog is used to estimate the seismicity parameters.

Earthquake Catalog

The hazard analysis report does not state which earthquake catalog was used in the analysis. It is simply stated that a historical data base was compiled from several different published catalogs. It would have been useful to have a plot of the regional seismicity and to identify in the text which earthquake catalogs were used.

Earthquakes that occurred in the period 1870 to 1979 were used, for which it is claimed that the data base is complete for Modified Mercalli intensities (MMI) greater than IV. No basis is given to support this statement.

Seismic Activity Rates

For each seismotectonic zone, seismic activity rates for earthquakes of Modified Mercalli intensity V and greater were estimated which describe the annual frequency of occurrence of earthquakes. In the Oconee hazard analysis, the seismic activity rate was estimated by simply counting the number of events of MMI \geq V in the 110-year period from 1870 to 1979, and calculating the per annum rate.

It is generally recognized that problems of accuracy and completeness exist in most earthquake catalogs. Potential problems are not restricted to historical earthquakes, where they are naturally expected. Recent studies have pointed to inconsistencies in magnitude estimates of smaller events (i.e., m_D less than 5.0). This could influence estimates of activity rates, because of the relatively high rate of occurrence of small magnitude earthquakes.

In order to assess the Oconee hazard analysis in light of the interim results of the SHCP, the activity rate estimates of the two studies were compared as follows. For each zone on an expert's base map reported in Reference 5 that encompasses the Oconee site, the seismic activity rate in terms of the number of events per year per square kilometer was estimated. In this way the effect of zone size does not influence this initial comparison of seismic activity rates in the two hazard analyses. This approach assumes that the zone which encompasses the Oconee site has the greatest contribution to the hazard. The SHCP values are given in Table 3.2. Note that they should be compared to the Piedmont source area in the Oconee study (see Table 3.1). A weighted average of the SHCP estimates (as determined by the experts' selfweights) gives 5.35E-6 events per year per square mile, which is about 12% higher than the estimate for the Piedmont and Upper Coastal Plain source in the Oconee study. This difference is considered small. It should be noted, however, that the experts' estimates of seismic activity rates vary widely.

Richter b-values

The Richter b-value defines the relative distribution of large earthquakes to smaller events that occur in each seismogenic zone. In the Oconee seismic hazard analysis, b-values were estimated by a least squares fit to the data. As pointed out in Appendix D, the estimated b-values fall in the range of values estimated in other studies for this region

In Table 3.2, best-estimate b-values provided in the SHCP are given for the host zone for each seismicity expert. To convert b-values estimated in terms of magnitude, a factor of 0.50 was used to obtain a b-value for intensity. The experts in the SHCP exhibit a wide range of b-values, from 0.44 to 0.64. The weighted average of the experts is 0.52 which compares to 0.56 for the Piedmont source in the Oconee study. These estimates are considered to be in close agreement.

In the Oconee study, variability in b-values was not formally incorporated in the hazard analysis, where as in the SHCP, the best estimates of the experts vary by 20%. In addition, there is the variability assigned by each expert. Thus, the range of possible b-values may actually vary by 40%.

We would agree that variations of b-values of 20% do not, by themselves, have a major effect on the seismic hazard results. However, since the b-values are correlated with estimated activity rates, they must be considered jointly in order to determine their influence on the frequency of earthquake occurrences. They also depend on the seismic source areas. It is difficult therefore to subjectively assess the overall interaction effect of activity rates and b-values. In general, it is our opinion that it is appropriate to formally consider the variability in b-values, along with the variability in other key parameters such as seismotectonic sources, activity rates, and maximum magnitudes. The effect of not considering the variability in b-values is believed to be small (i.e., less than a factor of 2).

Although b-values were estimated for each seismic source, discussions with the authors of the Oconee hazard analysis indicated that a constant b-value of 0.51 was used for all seismic zones. This value was based on data for the entire southeastern U.S. This approach is not appropriate in our opinion, nor would it appear to be the consensus of the seismicity experts in the SHCP.⁵ Although the difference between the assumed b-value of 0.51 and the values obtained for each seismic source is not great (i.e., about 20%), this approach introduces a bias in the analysis. For example, by selecting a constant b-value of 0.51 for the Piedmont and Upper Coastal Plain source area, greater relative likelihood of occurrence of large-intensity events is estimated, since the estimated value of b for this zone was 0.56. The effect on the hazard analysis results of adopting a constant b-value and not incorporating the uncertainty in the estimate is believed to be small (i.e., less than a factor of 2).

Maximum Earthquakes

The estimate of the maximum earthquake size for each seismogenic zone is an important step in the hazard analysis. In the Oconee study, three different hypotheses were used to estimate the maximum intensity earthquake that can occur in each seismic source:

- Maximum earthquake in each seismotectonic region is equal to the historical maximum in that region.
- Maximum earthquake is equal to the intensity of the event with a 1000-year return period.
- Maximum earthquake is equal to the intensity of the event with a 1000-year return period plus one intensity unit.

Each estimate of a maximum epicentral intensity was assigned a subjective probability. For hypotheses 1, 2, and 3, the corresponding weights assigned were 0.50, 0.30, and 0.10, respectively.

Because of the brevity of the historical record and the scarcity of scientific evidence, maximum intensities are difficult to estimate. The approach taken in the Oconee study is yet another attempt to make such estimates. As noted in the hazard report and by Professor Talwani in Appendix D, the use of hypothesis 2, and thus hypothesis 3, is not a very defensible method; however, this method does seem to give reasonable results for the eastern U.S.

In the Oconee study, the Piedmont source area is stated to have a maximum historical intensity of VII and thus, according to hypothesis 1, this is the lowest estimate of the maximum intensity. In his report, Professor Talwani (see Appendix D) notes that the 1913 earthquake in Union County, South Carolina, was assigned an intensity VIII by Taber²³ and VII to VIII by Bollinger, ²⁴ rather than the value of VII stated in the report. This would change the maximum-intensity estimate made according to hypothesis 1 and would undoubtedly change the Richter b-value estimate, and thus the estimates for hypotheses 2 and 3.

In summary, assigning an intensity VIII to the Union County event changes the maximum earthquake in the Piedmont and Upper Coastal Plain as follows:

Maximum Earthquake

Hypothesis 1	VIII
Hypothesis 2	IX
Hypothesis 3	X

Although the assignment of probability weights to alternative hypotheses is a subjective process, we would disagree with the values assigned to the three methods used to estimate maximum intensities. In particular, it is our opinion that less weight should be assigned to the estimate given to hypothesis 1, and more weight to the other approaches. In past seismic PRAs,^{1,2} and in other studies,⁵ there is precedent to assign the highest subjective probability to a value equal to approximately one intensity unit higher than the maximum historical event itself. There are of course possible exceptions to this approach (i.e., the New Madrid and the Charleston source areas). The effect of shifting the probability weights is, in itself, probably small.

The distribution of maximum intensity used in the Oconee study can be compared to the estimates given by the experts in the SHCP (see Table 3.2). Recall that the values reported in Table 3.2 for the experts correspond to the host source zone on the expert's base map.

In the Oconee study, the average estimate of the maximum intensity is 7.7 for the Piedmont source area, whereas the weighted average of the SHCP experts is 8.98. This is considered to be a large difference. In addition we note that the SHCP study has a wide range of possible maximum intensity values from VII to XII. (One expert assigned an upper-bound estimate of the maximum magnitude of 8.5.) For the Piedmont source area in the Oconee study, the range of intensities is from VII to IX, whereas the average estimate in the SHCP was nearly IX.

In summary, two major points are raised with respect to the estimate of maximum intensity. First, using the procedure in the Oconee study, the

maximum intensities for the Piedmont source area appear to be underestimated by one intensity. Second, there is a large discrepancy between the estimates of maximum intensities developed in the SHCP and the Oconee study. The average values for the SHCP host zones differ are 1.3 intensity units higher. In addition, the SHCP distribution of maximum intensities includes values much higher than the Oconee analysis. The effect of these differences on the seismic hazard curves is estimated in Section 3.2.8.

3.2.5 Ground Motion Characterization

In this section the approach used to model the attenuation of ground motion is reviewed. Comments are also given on the use of upper-bound accelerations.

3.2.5.1 Ground Motion Attenuation

To describe the variation of earthquake ground shaking as a function of MMI and distance, the following two attenuation models were used.

- McGuire MMI attenuation relation²⁵ and the Computer Science Corporation²⁶ ground acceleration-MMI relation,
- o Tera Corporation²⁷ ground acceleration-MMI intensity relation.

These models served as the basis for predicting median peak ground acceleration values. A lognormal distribution about the median attenuation curve is used to account for the randomness in ground motion estimation.

To aid us in our review, we have used the compilation of attenuation relationships in the SHCP^5 as a basis of comparison with the models used in the Oconee study.

As a general comment, the use of intensity-based attenuation relationships raises a number of concerns. As pointed out in the Oconee report, the development of acceleration attenuation relations using MMI is a two-step process. First, site intensities are determined, followed by a conversion to peak ground acceleration. The range of alternative models to make this transformation is wide, and thus there is considerable model uncertainty in predicting strong ground motion.

A second concern centers on the basic use of intensity as a means to predict ground motion. Since intensity is, by definition, a subjective measure of the severity of building <u>response</u>, it is limited in its ability to resolve the issue of <u>ground-shaking intensity</u>, as indicated by the wide variation in PGA for given levels of MMI. As a result, there is an inherent bias in intensity relationships designed to predict peak ground acceleration, related to the fact that the process of damaging structures has filtered the input ground motion to produce a particular observed outcome (i.e., structural damage). Note that engineers have generally agreed that PGA is a poor measure to use in predicting structural damage. The converse is equally true.

In the seismic hazard report documentation or discussion of the basis for selecting the two ground motion models used in the analysis is limited. For

example, was Bollinger's²⁸ intensity attenuation relationship based on data from the 1886 Charleston, South Carolina, earthquake considered? Similarly, was the MMI-PGA relationship of Trifunac and Brady,²⁹ or the other attenuation models developed by Tera Corporation considered? These questions reflect a general concern that other reasonable attenuation models were not considered in the study.

In Figure 3.7, the two attenuation models used in the Oconee study are compared to the best-estimate ground motion models (for rock) selected by the experts in the SHCP⁵. To plot the curves, the relationship, MMI = $2 M_b$ -3.5 was used to convert MMI to magnitude. Figure 3.8 presents a similar comparison with other models provided by the ground motion experts.

As observed in these figures, the Tera model [Eqs. (8) and (9) in the Oconee report] provides higher estimates of PGA than the McGuire-Computer Science Corporation model [Eqs. (6) and (7) in the OPRA] for distances beyond ~15 km. Equations (8) and (9) are in general agreement with the best-estimate models, except for SHCP Model 27. Equations (6) and (7) provide estimates lower by as much as a factor of 1.80 to 2.0 at a distance of 100 km. The difference is less at shorter distances.

Figure 3.8 provides a different comparison. In this figure it can be seen that considerable variability can exist in the assessment of ground motion. In particular, the Oconee study incorporates only a part of this variability. A site-specific analysis, however, may not be expected to have as large a variability as a regional study, such as the SHCP. However, as shown in Figure 3.8 for magnitude 7.0, where the dispersion among the experts is great, the Oconee study accounts for only a fraction of the total variation.

We conclude from Figures 3.7 and 3.8 that the weighted average of the two Oconee models gives lower predictions of PGA, particularly for distances beyond 20 km. In comparison to the SHCP, overall lower hazard levels are estimated in the Oconee analysis. In our judgment, the effect on the mean hazard curve is considered to be small (i.e., less than a factor of 2) for accelerations less than 0.40g.

As pointed out in the hazard analysis report, the two-step process of developing an acceleration-MMI relationship results in greater uncertainty in the prediction of ground motion. This was taken into account by considering different logarithmic standard deviation values. We agree that it is appropriate to consider this added variability. The standard deviation values of 0.26, 0.31, and 0.39 seem reasonable, although no specific basis for the 0.39 estimate is given. Also, no basis is given for the probability weights assigned to the standard deviation values; however, they appear to be reasonable as well.

3.2.5.2 Upper-Bound Accelerations

In the Oconee analysis a limit on predicted ground acceleration levels was imposed by truncating the lognormal distribution about the median groundshaking estimate at the two standard deviation level. This approach to truncation establishes a limiting acceleration value as a function of the median acceleration estimate, which itself is a function of intensity and distance. Table 3.3 summarizes the truncation values on sustained acceleration for the two attenuation models, the maximum intensity and the logarithmic standard deviation. The truncation values are defined for distances less than six miles. No basis is given for the choice of a two standard deviation cutoff level, although it is commonly used.²⁰

In previous seismic PRAs,^{1,2} limits imposed on acceleration have been based on arguments that suggest limits on ground acceleration are dependent on the level of damage associated with the Modified Mercalli Intensity. This type of an approach attempts to use physical arguments to establish truncation values.

The limits on acceleration used in the OPRA for distances less than 6 miles are greater than or equal to the values used in previous studies. For distances greater than 6 miles, the truncation levels will be lower. However, a check of the hazard curves suggests that the truncation point closely tracks the value at distances less than six miles.

The reviewers believe that it is more appropriate not to truncate the hazard curves but to reflect a limit on damageability in the fragility curves; however, the effect of modifying the hazard curves produces the same result. Although we believe that limits on acceleration exist, it is difficult to quantify this belief at this time.

The effect on risk calculations of truncating hazard curves is generally considered to be small, since the accelerations that contribute to risk are usually less than the truncation limits. However, if higher accelerations become important to estimates of plant risk (i.e., in the 0.30 to 0.70g range) the effect could be greater, thus warranting further consideration. In Section 3.2.8, seismic hazard calculations were made to evaluate the effect of different truncation levels on the hazard curves.

3.2.6 Seismic Hazard/Fragility Interface

The process of establishing a compatibility between the hazard and fragility parts of the seismic analysis is an important interface task. To characterize the intensity of ground shaking at the Oconee site, a sustained peak ground acceleration was defined. The use of sustained acceleration is an attempt to characterize the potential of ground motion to damage structures and components.

Beginning with the Zion and Indian Point studies, a number of alternative approaches have been used to characterize the hazard/fragility interface. In our review of past seismic PRAs, a summary of hazard/fragility interface models and review comments is provided. $^{16-19}$ The reader is referred to the references cited for background on previous efforts in this area.

In the OPRA, a sustained peak acceleration is defined as being equal to 0.70 times the predicted peak ground acceleration. Although not stated as such, the sustained acceleration is equivalent to, or at least used as, a damage effective acceleration. To complete the ground motion characterization, a broad-band ground response spectrum was used.³⁰

As a general comment, the discussion in the hazard analysis report concerning the use of sustained acceleration and the 0.70 factor is limited. In fact, no scientific evidence is presented explicitly or by reference to support the approach used. This is a drawback of the report in that it is unclear whether an adequate basis exists to support the procedure and parameter estimates adopted.

To characterize the hazard/fragility interface, the interaction of a number of attributes of the seismic hazard and the components important to risk must be considered.^{14,31} For components these attributes are:

- o response frequency
- o energy absorption capability
- o damping characteristics
- o failure mode.

For seismic hazard the important factors are:

- o earthquake magnitude
- o strong motion duration
- o site-response spectrum characteristics.

Proper consideration of these factors in a seismic risk assessment is therefore site specific, in that the above factors vary from site to site. It is not clear that these factors were considered in the Oconee analysis.

The procedure used in the Oconee study to define an effective ground motion parameter is different from that used in any previous seismic PRAs we are aware of, except for the GESSAR-II study. In effect, however, the 0.70 factor used to scale a peak ground acceleration is similar to that used in the Zion and Indian Point studies, although the two procedures are different. Both procedures (Oconee and Zion and Indian Point), effectively characterize ground shaking in terms of 0.70*PGA and a broad-band ground response spectrum. Our comments in References 3 and 4 related to these studies are therefore appropriate here as well.

The state of the art suggests that the procedure used in the OPRA to model the hazard/fragility interface oversimplifies a complex interaction process.¹⁴ As discussed above, a number of factors related to seismic hazard and fragility must be considered. In fact, the damage potential of an earthquake is extremely dependent on the structures and components affected. In the case of nonductile, acceleration-sensitive equipment, the notion of a sustained acceleration is inappropriate, whereas for ductile structures, the damage effectiveness of ground shaking is very relevant.

In summary, the hazard/fragility interface approach used in the OPRA is an oversimplification of the interaction process (even over the Zion and Indian Point studies). As we concluded in our review of the Zion and Indian Point studies, the ground acceleration level used to scale a broad-band response spectrum shape is probably conservative for structures or equipment which have ductility; however, for acceleration-sensitive items, it provides an unconservative characterization of the damage potential of ground shaking. We expect, however, that the overall effect of variations in the hazard/fragility interface on the risk calculations would be small.

3.2.7 Comparison with USGS Results

In this section a qualitative comparison is presented between the results of the Oconee seismic hazard analysis and the USGS hazard assessment conducted for the contiguous U.S. (Comparison of SHCP seismicity parameter estimates were made in Section 3.2.4.) At this time, SHCP seismic hazard calculations are not available for the Oconee site. In this section a comparison is made with USGS study results, including the seismic hazard estimates for the Oconee site.

In Figure 3.3 the USGS source areas in the vicinity of the Oconee plant are shown as defined in their 1976 study. The USGS zonation is considerably different from that used in the Oconee study. In fact they parallel closely the source areas suggested by Bollinger²² (see Figure 3.6). In Table 3.4, the seismicity parameters for source areas 64 and 65 are reproduced from Reference 21. Recall that the 1976 study was completed before the Oconee analysis.

In Figure 3.9, the USGS source areas from their 1982 study are shown. The seismicity parameters for source areas 99, 100, and 101 in this study are also reported in Table 3.4.

A comparison between the USGS source areas and seismicity parameters leads to a number of observations. First, the USGS Charleston zone, in both studies, is in much greater proximity to the plant site than its counterpart in the Oconee study. As a result, maximum intensity X earthquakes used in all three studies (two USGS studies and the Oconee analysis) can occur considerably closer to the plant in the USGS analyses, thus resulting in higher expected ground motions.

Richter b-values estimated in the Oconee study are quite close to 1982 USGS analysis, 0.48 and 0.50, respectively. However, the 1976 USGS study estimate of b was 0.33. This value would tend to estimate higher frequencies of occurrence for larger magnitude earthquakes.

Figure 3.10 compares the USGS 1976 and 1982 hazard curves with the mean Oconee hazard curve. It is difficult to make a true qualitative comparison between these studies since the USGS studies do not account for the variability in ground motion. If they had, their results would be higher.

The USGS hazard curve (1976) is higher by a factor of about 2 than the Oconee mean hazard curve estimated in the Oconee study. However, if the variability in ground motion was incorporated in the USGS calculation, a greater difference would be expected.

3.2.8 Sensitivity Calculations

As part of the review of the seismic hazard analysis, a limited sensitivity study was performed to verify the results reported in the Oconee report and to evaluate the sensitivity of the hazard assessment to different parameter assumptions. The results of these calculations are reported below.

From information in the Oconee report and discussions with the author of the analysis, data input to a standard seismic hazard analysis routine were prepared.³² As a first step, calculations were performed in an attempt to verify the results reported in the Oconee study. The following two cases were used as a basis for comparison:

0	Case 1 -	Attenuation model		Eqs. (6) &	(7)	(OPRA)
		Logarithmic Standard Deviation	-	0.26		
		Maximum Earthquake	-	Hypothesis	2	
		(See Table 1 of the hazard report)				

o Case 2 - Attenuation model - Eqs. (8) & (9) (OPRA) Other factors the same as for Case 1

The results of our calculations and those reported in the OPRA are shown in Figure 3.11 in terms of sustained ground acceleration. From the figure it can be seen that beyond 0.40g, there are differences of greater than a factor of 2 between the two estimates for case 1 (i.e., OPRA and this study). For case 2, there is a factor of 2 difference at 0.20g's. Note that the Oconee hazard estimates are higher.

These findings were discussed with the author of the Oconee hazard analysis. In this discussion we first confirmed that the input to our calculations were correct. It is unclear, from the results reported in Figure 3.11 and our discussions with the analysts involved in the Oconee study, exactly what truncation level was used in the Oconee calculations. We understand that possibly a truncation at greater than two standard deviations was used, although this could not be confirmed. Secondly, it was learned that the seismic hazard results reported in Table 3 of the Oconee hazard report are smoothed values. The ill effects of this smoothing, which was greatest near the truncation limit, are apparent in case 1. From the results of our hazard computations, we find that only the Piedmont and Upper Coastal Plain source area contribute to the results for accelerations greater than 0.50g. Therefore, the truncation of the hazard curves is limited by the maximum intensity in this source area (i.e., MMI = VIII). From Table 3.3, the limit on acceleration is 0.681g; therefore, beyond 0.681g the frequency of exceedance should be zero. The Oconee hazard curve at 0.70g has a frequency of exceedance of 2.8E-7, which is incorrect.

To reconcile our calculations with the Oconee results for cases 1 and 2, a three standard deviation truncation level was used. Figure 3.12 shows this comparison. For accelerations <0.70g, the results are reasonably close. Above 0.70g, the results are not as favorable since the truncation level is higher for a three standard deviation cutoff.

Similar comparisons apply for other standard deviation levels and maximum earthquake hypothe ~s. From these results we conclude that a truncation level higher than two st lard deviations may have been used in the Oconee hazard analysis. On balance, the hazard curves reported are conservative by as much as a factor of 2 at low accelerations. Greater differences exist at accelerations above 0.40g. In Figure 3.13, a much higher truncation level of five standard deviations is considered for cases 1 and 2 and are compared to the Oconee results. Using a higher truncation limit produces hazard curves two to three times higher for case 1 and less for case 2. This suggests that for accelerations less than 0.40g, the effect on the hazard results of applying a truncation on acceleration is small.

As part of our review, it was stated that the maximum intensity for the Piedmont and Upper Coastal Plain source area for all three hypotheses may be underestimated by one intensity unit. Figures 3.14 to 3.16 compare the effect of increasing the maximum intensity for cases 1 and 2, one unit for each hypothesis on the maximum size earthquake. For the Piedmont and Upper Coastal Plain source, the following revised estimates are made:

Hypothesis	Maximum Earthquake	Figure		
1	¥III	3.14		
2	IX	3.15		
3	X	3.16		

For comparison, the hazard curves with a three standard deviation truncation are used. From each figure we see that for accelerations less than 0.30g, there is at least a factor of 2 to 3 increase in the hazard estimate corresponding to a one unit increase in the maximum intensity.

For accelerations beyond 0.30g, the difference increases to as much as a factor of 10 (see Figure 3.14). The largest differences are realized for hypothesis 1, while for hypotheses 2 and 3 the largest variations are less than a factor of 5.

From these sensitivity calculations we conclude the following. As provided in the Oconee study, the seismic hazard curves are conservative, for the parameter values stated in the report. At accelerations less than about 0.40g, the Oconee seismic hazard curves were as much a factor of 5 higher. We concluded from our discussions with the author, and our own calculations, that a higher truncation level (i.e., three rather than two standard deviations) may have been used. The effect of using a three standard deviation truncation as opposed to a much higher truncation level is small.

If the maximum intensity estimates for the Piedmont and Upper Coastal Plain source are increased by one intensity unit, the calculated seismic hazard at Oconee is increased. According to our calculations, there is at least a small effect (i.e., a factor of 2 or greater) on the frequency of exceedance. The effect varies for each hypothesis on maximum-size earthquake. Overall, we judge that increasing the maximum intensities estimated for the Piedmont and Upper Coastal Plain source will result in a moderate (i.e., a factor of 3 to 10) increase in the seismic hazard.

3.2.9 Summary

Comments raised in the review of the seismic hazard analysis are summarized. In general, the seismic hazard report provides inadequate documentation for all parts of the analysis. In addition, we judge that important parts of the analysis are incomplete, such as the delineation of seismic sources. The following summarizes our major comments.

- The methodology used to evaluate the frequency of exceedance is adequate and appropriate to characterize the seismic hazard at the Ocone site.
- o Throughout the seismic hazard analysis, an ad hoc approach was used to make subjective assessments. Annex Jl is not, in our opinion, a tractable presentation of the information used in the analysis or of the process by which the model uncertainty was quantified.
- Consideration of only one seismotectonic model is believed to be inappropriate and probably unconservative.
- o The seismotectonic model used in the analysis is reasonable.
- The maximum intensity for the Piedmont source area is believed to be underestimated by one intensity unit. According to sensitivity calculations, this results in a moderate increase in the seismic hazard (i.e., a factor of 2 to 10).
- o An independent attempt to verify the Oconee seismic hazard curves suggests that possibly a three standard deviation truncation was used in the analysis, rather than the two standard deviation level stated in the report.
- Sensitivity calculations indicate that truncation of the hazard curves at a much higher truncation level (i.e., higher than three standard deviations) has a small effect on the results.
- Comparisons with the SHCP show that the attenuation models used in the Oconee study provide lower estimates of the seismic hazard. Overall, we judge this would have a small effect on the hazard estimates (<0.40g).
- The estimates of seismicity parameters, activity rates and b-values, are in reasonable agreement with estimates from other studies.
- o In our opinion, the uncertainty in the frequency of exceedance is underestimated, leading to an unconservative estimate of the site hazard. If a complete family of seismotectonic models and attenuation relationships is included in the analysis, we estimate there will be a moderate increase in the frequency of exceedance (i.e., a factor of 3 to 10).
3.3 Seismic Fragility Analysis

3.3.1 Introduction

A preliminary review of the seismic fragility data for structures and equipment used in the OPRA was conducted. The fragility analysis is contained in Annex J2 of Appendix J, entitled: "Conditional Probabilities of Seismic Induced Failures for Structures and Components for Oconee Generating Station Unit 3." The fragility data used in Section 9.1 of the OPRA were also reviewed. In addition to the report sections, a copy of the fragility calculations was obtained and reviewed, and a two-day plant visit was made to inspect the Oconee buildings and equipment.

The adequacy of the fragility calculations is discussed in this section. A sensitivity analysis was conducted to investigate the influence of different fragility assumptions for the significant contributors and is presented in Section 3.4. The plant background information and the fragility methodology used in the OPRA is discussed in the next section, followed by general comments and then specific comments on the significant contributors to the analysis. Conclusions based on the fragility analysis review are given in the last section.

Note that in the review of the fragility analysis all capacities are peak ground acceleration capacities unless otherwise stated.

3.3.2 Plant Background and Fragility Methodology

The Oconee nuclear station was designed in the late 1960s and thus was analyzed for pre-1973 seismic requirements. The plant was designed for a maximum hypothetical earthquake (MHE) of 0.1g (similar to the current safe shutdown earthquake) and a design earthquake of 0.05g (similar to the current operating basis earthquake). The Oconee plant is founded on rock. All structures are located on rock except for parts of the turbine building and the borated water storage tank. Class I structures founded on overburden were designed for a 0.15g MHE. The seismic response spectra used in the structural design were the Housner spectra. Both the effects of the MHE and a loss-of-coolant accident (LOCA) were considered simultaneously in the plant design.

One exception is the safe shutdown facility (SSF) which was constructed post-1973 and is designed for current USNRC criteria. It is founded on rock, and it was designed for an SSE of 0.10g.

The Zion method¹³ was adapted to develop the fragility information used in the OPRA. It was assumed in the fragility analysis method that the underlying response and capacity parameters are related in a multiplicative manner, which leads to the assumption of the lognormal model. Variability is separated into randomness and uncertainty components where randomness is inherent in the model and cannot be reduced by any additional analysis or testing and is largely attributable to the randomness in the ground motion "signature." In contrast, uncertainty is the portion of variability which can be reduced by additional work (e.g., analyses and testing). Uncertainty is characterized by a lognormal distribution on the median capacity. Thus both the model for the median capacity and the model for the capacity conditional on knowing the median value are assumed to be lognormal. The parameters of the lognormal fragility model consist of a median capacity, A (i.e., the median of the medians) and logarithmic standard deviations B_{Γ} and B_{U} for randomness and uncertainty, respectively.

The method used for characterizing structure fragility in the OPRA has two important features. First, it is based on a double lognormal distribution model as described above. Second, the probabilistic analysis uses the results of the original design analysis as the basis for developing the seismic fragility information. The median fragility values were obtained using the response and capacities from the design analyses which were scaled to eliminate conservatisms. The estimated variabilities, which include randomness and uncertainty, were based on some data, but mostly on engineering judgment.

In the Zion method, the fragility analysis for a structure starts with the design capacity usually corresponding to the MHE design. Then the factors of safety are systematically factored out for the following response and capacity factors in order to develop the median capacity:

- o Spectral shape
 - o Soil-structure interaction
 - o Frequency
 - o Mode shape
 - o Damping
 - o Modal combination
 - o Combination of earthquake components
 - o Strength
 - o Inelastic energy absorption

Because of the assumed underlying lognormal model, median factors of safety are simply multiplied together and the result is used to scale the design ground motion capacity. Logarithmic standard deviations obtained for each factor are based on a first-order Taylor series expansion of the relationships between response or capacity and the underlying physical variables. It is generally assumed that the factors are independent and the corresponding logarithmic standard deviations can be combined by the square-root-of-the-sumof-the-squares (SRSS) procedure.

The methodology for characterizing the equipment fragility uses the same lognormal distribution model as assumed for structures. However, in addition to some of the structure response factors which affect the equipment fragility, a corresponding set of equipment response factors are considered. In total, the effects of the following three categories of factors are included in the equipment fragility analysis:

- o Equipment capacity
- o Equipment response
- o Structure response

More complete description of the characteristics of the Zion method can be found in References 13 and 14.

3.3.3 General Comments on the Seismic Fragility Analysis

The OPRA fragility calculations were performed back in 1980 to 1981 immediately after the Zion and Indian Point PRAs. The approach and quality of the calculations for the OPRA are similar to these first PRAs. No evidence was found in either the OPRA or supporting calculations to indicate that an iterative fragility analysis was performed in which the capacities of the more significant components were refined as their importance was discovered during the course of the systems analysis. Apparently, a single fragility analysis was conducted with emphasis distributed equally among all the components.

As discussed in Section 3.3.4, generic fragility values were developed for many of the equipment items. This approach is particularly conservative for Oconee where the capacity factors were based on the assumption that the design earthquake, referred to in the OPRA as the Operating Basic Earthquake (OBE), caused stresses in a critical element close to the code capacity. Since the OBE is only 0.05g, the resulting generic equipment capacities are conservatively low when compared to historical earthquake data for heavy industrial facilities. This observation is discussed for specific components in Section 3.3.4.

One area receiving particular attention in the review was the computed capacities of the block walls which appear to be relatively low (i.e., -0.3g). The mean frequency of core melt from the OPRA is 6.3E-5 per year. If the block walls are eliminated from the analysis, the mean frequency of core melt drops to 5.4E-5/yr (compared to 7.5E-5/yr obtained in this review). These sensitivity analyses are discussed in more detail in Section 3.4.

Block wall tests are currently being conducted by Duke Power Company on wall configurations similar to block walls at the Oconee plant. As part of the OPRA review, one set of these tests was observed, which confirmed that an arching mode of resistance would occur and greatly increase the strength of the walls above the capacities assumed in the OPRA.

Several important issues raised in past seismic PRA reviews are also pertinent to the OPRA. These include the question of relay chatter, dependency, failure of secondary components, and design and construction errors. Each of these concerns is discussed below.

It is assumed in the OPRA that relay chatter is fully recoverable and thus was not included in the systems analysis. Various median values have been used in past seismic PRAs and range as low as 0.41g as reported in the Seabrook PRA.²² The recently published Handbook for Nuclear Power Plant Seismic Fragilities gives relay chatter fragility values.³³ The spectral acceleration median capacities for 5% damping in the 5- to 10-Hz frequency

range are 5.67g and 2.59g from expert opinion and test data, respectively. The combined median capacity based on a statistical procedure described in Reference 33 is 1.66g with a logarithmic standard deviation of 1.51. Note that the median spectral acceleration capacities must be reduced by a factor of 2 to 5 to obtain the equivalent peak ground acceleration capacity because of dynamic amplification in both the electrical cabinets and the building structure.

It is generally accepted that relays will chatter at low ground motion acceleration levels. The following statement made in the OPRA in Annex J2 (page J-5-125) supports this conclusion:

"Consequently, chatter in large relays and contactors can be expected at seismic levels above the MHE."

The important questions become:

- Which relays could chatter and cause breakers to trip or cause other electrical maifunctions that would affect the safe operation of the plant?
- Can the plant operators reliably reset the relays in a timely manner after the earthquake?

Currently, relay chatter is being investigated. Some systems engineers believe that as few as 10 to 30 relays are critical, and if they can be determined then they could be replaced with rugged units that will not chatter. In Section 3.4, the implications of including relay chatter in the OPRA are discussed.

The issue concerning dependence affects both systems in parallel and systems in series. The OPRA and most past seismic PRAs conservatiely assume that identical components in parallel are considered to be a single component. For components in series, previous practices have conservatively modeled them as independent. However, in the case of piping, the piping segments are connected in series; thus, the frequency of failures for a piping system may not be conservatively represented by the frequency of the weakest component, unless the capacities and responses of all segments are individually (i.e., capacity with capacity and response with response) perfectly correlated or unless the capacity is dominated by a single weakest component.

Because piping often extends a relatively long distance and is supported at many places in a structure, piping response will not be perfectly correlated. Also, because different components may come from different manufacturers or material runs, capacity also is not perfectly carrelated. A similar problem also exists for the electric cables supported by cable trays. Because the capacities for piping are conservatively low (see Section 3.3.4), it is unlikely that the concern for dependence is significant; however, if the capacities are revised upward, the contribution due to dependence should be reconsidered.

Failure of secondary components which could fall and affect safetyrelated components is an important consideration in a seismic PRA because of the common cause effect of seismic motions. The fragility report gives no indication that this issue was addressed in a systematic manner, although it obviously was considered to some degree, since block wall failure was included in the analysis and is an example of such a problem. During the plant walkdown, no obvious concerns were observed. However, because capacities in seismic PRAs are generally high, the potential for contribution from nonsafety-related components becomes greater. The OPRA seismic capacities are generally conservatively low which tends to compensate for any secondary failures that may have been missed.

Design and construction errors were not considered systemically in the OPRA. A troublesome concern is the quality of information generated during the plant design which is used as a basis for developing the structure and equipment fragilities. The calculations for the original design were not checked, except for the auxiliary building where a new dynamic model was developed and used in the building analysis. Unfortunately, no floor response spectra were computed to benchmark the extrapolations which were required in the equipment fragility analysis for the auxiliary building earthquake response factors, since response spectra for only up to 2% damping were available. In general, the analysis and conditions of the plant were accepted on faith.

The above concerns have been discussed in past PRA reviews³,⁴,¹⁶⁻¹⁹ and also in the PRA procedures guide, which it is hoped will provide a frame-work for incorporating these issues in future PRAs.¹⁴

3.3.4. Review of Significant Components

The significant components in the OPRA were reviewed to determine the reasonableness of the fragility parameter values used in the analysis. The 34 components given in Table 9.6 of the OPRA were selected for detailed review. These components were used in the final seismic sequence logic. In addition, the capacities of other components were considered, and the basis for eliminating them from the final fault trees was evaluated. From this review, seven additional components were added to the detailed review list. Table 3.5 gives the 41 components which were considered in more detail in the review.

The basis of the review included Appendix J, Annex J2, in the OPRA entitled "Conditional Probabilities of Seismic Induced Failures for Structures and Components for Oconee Generating Station Unit 3," calculations performed in support of Annex J2, results from past PRAs, earthquake experience at industrial facilities, and the engineering experience and judgment of the reviewers.

First, comments concerning the review of the fragility calculations are given, which apply generally to many of the components. Comments for the specific components in Table 3.5 are presented next. Additional comments are made concerning safety-related components with relatively high capacities which were excluded from the analysis. These comments are given in Section 3.4 as part of the sensitivity analysis.

General Comments: Half the calculation file, which is approximately 4 inches thick, is devoted to the analysis of major structures (i.e., reactor, auxiliary, and turbine buildings, safe shutdown facility, intake structure, and borated water storage tank). The remainder of the calculations were for the equipment.

In general, the calculations for the equipment were poorly organized and difficult to follow. Many inconsistencies were found, which were more bothersome than substantive. For some components, the calculations were not found at all, although they may be concealed in some hidden corner of the calculation file. Part of the difficulty is related to the fact that many of the fragility factors (i.e., structure response, equipment response, and equipment capacity) were developed generically for different groups of equipment.

The floor response spectra used in the fragility analysis were extrapolated from lower damping spectra or spectra from other buildings. Thus, the reasonableness of the equipment response shape factors was difficult to determine, partly because the response spectral ordinates were scaled from plots of spectra and, apparently, the scaling process was done multiple times even for the same location. This resulted in slightly different values for equipment at the same location, which made it difficult to systematically find common logical threads throughout the equipment fragility analysis.

As explained above this was further complicated by the lack of response spectra at certain locations or for higher damping values, which required extrapolations of spectra from other buildings and even from other seismic response studies. In summary, it was nearly impossible to verify that the response spectra shape factors for equipment were rationally developed. Because the strength of the buildings at Oconee is high relative to the enclosed equipment, elastic building response spectra are relevant to the equipment response and must be carefully used.

At failure, median damping for equipment is assumed to be 5%. However. for the reactor building only 0.5% damping spectra are available from the plant design files. For the auxiliary building, 0.5%, 1.0%, and 2% are available and, finally, for the turbine building, no floor response spectra were claimed to be available. However, in Reference 34, which was the primary basis for the floor response spectra, response spectra in the turbine building at elevation 796 feet 6 inches and 822 feet are given for 1/2% to 2% damping in two directions.

In the fragility analysis, response spectra were extrapolated between buildings (e.g., auxiliary building to the turbine building), and factors of safety were developed to adjust lower damping spectra to spectra corresponding to median (5%) damping. These factors are very sensitive to the building configuration, height above ground of the floor under consideration, and the frequency of the equipment being analyzed. It is almost certain that if new analyses had been conducted to derive the floor response spectra directly, many of the response spectra shape factors for equipment with fundamental frequencies in the 5- to 10-Hz range would be significantly different.

It was noted above that a new dynamic model was developed for the auxiliary building for the structure fragility analysis. It would have been relatively straightforward to develop realistic median-centered floor response spectra directly, which would have been significantly more certain than the spectral ordinates based on questionable extrapolations and adjustment factors. About 40% (i.e., 16 out of 41) of the component fragilities listed in Table 3.5 were based on a generic analysis for the capacity factor. The generic analysis is described on pages J-5-116 through J-5-119 of Annex J2. Four categories of generic equipment were considered which consisted of crossing light and heavy cases with rigid and flexible cases. It was assumed that the equipment supports are the critical element and that combined normal plus OBE loading ranges from 0.2 to 1.1 times the design load. This approach has been used in other seismic PRAs; however, for the OPRA, the referenced OBE maximum ground acceleration is only 0.05g. Thus the resulting factor times the OBE acceleration results in very low capacities.

The anchorage systems at Oconee did not appear to be significantly weaker by comparison with similar anchorage systems for other eastern U.S. nuclear power plants which have twice the seismic input. This is particularly true for active equipment such as pumps, diesel generators, etc., which generally have more severe normal operational loads. It is likely that the anchor bolts and other support hardware were selected for other reasons and checked for seismic requirements. In general, the capacities of components developed from generic assumptions on the design stress level appear to be conservatively low in the OPRA.

One measure of the reasonableness of the fragility parameter values is the high-confidence low-probability of failure (HCLPF) values.³⁵ The HCLPF value is defined in Reference 35 to be simply the earthquake level at which it is extremely unlikely that a component will fail. From the mathematical perspective of the lognormal model used in PRA calculations, the HCLPF is defined to be approximately equal to a 95% probability of not exceeding about a 5% fraction of occurrence. Table 3.5 lists the HCLPF values for the 41 components that were reviewed. Note that these values have been modified by a factor of 1.23 to be consistent with values reported in Reference 35, where all capacities are referenced to a common peak ground acceleration parameter. This was necessary since in some seismic PRAs an effective peak ground acceleration parameter was used while a ceak ground acceleration, the component capacities have been increased. Many of the HCLPF values from the OPRA appear to be unreasonably low from the standpoint of historical earthquake experience at industrial facilities and engineering judgment.

Offsite - Power Insulators(1) and 100-kV Lee Feeder(2): Both these components are located on power equipment outside the plant where the limiting capacity is due to brittle ceramic insulators. A review of insulator failure in six major earthquakes ranging from 0.11g to 0.4g was used to obtain the median capacity of 0.20g. The median capacity is low, so that offsite power is essentially lost at acceleration levels corresponding to core melt. This is truer for other PRAs where the capacities of other components are much higher. In the case of Oconee, other critical components have capacities only slightly larger (e.g., auxiliary building masonry walls with 0.28g median capacity, which is a single after conceding loss of offsite power).

In summary, the median capacity used for the ceramic insulators is reasonable,

Condenser(3): This is a large component which is apparently supported by four $5-ft^2$ concrete pedestals anchored to the turbine building basement slab. In

the calculations for the condenser seismic capacity, the following failure modes were considered:

- Shear failure of the weld between the bottom plate of the condenser and the pedestal anchor plate.
- Bending failure of the embedded (in the pedestal) shear plates and anchor bolts.
- Pedestal failure (i.e., failure of the anchors and shear plate at top of pedestal).

The calculated median capacity of the condenser is reported to be 0.21g. The true capacity is probably much larger. Even if there was no connection between the condenser and the pedestals, the seismic inertial forces would still be resisted by sliding friction with a coefficient of friction at least equal to 0.4 to 0.5. Assuming that the pedestals do not fail as a unit because of shear or moment effects, the median capacity of the condenser could approach 0.5g.

Details for the pedestal were not available and hence were not checked. In reality the failure modes listed above that were checked contribute to increasing the coefficient of sliding friction and hence the overall capacity of the condenser. Even if the elements considered in the analysis fail, sliding friction will resist movement of the condenser. In order to obtain a more accurate estimate of the condenser capacity, the capacity of the pedestals must be determined; however, it is likely that the median capacity is significantly larger than 0.21g. In addition, the HCLPF value of 0.16g is not reasonable for heavy rigid-equipment-based earthquake experience.

Note that the fragility data for the condenser are not given in Table 5.5 of Annex J2. The date on the condenser calculations occurred after the date of Annex J1. Evidently the condenser analysis was performed after Annex J2 was published.

600/208-V Transformer(4): This transformer is a rigid passive component located in the auxiliary building at elevation 796 feet. Generic capacity factors were used for this component in the OPRA. The median equipment response factor is 0.81, and except for the letdown cooler and the auxiliary power and station transformers is the only other component with this factor being less than 1.0. Generally, this factor is greater than 1.0, primarily because of conservatism in damping and use of the El Centro earthquake time history which is the basis for the design floor response spectra development. The ground response spectra for the El Centro time history is higher (i.e., more conservative) than the design ground response spectrum.

The structure response factor for this transformer is 0.89 and is consistent with other components. In general, the structure response factor includes the unconservatism between the median ground response spectra assumed for the site and the design response spectrum. This factor is generally less than 1.0. In contrast, the equipment response factor is generally greater than 1.0 as discussed above. Because this component is considered to be rigid, it may have been erroneously assumed that the effect of the difference between the El Centro earthquake and the design spectra should not be included in the equipment response factor. The following is stated on page J-5-138 of Annex J2 for the pressurizer analysis (another rigid passive component):

"The spectral shape factor, previously stated to be based upon using the El Centro time history instead of an artificial time history generated from the site specific spectra, is also eliminated (i.e., equal to 1.0) since both earthquake descriptions were anchored to the same peak ground acceleration and the pressurizer support structure is considered to be rigid."

Since a rigid component in a building is sensitive to the peak floor acceleration, the difference in the ground response spectra (i.e., El Centro vs design) at the frequency of the building (not the frequency of the equipment) should be used to determine the equipment spectral shape factor. For the pressurizer, the fundamental frequency of the reactor building is 3.5 Hz and the factor is roughly 1.2.

For the 600/208-V transformer, the auxiliary building fundamental frequency is 9.4 Hz and the spectral shape factor is close to 1.0. However, the median response spectrum and the design response spectrum are also close; thus the structure response factor also should be close to unity (it is 0.89 in Table 5.5 of Annex J2). In total, the product of the three factors (i.e., structure response, equipment response, and equipment capacity) and the OBE acceleration of 0.05g is 0.27g, which appears to be low.

More important is the capacity of the actual component rather than a generic equivalent. During the plant tour the 600/208-V transformer was inspected. It is approximately 18 by 24 in. in plan and 3 ft high, and is anchored to the floor slab by four bolts, which are at least 5/8 in. in diameter. It is likely that the median capacity for this component is over 1g and probably much higher.

Auxiliary Building Block Walls(5): The block walls represented by component 5 are located throughout the auxiliary building. The calculations for the capacity of the block walls were reviewed in detail. It was assumed in the analysis that the walls are hollow and constructed with type-N mortar with a median tensile capacity of 20 psi. This is a design value, and judging from tests conducted for Duke Power Company this value is on the low side.

The failure history assumed in the fragility analysis was postulated to consist of the following three stages:

- Cracking at the bottom (the bottom of the wall and two sides were assumed to be fixed and the top pinned).
- Cracking at the middle of the wall (the bottom and top were assumed to be pinned).
- Rigid body rocking at the top and bottom wall sections (i.e., above and below the crack in the middle of the wall) leading to the final failure.

It was found in the analysis that the highest capacity corresponded to the first stage (i.e., cracking at the bottom) and the other two stages would follow directly.

The possibility of a higher capacity corresponding to arching action was not considered. During inspection of the plant by the reviewers, the condition of the block walls was examined. The walls appeared to be tightly constrained by the side supports and top. No noticeable cracks or spaces were observed at the top of the walls (the walls were spot checked, but not systematically inspected). It appears that arching action is a realistic mode of resistance. An approximate arching-action analysis suggests that the actual block wall capacities may be many times larger than assumed in the OPRA.

Recent block tests being calculated for Duke Power Company also indicate that the capacities are larger and that arching action does occur. Dr. John Reed observed the last series of tests of a block wall panel which was constructed with an opening. The panel was subjected to multiple motions over three and one-half times higher than the design input. There was only minor cracking and the wall definitely developed an arching mode of resistance. The median capacity of 0.28g used in the OPRA is lower than the input level used in the block wall tests.

In conclusion, the block wall capacity used in the OPRA is conservatively low and the walls which are constrained between the floor and the concrete above have median capacities which are generally greater than 1g (also see comments for component 9 below).

The variability assumed for the block wall capacities used in the OPRA is also low. By not considering the possibility of arching action, both the median capacity and uncertainty logarithmic standard deviation are on the low side. In addition the calculated variabilities for the strength factor are incorrect. It was assumed that the standard deviation, σ , for a lognormal distribution is related to the logarithmic standard deviation, β , as follows:

$$r = m (1 - e^{-\beta}),$$
 (1)

where m is the median capacity. However, the correct relationship is as follows:

 $\sigma = m e^{\beta^2/2} \sqrt{e^{\beta^2} - 1} .$ (2)

The values calculated by Eq. (1) are low by as much as a factor of 2 for typical β values. For example, the final β_U value for the strength factor using Eq. (1) for block walls at elevation 838 ft is 0.23 compared to the more correct value of 0.41 using Eq. (2).

Transformer CT3(6), Transformer CT5(7), and Transformer CT4(8): These three transformers are large components located at the ground level. Transformer CT4 is enclosed in the blockhouse. The capacities of CT3 and CT5 were based on generic capacity values for rigid components corresponding to support failure. Transformer CT5 was observed during the site inspection and was found to have ceramic insulators on top. In addition, the oil coolers probably are not rigid. However, the resulting median capacity of 0.30g is reasonable.

The capacity of transformer CT4 was determined on the basis of a specific analysis of its supports. This transformer may also have ceramic insulators and definitely has flexible oil coolers. The median capacity of 0.31g also appears to be reasonable.

Transformer CT4 Blockhouse(9): The controlling element is the block wall which separates the CT4 transformer from the switchgear. It is constructed of 12-in.-wide unreinforced block and is 28 ft high. Two intermediate steel wide flange sections spanning vertically at the one third points strengthen the wall. As in the analyses performed for the auxiliary building block walls (see discussion for component 5 above), it was assumed in the fragility analysis that failure is governed by either cracking or rigid body rocking. Again, the potential capacity of arching action was not considered.

We learned in a discussion with Duke personnel that the wall is constructed up to and tightly against the roof of the blockhouse. However, conventional flexural capacity has been used to verify that this wall can resist the MHE in response to USNRC IE Bulletin 80-11. Because of its high flexural capacity, arching action was not used as a licensing defense.

An approximate arching analysis indicates that the median capacity of the wall is greater than 0.5g. Hence the 0.31g median capacity used in the OPRA appears to be low.

HPSW Elevated Storage Tank(10): This is a 100,000-gal elevated storage tank located on a small rise just west of the plant. No analysis for this tank was found in the calculation file. Table 5.5 in Annex J2 indicates that the median capacity of 0.35g was based on an engineering estimate, without a specific calculation.

Drawings for the tank are available from Duke Power Company, and a specific fragility analysis should have been performed for this component. Although the capacity is consistent with past earthquake experience (i.e., tanks often have low seismic capacity), the capacity of this tank may be higher.

<u>4160/600-V Transformer(11)</u>: The 4160/600-V transformer is located in the turbine building at elevation 795 feet which is the mezzanine level. This is a rigid passive component and the fragility parameters were estimated on the basis of engineering judgment. An analysis probably was not performed since floor response spectra were apparently unavailable for the turbine building (see general comments section).

The median capacity of 0.40g for component 11 is contrasted to the median capacity >2.00g for the 4160/600-V transformer located in the safe shutdown facility (SSF). The latter capacity is based on generic test data. Although Table 5.5 in Annex J2 indicates that the failure mode for component 11 is relay chatter (transformers per se do not have relays), it is believed that the capacity is really a structural failure mode.

The transformer in the SSF was inspected during the plant visit. It is anchored by welding to plates embedded in the floor slab at four locations. Although the anchorage in the turbine may not be as rugged as the support in the SSF, it is likely that the median capacity of component 11 is larger than 0.40g. The HCLPF value given in Table 3.5 is only 0.18g, which is unreasonably low.

Feedwater Heaters(12): The supports for the feedwater heaters are located at elevation 775 feet (basement) in the turbine building. These components are tail, slender, vertical tanks which are each supported on four steel pipe columns anchored to the floor slab. As with other components in the turbine building, the capacity was estimated on the basis of judgment. No horizontal support was observed during the plant inspection at the top of the tanks. Each tank is confined to deflect approximately 6 inches on each side by the floor slabs through which it passes. Without further analysis, the median capacity of 0.40g assumed in the OPRA appears to be reasonable although on the low side compared with vertical heat exchangers from past seismic PRA studies.

Jocassee Dam(13) and Keowee Dam (east intake)(15):

A qualitative review of the Jocassee and Keowee dams has been performed from the engineering point of view by the Structural Analysis Division of BNL. Certain difficulties in evaluating the strength and reliability of these flood-control structures have been observed and identified in Appendix E. In the reviewers' opinion, the median capacity of these structures is underestimated; however, because of the limitations of this review no reevaluation was performed in the results presented. In Section 3.4, the same fragilities as those generated by the OPRA were used.

High Voltage Bus Ducting(14): The specific high voltage bus ducting as presented by component 14 is not known, since no information was found in the calculation file. From Table 5.5 in Annex J2, it is indicated that the bus ducting is located at the ground level. The equipment response and capacity factors are different from any other component listed in Table 5.5, which suggests that a specific analysis was conducted for this component.

The median capacity of 0.56g is low enough that it is unlikely that the ducting per se has a lower capacity; however, its capacity may be controlled by relative displacements of the equipment to which it is attached. It is likely that the capacity used in the OPRA for this component is conservative.

Upper Surge Tank(16): This tank is located at elevation 838 feet in the turbine building on a platform above and to the side of the turbine deck. Table 5.5 of Annex J2 indicates that this component is rigid, and thus generic capacity factors for rigid equipment were used. In fact, each of the two horizontal surge tanks is supported by four steel columns and the vertical tank in the middle is located high off the floor also on four braced columns. All three tanks and supporting structure are flexible, hence the higher flexible equipment capacity factors should have been used. Although it is not clear that these tanks were designed for seismic forces, past experience in industrial facilities indicates that tanks of this type when anchored can resist earthquakes up to at least 0.5g. Thus the median capacity of 0.62g is consistent but probably on the conservative side.

<u>Pressurizer Supports(17)</u>: The pressurizer is a rigid component located at elevation 813 feet in the containment. There is considerable confusion in the fragility report and in the calculation file for the pressurizer. The median

capacity factor given in Table 5.5 of Annex J2 is 5.30; however, the calculations give a value of 2.36 corresponding to the capacity of the anchor bolts. This calculation is dominated by the assumption that the design loads correspond to 80% of the yield capacity. This is likely to be conservative; hence, the 5.30 value for the capacity factor appears to be more realistic.

The calculations for the equipment response factor are even more confusing. On page J-5-139 of the fragility report the product of the qualification method, modeling error, and earthquake component combination factor is 1.55, but the corresponding value in Table 5.5 is 1.24. Actually, there are computations leading to both factors at two different places in the calculation file: at one place the floor zero peak acceleration (ZPA) is 0.12g while at the other it is 0.15g. This accounts for the difference between 1.55 and 1.24. At one other place in the calculation file the ZPA value is 0.135g. It is guessed that the values were scaled off a response spectrum plot at three different times.

As discussed above for the 600/208-V transformer (component 4), the spectral shape factor for the difference between the El Centro spectrum and the design spectrum for both component 4 and the pressurizer has been erroneously eliminated. This factor is approximately 1.2 for the pressurizer. In conclusion, it is likely that the median capacity of the pressurizer is higher. If the more rational factors are used, a median capacity of 0.9g is found which is higher than the value of 0.62g used in the OPRA.

One glaring error between Table 5.5 in Annex J-2 and Table 9.2 in Chapter 9 of the OPRA is the $B_{\rm T}$ value used for component 17. In Table 5.5, the value is given as 0.20, but in Table 9.2 it is shown as 0.70. It is likely that the incorrect value was used in the core melt fraquency analysis since the example on page 9-40 of the OPRA uses component 17 and the incorrect value of 0.70 for $B_{\rm T}$. However, it is unlikely that the higher value has a significant influence on the mean frequency of core melt.

Letdown Coolers(18): The letdown coolers are located in the containment at elevation 771 feet. The capacity factor for component 18 is based on generic values which are conservative as discussed above. The equipment and structure response factors in Table 5.5 of Annex J2 seem consistent. Compared with capacities from past seismic PRAs, the median capacity of 0.65g is low. A component-specific analysis likely would increase the capacity of this component significantly.

<u>Reactor Coolant Pump Supports(19)</u>: It was assumed in the development of the capacity factor that the reactor coolant pump supports are stressed to 80% of yield for the design earthquake. This single conservative assumption controls the analysis leading to the low median capacity of 0.72g. The equipment response factor calculations do not agree with the resulting median factor of 2.62 and appear to be on the high side relative to other Oconee component factors. If the following fragility report and calculation file factors are used, the resulting equipment median equipment response factor is only 2.14.

Spectral Shape	1.44
Qualification Method	1.0
Damping	1.6
Frequency	1.0

Mode	Shape		1.0
Mode	Combinations		1.0
Earth	nguake Component	Combination	0.93

Compared to past seismic PRAs, the median capacity of the reactor coolant pump is at least 1.5g and generally much larger. On this basis it is likely that the median capacity for the Oconee reactor coolant pump is much larger than 0.72g.

Main Feeder Buses(20) and Standby Buses(21): Both these buses are located at the ground level and, again, the generic support capacity values for heavy flexible equipment were used in the development of the capacity factor. The resulting median capacity of 0.72g is on the low side. Capacities from past seismic PRAs for structural failure of switchgear and motor control center cabinets give median values generally greater than 1.5g.

SSF 600/120-V Transformer(22), SSF 600/208/120-V Transformer(23), and SSF 600/208-V Transformer(24): All three of these components are located at elevation 817 feet in the SSF. Inspection of these components during the plant tour verified that they are securely anchored to the floor slabs. Again the generic support capacity value for heavy flexible equipment was used, although this equipment is in reality relatively light. As discussed for component 4, the median capacities of these smaller transformers are, at the least, greater than 1g.

Auxiliary Building Shear Walls(25) and Auxiliary Building Moment Frames(28): The capacities for these components were not reviewed in detail. Past PRA studies suggest that it is likely that the median capacities are larger than the reported values of 0.74g and 0.94g, respectively.

However, without further review and analysis, these values are accepted as calculated.

Letdown Line Piping(26): As in other seismic PRAs, a generic fragility analysis was conducted for piping and supports. Because the MHE is only 0.10g, the resulting median capacity is only 0.85g, which is considerably below the corresponding median capacities from past PRAs with values generally above 2.0g. The HCLPF capacity for this piping is only 0.28g. Experience in industrial facilities for earthquakes up to about 0.5g indicates that, if the equipment to which the pipe is attached is properly anchored or if other special problems do not exist, piping does not fail. It is believed that the median capacity of the piping at Oconee is much larger than 0.85g.

Reactor Vessel Internals(27): A specific analysis was conducted for the reactor vessel internals. Information concerning the capacity of the internals is difficult to obtain and generally considered to be proprietary by the NSSS suppliers. At face value the analysis seems reasonable except in the equipment response factor calculation where the ground response spectra shape factor (i.e., due to the differences between the median and design spectra) appears to have been included twice--once in the equipment response factor and then again in the structure response factor. The spectral shape factor of 0.87 in one of the factors should be removed, which will increase the median capacity to 1g.

<u>SSF Diesel Oil Storage Tank(29)</u>: The median capacity of the diesel oil storage tank for the SSF is listed in Table 5.5 of Annex J2 as 0.55g, while a value of 1.00g is used elsewhere in the OPRA. In the calculation file, a onepage summary of diesel oil storage tank capacities from past PRAs (one published and three unpublished) is given. From the data, a 1.0g median capacity was selected as indicated on the calculation page.

Since the tank is buried, it is unlikely that any realistic earthquakeinduced soil pressures could cause seismic failure. Failure would occur only if relative movement severed the attached piping. For Oconee, the 1.0g median capacity is probably on the low side.

Reactor-Vessel Skirt(30): The capacity of the reactor pressure vessel is based on the capacity of the skirt anchor bolts; however, the strength factor given in Table 5.5 of Annex J2 is 6.73, whereas the calculated value in the computation file is only 2.36. The latter value was developed on the assumption that the MHE loading equals 80% of the allowable value, which probably is conservative.

In comparison with results from past seismic PRAs, the median capacity of 1.18g is low. For example, Zion and Indian Point had corresponding values of 4.6g and 3.8g, respectively. Midland, which also is a B&W plant, had a value of 3.3g based on studies in the LLNL load combination program. It is likely that the actual median capacity is higher then the 1.18g value used in the OPRA.

Large Reactor Coolant Pipe(31): The reported median capacity of the large reactor coolant pipe in the OPRA is 1.23g and is based on a specific analysis of one of the reactor coolant pipe lines. The analysis for the capacity factor is straightforward, except for one discrepancy where the normal operating pressure appears to have been included twice in the calculation of the median factor. The more correct median capacity value is approximately 10% higher.

In comparison with other seismic PRAs, the reactor coolant pipe median capacity is a factor of 2 lower than typical values. It is difficult to argue with the calculations if, in fact, the calculated OBE stresses are as high as stated. However, no indication is given that the design stresses were checked to verify that they are reasonable. It seems surprising that a 0.05g earthquake can cause 8000 psi in the reactor coolant pipe. Past PRAs and the general belief among some engineers in the nuclear industry that piping capacities are much higher than calculated indicate that the reactor coolant pipe capacity is low.

SSF Diesel Generator(32), SSF DC Batteries and Racks(33), and SSF Diesel Day Tank(34): The capacities of all three components were developed using the generic capacity factors given on page J-5-118 of Annex J2 for flexible equipment (i.e., $F_c = 11.9$); however, the diesel generator and the day tank are listed as rigid components in Table 5.5 of Annex J2. The capacity factors for rigid equipment are lower than for flexible equipment (i.e., F_c between 7.2 and 7.5); although, as discussed above, the generic factors are conservative. A visual inspection revealed that the batteries and racks are flexible and the diesel generator and day tank are rigid. The median capacity of the diesel generator is reported to be 1.42g which is reasonably consistent with past PRAs. In past experience with earthquakes, diesel generators have not been a problem. This is intuitively reasonable, since start-up and operating forces on the anchor bolts is a severe loading which diesel generators and motors frequently see. However, in the Millstone PRA,¹¹ which was performed more recently than the OPRA, the median capacity of the lube oil cooler anchor bolts was calculated to be 0.91g. It is generally believed that the diesel generator peripherals are more vulnerable to seismic motions than the diesel generator itself. There is no indication in the PRA calculation that any attempt was made to investigate the capacities of other components associated with the diesel generator.

The median capacity of the SSF batteries and racks is 1.59g, which is consistent with past PRAs. A visual inspection suggested that the racks have above average capacity, i.e., not low as found in some older plants, which had racks with wooden battens, but not as strong as some of the newer plants such as Limerick. The racks are welded to floor plates, but the lateral bracing consists of thin bars. Styrofoam-like material is used to separate the batteries which, over the length of a row of batteries, may crush and deform at high seismic motions; however, it is doubtful that this would cause a failure. In conclusion, the capacity estimate appears to be reasonable.

The diesel day tank looked extremely rugged and is anchored to the floor with four 3/4-inch or larger anchor bolts. The capacity given in the OPRA appears to be reasonable.

600-V Distribution Center(35): This component is located in the auxiliary building at elevation 796 feet. The relay chatter median capacity is reported in the OPRA to be 1.20g on the basis of generic tests. The calculations are confusing, and it is difficult to determine how the capacity was obtained. It appears that the Corps of Engineers' shock test data were used which gives a median spectral acceleration capacity of 2.6g and a logarithmic standard deviation of 1.59. As discussed in Section 3.3.3, there is considerable uncertainty concerning the appropriate relay chatter capacity to be used.

SSF Electrical(37): A generic relay chatter capacity for the SSF is represented by component 37. In Table 5.5, all relay chatter median capacities for the SSF electrical components are indicated as being greater than 2.0g. No basis for this conclusion could be found in the report or in the calculations. The same comments given above also apply to these relays.

Letdown Storage Tank(38): The letdown storage tank is located in the auxiliary building at elevation 771. In Table 5.5 of Annex J2, the median capacity is 0.47g which is based on a generic heavy flexible equiment capacity factor. The tank was not inspected during the plant tour; hence, no specific comment can be made. This component was included because a low capacity was given and its influence on the final results was investigated.

Borated Water Storage Tank(39): This component was included to investigate its impact on the final results. The median capacity is 0.83g. The calculations were not reviewed, but the analysis approach is reasonable and consistent with similar analyses for past seismic PRAs. Although flat bottom tanks have had low capacities in past PRAs (e.g., the condensate storage tank in the Limerick PRA¹⁰ had a median capacity of 0.24g), the capacity for the Oconee BWST is reasonable and consistent with the large number of anchor bolts used in the tank anchorage system.

Core Flooding Tank(40) and LPI Cooler(41): These two components were added to the list to investigate the assumption that the capacities of these components are sufficiently high to justify eliminating them from the analysis (see Table 9.5 of OPRA). The core flooding tank is located in the containment at elevation 787 feet and the LPI cooler is located in the auxiliary building at the ground level. The capacity factors for both these components was based on the generic capacity factor for heavy flexible equipment. The median capacities are 1.33g and 0.71g for the core flooding tank and LPI coolers, respectively. One LPI cooler in the decay heat removal room was inspected during the plant visit. It is a rugged horizontal tank supported by saddle supports anchored to the floor by eight 1-in. or larger diameter bolts. Its capacity is larger than 0.71g. It is likely that the capacity of the core flooding tank is also larger than used in the OPRA.

3.3.5 Conclusions

From the review of the fragility analysis and the resulting parameter values it is believed that the capacities used in the OPRA are conservatively low. Note that in a PRA the objective of the analysis should be to be unbiased. Thus, neither conservatism nor optimism is desi ble. Generic median values were used for many of the components which resu d in conservative capacities. It was difficult to follow the capacity callations which contain many inconsistencies and, in general, were not well organized. In addition, the high-confidence low-probability values which correspond to the fragility parameter values are in many cases inconsistent with earthquake experience.

The median capacity for the block walls was particularly low. The analysis did not take advantage of the additional capacity provided by arching action, which currently is being verified by testing. This is an important consideration since the block walls are important contributors to the mean frequency of core melt.

3.4 Systems Analysis

3.4.1 Introduction

The objective of this section is to provide a discussion of the seismic accident sequence definitions and quantification. Section 3.4.2 presents a qualitative review of the accident sequence delineation, i.e., a review of the seismic event tree with its supporting logic and of the systems fault trees.

Section 3.4.3 discusses the quantification of the seismic accident sequences and presents the review results, including sensitivity analysis.

3.4.2 Seismic Sequences Definition -- A Qualitative Review

In the OPRA, the seismic event sequences were delineated by constructing one event tree that combined aspects of the event trees that were developed for the internal events analysis, i.e., for transient initiating events and for small- and large-break loss-of-coolant accidents; this seismic event tree developed in the OPRA is reproduced here as Figure 3.17 and its top events are given in Table 3.6. Again, as with the event trees for internal events, supporting logic for each of the top events in the seismic event tree was developed in the form of a fault tree combining system-level fault tree top events to define the functional failures of interest.

Before the construction of these logic models (supporting logic and system fault trees), the list of components provided by Structural Mechanics Analysis (Appendix J--Annex J2 of OPRA) was substantially reduced by eliminating irrelevant failures and by discarding those events whose median capacities were sufficiently high for their occurrence to be probabilistically unimportant. Note that this step has also been performed in all seismic PRAs. Following this step, i.e., after constructing the logic models, the OPRA made a reduction in these logic models in two steps:

- By eliminating seismic failure events with much higher median capacities than other events with the same effects.
- By combining groups of failures into modules in the same way as the modularization done in the systems fault trees for the internal events.

Having the seismic event tree, the supporting logic, and the systems fault trees, the seismic sequences were obtained by linking the fault trees and obtaining the minimal cut sets using the SETS ³⁶ computer code.

This review examined all the steps used in the OPRA for the delineation of the seismic event sequences. Comments and modifications to each of the steps follows:

- Seismic Event Tree: This review is in agreement with the delineation of the seismic event tree presented in the OPRA (Figure 3.17).
- o Supporting Logic and System Fault Trees: As explained in step 1 above, the OPRA eliminated several seismic failure events (see Table 9.5 of the OPRA) for the construction of the supporting logic and systems fault trees. In this review, several of these events were kept in the fault trees (components 38 through 41 in Table 3.5).

In addition, the following events were also added to the supporting logic and systems fault trees used in this review:

 System random failures - included as a single developed event for all systems.

- b. Operator errors several operator errors were added to the system fault trees to obtain a more realistic modeling of the systems given specific failures.
- c. Relay chatter ~ added to the fault trees, where applicable. The failure of systems due to relay chatter was used in this review only in the sensitivity analysis.

The modified fault trees used in this review are presented in Appendix F.

The seismic sequences, i.e., minimal cut sets for every sequence in the event tree in Figure 3.17, were obtained by fault tree linking with the SETS³⁶ code. Note that because the failure probabilities of many components conditional on the occurrence of an earthquake are relatively high, the success states as well as the failures were accounted for in the seismic sequences.

The next section presents the results of the seismic sequences quantification. Note that the inclusion of all the modifications described above had a very small effect on the total core damage due to seismic events.

3.4.3 Review Results and Sensitivity Analysis

The quantification of the seismic accident sequences was performed by using the logic models described in Section 3.4.2. Several analyses were conducted to verify the results given in the OPRA and to investigate the effects of altering the component capacities, operator error, and random failure rates. In each sensitivity analysis a mean frequency of occurrence was calculated both for the individual eight sequences and for total core melt (i.e., the sum of the eight sequences). An independent check was performed using the systems equation for core melt which was derived directly. The results using this equation were compared with the case where the results of the eight sequences were added. Note that the mean frequency is more easily calculated than the entire probability of frequency distribution and provides a reasonable measure to investigate the effects of different assumptions.

These analyses were performed with a computer program developed by JBA which calculates system fragility curves using the system's equations and the lognormal distributions for the components, random failures, and operator errors. The program then integrates the system fragility curves with the site hazard curves. In the sensitivity analyses for Oconee, the logarithmic standard deviations for randomness and uncertainity were combined for each component into a common value. It can be shown theoretically that using a combined logarithmic standard deviation produces the mean fragility curve for a component and subsequently the mean frequency of failure when the mean system's fragility curve is integrated with the mean site hazard curve. Because of the complexity of the system's Boolean equations for the eight sequences, it was difficult to convert them to probability equations which could be used directly. Hence, a strategy was adopted whereby the system fragility curves were calculated using Monte Carlo simulation. One thousand simulations were made for each acceleration level to obtain the mean frequency of failure as a function of the acceleration level (i.e., fragility curve). For low frequencies of failure (i.e., 0.01 and below) where the Monte Carlo procedure is inaccurate, the Boolean equations were used directly as probability equations. This produces slightly conservative results since the redundant intersections

are not subtracted out. For cases which are significant, the low conditional frequencies of failure do not contribute to the calculated mean frequency of failure. Thus, the adjustment made in the tails of the fragility curves is not a major problem in calculating total core melt, but may affect the results for sequences which have very low frequencies of failure.

In the first analysis, an attempt was made to replicate the example calculation given on page 9-40 of the OPRA. The purpose of the calculation was to demonstrate the procedure used in the OPRA to discretize the hazard and fragility curves and to perform the integration leading to the probability distribution of frequency of core melt. The example considered a simple case of the failure of the masonry walls and the pressurizer supports. All of the 18 hazard curves were used, but only three fragility curves were used for each of the two components (i.e., a median curve and the curves one standard deviation higher and lower, which were weighted 0.6, 0.2, 0.2, respectively). The mean frequency of core melt reported in the OPRA on page 9-42 is 4.4E-6/yr.

The mean frequency was recalculated to be 5.1E-6/yr which is only 16% higher. Two additional calculations were made which used the three fragility curves for each of the two components, weighted in the same manner as in the OPRA. The first additional calculation used the same discretization for the hazard and fragility curves adopted in the OPRA, while the second used an acceleration spacing of 0.02g. The mean frequency of core melt values per year were calculated to be 5.4E-6 and 5.1E-6, respectively. On the basis of the close comparison, the discretization and analysis procedure appears to produce reasonable results.

Also, an unsuccessful attempt was made to verify the individual frequency values given in Table 9.12 of OPRA. It appears that the values used for the hazard curves were different from those in Table 9.9 of OPRA. The analysis procedure in the OPRA used the hazard curves directly rather than the derivative of the hazard curves, which is more correct. As long as a hazard curve is relatively steep (i.e., by comparison with the change in the fragility curve) and the spacing between integration points (i.e., acceleration values) is not too large, then use of the hazard curve directly will produce essentially the same results. However, it was not confirmed which hazard values were actually used in the example analysis, although, as stated above, the final results (at least the mean value) seem reasonable.

In the next analysis an attempt was made to replicate the mean frequency of failure for the six core melt bins defined in the OPRA (see Figure 3.17). Table 3.7 gives the results of the analysis and compares them with the results reported in the OPRA. In this analysis the fragility parameter values were assumed to be the same as those used in the OPRA. For structural components which were eliminated in the OPRA (see components 35 through 41 in Table 3.5) and the random failure and operator action which were not included in the analysis (only RCSRVLC and OP3 were included in the OPRA), appropriate extreme values were assumed to suppress their effect.

As seen in Table 3.7, the calculated total core melt is only 19% higher than given in the OPRA (i.e., compare 0.75E-4 to 0.63E-4 per year). All sequence bins except bins V and VI compare closely to the OPRA results. For bin V, the calculated value of 0.13E-4/yr is higher by a factor of 4 than the value of 0.32E-5 reported in the OPRA. Finally, for bin VI, the calculated

value of 0.58E-6/yr is a factor of 36 higher than the reported value of 0.16E-7. Since the frequencies of failure for the entire sequence 8 fragility curve are less than 0.01, the factor of 36 is partially due to the conservative approximation used in this review analysis as discussed above. Subsequently, an exact probability equation was developed for sequence 8 (i.e., bin VI) and an "exact" value of 0.15E-6 was calculated, which is a factor of 10 larger than the value of 0.16E-7 reported in the OPRA.

Note that the calculated values in Table 3.7 are referred to as the Base Case in subsequent comparisons. Figures 3.18a and 3.18b show the fragility curves for the Base Case 8 sequences and total core melt. It is also interesting to note that in the process of integrating the hazard and core melt fragility rurve 25%, 50%, and 75% of the mean frequency of core melt (i.e., 0.75E-4/yr) is accumulated at 0.17g, 0.21g, and 0.28g, respectively. This verifies the importance of the hazard curve accuracy in the acceleration region less than 0.4g.

In the next sensitivity analysis, the effect of including components 38 through 41 listed in Table 3.5 (i.e., letdown storage tank, BWST, core flooding tank, and LPI cooler) was investigated. These components were eliminated from the OPRA analysis as discussed in Section 3.4.2. Table 3.8 gives the comparison of the Base Case with the case where components 38 through 41 were included. As can be seen, the inclusion of components 38 through 41 has no significant effect on the results.

Next, the effect of alternative values for the random failures and operator errors was investigated, i.e., the effect of using the logic models developed in this review. Table 3.9 shows the values that were assumed in the OPRA (i.e., Base Case) and the alternative values that are proposed. Table 3.10 gives the comparison of the results of the Base Case with the alternative case. As seen in Table 3.10, the total core melt frequency is only slightly higher than the base case, i.e., 0.82E-4 compared with 0.75E-4. In addition, a bounding case analysis in which all operator errors were considered to occur with certainty was performed. Table 3.11 shows the results for the bounding case as compared to the Base Case.

The effect of considering relay chatter capacity values typical of past PRAs, i.e., median capacity equal to 0.6g, was investigated next. In addition, each relay chatter capacity was paired in parallel with a manual reset event (MANRESET) which was assigned a frequency of failure on demand value of 0.5. Table 3.12 compares the effects of using typical relay chatter capacities from past PRA analyses with the Base Case. As seen the effect is small. Relay chatter has produced significant effects in other PRA studies¹⁸; however, because of the low capacities of other components the effects of including relay chatter are masked. If more realistic capacities were used for the structural components, it is likely that relay chatter would be a more prominent contributor.

The effect of increasing the block wall and condenser capacities was investigated next. Table 3.13 presents the results of three cases. The first case considers more realistic capacities for the block walls in the auxiliary building and in the CT4 blockhouse (i.e., components 5 and 9, respectively) as discussed in Section 3.3.4. In the first column the block wall capacities are increased to realistic values, causing a decrease in core melt by a factor of 0.7. Even if the block wall components capacities are significantly increased (e.g., median capacities equal to 10.0g), the decrease in core melt (see second column) is essentially the same as shown in the first column. Finally, the last column in Table 3.13 shows the effect of increasing the condenser capacity to a more realistic value as discussed in Section 3.3.4. This change is added to the block wall capacity change as indicated in column 1. The combined change decreases the core melt capacity by a factor of 0.6 times the Base Case, as shown in the third column of Table 3.13.

Next, the effect of eliminating Jocassee Dam, the pressurizer and reactor coolant pump components was investigated. As given in the first column of Table 3.14, the mean frequencies of failure for sequences 2, 3, 6, and 8 become essentially zero when Jocassee Dam is eliminated. This result was expected since failure of long-term cooling is prevented if Jocassee Dam does not fail. As shown in the last two columns, elimination of the pressurizer and condenser components causes major decreases to sequences 7 and 8; however, since these sequences are not large contributions to core melt, the total mean frequency of core melt is essentially unchanged.

In the next series of analyses, components were systematically removed (i.e., median capacities increased to 10g) to investigate the robustness of the results. Table 3.15 gives the components which were removed for each of the eight cases considered. Cases 1 and 2 represent primarily the removal of singles and some doubles. Cases 3 through 8 remove additional doubles from the model. Table 3.16 gives the mean frequencies of failure to: the Base Case and each of the analysis cases listed in Table 3.15 for the 8 sequences and for total core melt.

As can be seen in Table 3.16, cases 1 and 2 cut the core melt mean frequency approximately in half compared to the Base Case. Also sequences 3, 4, and 6 go to zero and sequences 2 and 8 are also essentially zero, leaving sequences 1, 5, and 7 which contribute significantly to core melt. As components 4, 6, 11, and 17 (i.e., 600/208-V transformer, transformer CT3, 4160/600-V transformers and pressurizer supports) are sequentially included in the list of eliminated components, the mean frequencies of failure stay essentially the same (see analysis cases 3 through 6). In case 7 when component 1 (offsite power) is added, sequences 1, 5, and 7 and total core melt drop to essentially zero. Case 8 shows the results if component 1 is eliminated, but components 4, 6, 11, and 17 are added back in. When this is done, the mean frequencies of failure increase and are similar to the results obtained for cases 3 through 6.

3.5 References

- Commonwealth Edison Company, Zion Nuclear Plant Units 1 and 2 Probabilistic Safety Study, Dockets 50-295 (Unit 1) and 50-304 (Unit 2), September 8, 1981, available at NRC Public Document Room.
- Power Authority of the State of New York and Consolidated Edison Company of New York, Indian Point Probabilistic Safety Study, Units 2 and 3, Dockets 50-247 (Unit 2) and 50-286 (Unit 3), March 5, 1982, available at NRC Public Document Room.

- Berry, D. L. et al., "Review and Evaluation of the Zion Probabilistic Safety Study," Prepared for U. S. Nuclear Regulatory Commission, NUREG/CR-3300, May 1984.
- Kolb, G. J. et al., "Review and Evaluation of the Indian Point Probabilistic Safety Study," Prepared for U. S. Nuclear Regulatory Commission, NUREG/CR-2934, December, 1982.
- Bernreuter, D. L., J. B. Savy, R. W. Mensing, and D. H. Chung, "Seismic Hazard Characterization of the Eastern United States: Methodology and Interim Results for Ten Sites," Lawrence Livermore National Laboratory, Prepared for U.S. Nuclear Regulatory Commission, NUREG/CR-3756, 1984.
- Algermissen, S. T., D. M. Perkins, P. C. Thenhaus, S. L. Hanson, and B. L. Bender, "Probabilistic Estimates of Maximum Accelertion and Velocity in Rock in the Contiguous United States," U.S. Geological Survey, Open-File Report 82-1033, 1982.
- Electric Power Research Institute, "Seismic Hazard Methodology for Nuclear Facilities in the Eastern United States," Prepared by Dames and Moore, EPRI Project No. P101-29, April 1985 (Draft, in Peer Review).
- U.S. Geological Survey, "Studies Related to the Charleston, South Carolina, Earthquake of 1886 - Tectonics and Seismicity," Geological Survey Professional Paper 1303, 1983.
- 9. Talwani, P. and J. Cox, "Paleoseismic Evidence for Recurrence of Earthquakes Near Charleston, South Carolina," Science, Vol. 229, July 26, 1985.
- NUS Corporation, "Limerick Generating Station Severe Accident Risk Assessment," Prepared for Philadelphia Electric Company, 1983.
- Northeast Utilities Service Company, "Millstone Unit 3 Probabilistic Safety Study," August, 1983.
- Pickard, Lowe, and Garrick, Inc., "Seabrook Station Probabilistic Safety Assessment," Prepared for Public Service of New Hampshire and Yankee Atomic Electric Company, 1983.
- American Nuclear Society and Institute of Electrical and Electronic Engineers, "PRA Procedures Guide," Prepared for U.S. Nuclear Regulatory Commission, NUREG/CR-2300, January 1983.
- Bari, R. A. et al., "Probabilistic Safety Analysis Procedures Guide," Prepared for U. S. Nuclear Regulatory Commission, NUREG/CR-2815, Vol. 1, August, 1985.
- Cornell, Ca. A., "Probabilistic Seismic Hazard Analysis: A 1980 Assessment, "Proceedings of the Joint U.S.-Yokoslavia Conference on Earthruake Engineering, Skopje, Yugoslavia, 1980.

- Azarm, M. A. et al., "A Review of the Limerick Generating Station Severe Accident Risk Assessment," Prepared for U.S. Nuclear Regulatory Commission, NUREG/CR-3493, July, 1984.
- Garcia, A. A. et al., "A Review of the Millstone 3 Probabilistic Safety Study," Prepared for U.S. Nuclear Regulatory Commission, NUREG/CR-4142, May, 1984.
- Shiu, K. K. et al., "A Review of the GESSAR-II BWR/6 Standard Plant Seismic Probabilistic Risk Assessment," Prepared for U.S. Nuclear Regulatory Commission, NUREG/CR-4135P, Vol. 2, August, 1985.
- Garcia, A. A. et al., "A Review of the Seabrook Station Probabilistic Safety Assessment," Prepared for U.S. Nuclear Regulatory Commission, December 12, 1984, Draft.
- Bernreuter, D. L., "Seismic Hazard Analysis: Application of Methodology, Results and Sensitivity Studies," Lawrence Livermore National Laboratory, Prepared for U.S. Nuclear Regulatory Commission, Vol. 4, NUREG/CR-1582, 1981.
- Algermissen, S. T. and D. M. Perkins, "A Probabilistic Estimate of Maximum Acceleration in Rock in the Contiguous United Stats," U.S. Geological Survey, Open File Report 76-416, 1976.
- 22. Bollinger, G. A., "Seismic Activity in the Southeastern United States," Earthquake Engineering Conference, University of South Carolina, 1975.
- Taber, S., "The South Carolina Earthquake of January 1, 1913, "Bulletin of the Seismological Society of America, Vol. 3, 1913.
- Bollinger, G. A., <u>A Catalog of Southeastern United States Earthquakes</u>, <u>1754 through 1974</u>, in Research Div. Bulletin 101, Dept. of Geological Sciences, Virginia Polytechnic Institute and State University, Blacksburg, VA, 1975.
- McGuire, R. K., "Effects of Uncertainty in Seismicity on Estimates of Seismic Hazard for the East Coast of the United States," Bulletin of the Seismological Society of America, Vol. 67, No. 3, 1977.
- 26. Computer Science Corporation, "The Correlation of Peak Ground Acceleration Amplitude with Seismic Intensity and Other Physical Parameters," National Technical Information Service, PB-263972, 1977.
- TERA Corporation, "Seismic Hazard Analysis, A Methodology for the Eastern United States," Prepared for Nuclear Regulatory Commission, NUREG/CR-1582, Vol. 2, 1980.
- Bollinger, G., "Reinterpretation of the Intensity Data for the 1886 Charleston, South Carolina Earthquake," USGS Professional Paper 1028, Printing Office, Washington, 1977.

- Trifunac, M. D. and A. G. Brady, "On the Correlation of Seismic Intensity Scales with Peaks of Recorded Strong Ground Motion," Bulletin of the Seismological Society of America, Vol. 65, No. 139, 1975.
- Newmark, N. M., "A Study of Vertical and Horizontal Earthquake Spectra," WASH 1255, Nathan M. Newmark Consulting Engineering Services, Prepared for U.S. Atomic Energy Commission, April 1973.
- Kennedy, R. P., W. P. Tong, and S. A. Snort, "Earthquake Design Ground Acceleration Versus Instrumental Peak Ground Acceleration," Prepared for Nathan M. Newmark Consulting Engineering Services, Structural Mechanics Associates Report No. SMA 12501.01R, December 1980.
- McGuire, R. K., "EQRISK, Evaluation of Earthquake Risk to a Site," Open-File Report 76-67, U.S. Geological Survey, 1976.
- L. E. Cover et al., "Handbook of Nuclear Power Plant Seismic Fragilities," Prepared for U.S. Nuclear Regulatory Commission, NUREG/CR-3558, June, 1985.
- 34. Duke Power Specification 0S0278.0000002, Specification for the Seismic Displacement Response Spectra for the Turbine, Auxiliary, Reactor and Standby Shutdown Facility Buildings, Rev. 3, February 20, 1980.
- Budnitz, R. J. et al., "An Approach to the Quantification of Seismic Margins in Nuclear Power Plants," Prepared for U.S. Nuclear Regulatory Commission, NUREG/CR-4334, July, 1985.
- Worrell, R. B. and D. N. Stack, "A SETS User's Manual for the Fault Tree Analyst," NUREG/CR-0465, April 1980.



Figure 3.1 An example of the uncertainty in seismic hazard estimates.







Figure 3.3 U.S. Geological Survey seismic sources near the Oconee site (Reproduced from Reference 21).



Figure 3.4 Seismic source areas for Expert 3 in LLNL seismic hazard study (Reproduced from Reference 20).



Figure 3.5 Seismic source areas for Expert 11 in LLNL seismic hazard study (Reproduced from Reference 20).



Figure 3.6 Seismic source areas suggested by Bollinger (Reproduced from Reference 22).



Figure 3.7 Comparison of the best estimate ground motion models from the LLNL ground motion panel for magnitudes 5 and 7 and the Oconee attenuation relationships.



Figure 3.8 Comparison of the Oconee attenuation models and other models provided by the SHCP ground motion experts.



Figure 3.9 USGS seismic source areas defined in their 1982 study (Reproduced from Reference 6).



Figure 3.10 Comparison of the 1976 and 1982 USGS studies and the mean Oconee seismic hazard curve.



Figure 3.11 Comparison of cases 1 and 2.


Figure 3.12 Comparison of cases 1 and 2 results using a three standard deviation truncation.



Figure 3.13 Comparison of cases 1 and 2 using a five standard deviation truncation.









Figure 3.17 OPRA seismic event tree.



Figure 3.18a Sequence and core melt fragility curves.



Figure 3.18b Sequence fragility curves.

Seismotectonic	Area	Activity Rate	Activity Rate per sq. mile		Maxin	num Earthq lypothesis MMI	uake
Region	(Square Miles)	$(I_0 \ge V)$	(x 10-6 yr)	b _I -value	1	2	3
Piedmont and Upper Coastal Plain	127,400	0.609	4.78	0.56	VII	VIII	IX
Blue Ridge	28,850	0.282	10.91	0.59	VII	VIII	IX
Charleston Zone	8,800	0.173	19.66	0.48	X	X	Х
Deformed Appalachian Highlands	63,160	0.473	7.49	0,59	VIII	VIII	IX
Central Stable Region	309,400	0.882	2.85		VIII	VIII	IX
New Madrid Zone	12,140	0.764	62.9		XII	IIX	XII
Florida Platform	210,000	0.054	0.26	- il	VI	VII	AII
Subjective Probabili	ty (Maximum Earth	quake)			0.5	0.3	0.2

Table 3.1 Oconee Seismic Hazard Analysis Seismicity Parameters

Seismic Expert	Höst* Zone	Self Wt.	Activity Rate Events/sq mile $(I_0 \ge V) \ge 10^{-6}/yr$	b _I -value	Low	I max Best	Upper
1	3	9	9.80	0.63	8.5	8.8	9.0
2	29	6	5.34	0.448	8.5	9.5	10.1
3	11	5	2.13	0.47	8.5	10.1	11.5
4	9	7	2.85	0.45	8.5	8.9	9.3
5	10	8	1.17	0.64	7.0	8.0	9.0
6	10	5	14.52	0.545	11.5	12.5	13.5
7	8	7	2.25	0.45	7.5	8.5	9.5
8			1	-			-
9		-			-		-
10	28	6.3	0.16	0.50	7.1	7.5	8.5
11	7	6.3	1.57	0.50	8.9	9.5	10.5
12	3	8.5	13.05	0.48	7.7	7.9	8.1
13	8	6.5	4.93	0.51	8.9	9.5	9.9

Table	3.2	LLNL	Seismi	C H	lazard	Characteri	zation	Project
		Seism	icity	Par	ameter	Estimates	(Refer	ence 5)

*From the experts' base map.

Attonuation	Standard .				MM Inter	nsity			
Model	Deviation		٧I	VII	VIII	1 X	X	ΧI	XII
		Median (g's)	0.0065	0.116	0,206	0.365	0.650	1.156	2.057
	0.26		0,216	0.382	0.681	1.210	2.153	3.829	6.812
Eq. (7)	0.31		0.271	0.482	0.858	1.524	2.712	4.822	8.579
	0.39		0.392	0.696	1.241	2.203	3.921	6.973	12.406
		Median (g's)	0.059	0.102	0.177	0.307	0.533	0.923	1.600
	0.26		0.195	0.338	0.586	1.017	1.764	3.057	5.298
Eq. (9)	0.31		0,245	0,426	0.739	1.281	2.221	3.850	6.673
	0.39		0.355	0.615	1.068	1.853	3.212	5.567	9.649

Table 3.3	Maximum	Sustained	Accel	lerati	ons
-----------	---------	-----------	-------	--------	-----

Source	Activity* Rate	h svalue	м	ı
Area	(*0 - •)	o I -value	'ma x	*ma x
		1976 Study, Reference 21 (See Figure 3.2-3)		
64 65	0.544 0.199	0.59 0.33	6.1 7.3	X IIIV
		1982 Study, Reference 6 (See Figure 3.2-9)		
99 100 101	0.248 0.423 0.187	0.50 0.50 0.50	7.3 7.3 7.3	X X X

Table 3.4 USGS Seismic Source Parameters for Sources Near the Oconee Site

*Number of Modified Mercalli Intensity events/year.

	Component	Variability Median (g)	β _r	ßu	HCLPF(g)*
1.	Offsite-power insulators	0.20	0.20	0.25	0.12
2.	100-kV Lee Feeder	0.20	0.20	0.25	0.12
3.	Condenser	0.21	0.14	0.14	0.16
4.	600/208-V Transformer	0.27	0.20	0.41	0.12
5.	Auxiliary Building Masonry Walls	0.28	0.23	0.28	0.15
6.	Transformer CT3	0.30	0.21	0.43	0.13
7.	Transformer CT5	0.30	0.21	0.43	0.13
8.	Transformer CT4	0.31	0.17	0.26	0.19
9.	Transformer CT4 Blockhouse	0.31	0.27	0.31	0.15
10.	HPSW Elevated Storage Tank	0.35	0.30	0.52	0.11
11.	4160/600-V Transformers	0.40	0.20	0.41	0.18
12.	Feedwater Heaters	0.40	0.30	0.52	0.13
13.	Jocassee Dam	0.48	0.40**	0 32**	0.18
14	High Voltage Rus Ducting	0.56	0.30	0.52	0.18
15.	Keowee Dam (using east intake fragility)	0,58	0.32**	0.34**	0.24
16.	Upper Surge Tank	0.62	0.26	0.47	0.23
17.	Pressurizer Supports	0.62	0.20	0.28	0.35
18.	Letdown Coolers	0.65	0.27	0.40	0.27
19.	Reactor Coolant Pump Supports	0.72	0.30	0.44	0.26
20.	Main Feeder Ruses	0.72	0.30	0.62	0.10
21	Standby Ruses	0.72	0.30	0.62	0.19
22	SSE 600/120-V Transformer	0.72	0.24	0.44	0.19
23	SSF 600/208/120-V Transformer	0.73	0.24	0.44	0.29
24	SSF 600/208-V Transformer	0.73	0.24	0.44	0.29
25.	Auxiliary Building Shear Walls	0.74	0.21	0.25	0.43
26.	Letdown Line Piping	0.85	0.29	0.50	0.29
27.	Reactor Vessel Internals	0.86	0.29	0.37	0.36
28.	Auxiliary Building Moment Frames	0.94	0.28	0.28	0.50
29.	SSF Diesel Oil Storage Tank	1.00	0.25	0.43	0.40
30.	Reactor Vessel Skirt	1.18	0.21	0.29	0.64
31.	Large Reactor Coolant Pipe	1.23	0.32	0.43	0.44
32.	SSF Diesel Generator	1.42	0.31	0.50	0.46
33.	SSF dc Batteries and Backs	1.59	0.30	0.51	0.52
34.	SSF Diesel Day Tank	1.95	0.27	0.54	0.63
35.	600-V Distribution Center	1.20	0.56	1.55	0.05
36.	4160-V Transformer	0.46	0.26	0.45	0.18
37	SSE Electrical	2 00	N/A	N/A	N/A
38	Letdown Storage Tank	0.47	0.24	0 49	0.19
39	RWST	0.83	0.29	0.40	0.20
40	Core Flooding Tank	1 22	0.20	0.51	0.39
41.	LPI Cooler	0.71	0.27	0.48	0.25

Table 3.5 Component Fragility Farameter Values

*HCLPF = 1.23 x Median x e $-1.645(B_r + B_u)$. **Values calculated based on fragility curves in Annex J3.

Event Designation	Name					
E	Seismic initiating event					
A	Large-break LOCA					
Q	Small-break LOCA					
к	Failure of the reactor protection system					
В	Failure of RCS heat removal					
U ₁	Failure of coolant injection (large-break LOCA)					
U ₂	Failure of coolant injection (small-break LOCA)					
U ₃	Failure of coolant injection (loss of all feedwater)					
X	Failure of long-term cooling					

Table 3.6 Top Events for the Seismic Event Tree

			Mean Frequen	cy of Failure	(per year)		
0			Calcu	lated*		Ratio	
PRA Bin	n Sequence		Sequence	Bin	OPRA	OPRA	
I	1.	EÃQU2	0.99E-5	0.99E-5	0.11E-4	0.90	
II	2.	EÃQU ₂ X	0.80E-7	0.45E-5	0.26E-5	1.73	
	3.	EAQKBX	0.44E-5				
III	4.	EAOK	0.16E-5	0.50E-4	0.46E-4	1.09	
	5.	EAQKBU3	0.48E-4				
IV	6.	EAQKBU3X	0.79E-7	0.79E-7	ε		
٧	7.	EAU1	0.13E-4	0.13E-4	0.322-5	4.06	
VI	8.	EAUIX	0.58E-6	0.58E-6	0.16E-7	36.25	
	Total	Core Melt	0.75E-4	0.75E-4	0.63E-4	1.19	

Table 3.7 Comparison of Calculated Mean Frequency Values With Results in OPRA

*These values are referred to as the Base Case.

		Mean Frequency of Failu	ure (per year)
Sec	uence	Including Components	Base Case
1.	EAQU2	0.99E-5	0.99E-5
2.	EAQU2X	0.972-7	0.80E-7
3.	EAQKBX	0.43E-5	0.44E-5
4.	EAQK	0.16E-6	0.16E-5
5.	EAQKBU 3	0.49E-4	0.48E-4
6.	EAQKBU 3X	0.79E-7	0.79E-7
7.	EAU 1	0.13E-4	0,13E-4
8.	EAU1X	0.57E-6	0.58E-6
Tot	al Core Melt	0.77E-4	0.75E-4

Table 3.8 Effect of Including Components 38 Through 41

		Frequency of	of Failure on Demand
Random F	ailures	Base Case	Alternative Values
RCSRVLC R1 R2 R3 R4 R5	Closure of two SRVs (RCSRVLC) Main Feedwater (MFWRAND) Turbine-Driven EF (TDEFWRAND) Motor-Driven EF (MDEFWRAND) HPI (HPIRAND) Low Pressure Injection (LPIRAND)	0.1 N/A N/A N/A N/A	0.1 0.04 0.091 0.01 2.0E-4 5.0E-4
Operator	Errors		
CP1	Failure to transfer to EFW Suction given loss of 4160 VDC 600-V Power (OP1)	N/A	0,2
OP2	Failure to Transfer to EFW Suction give loss of power from Transformer CT3 (OP2)	n N/A	0.15
OP3	Failure to activate SSF Feedwater (includes hardware) (SSFFDWF)	0,1	0.30
OP4	Failure to supply power from Lee Station (OPLEEH)	N/A	0.10
0P5	Failure to Transfer to Recirculation given small LOCA (OPRECH2)	N/A	3.0E-4
0P6	Failure to Transfer to Recirculation given large LOCA (OPRECH3)	N/A	0.05
0P7	Failure to provide cooling through "Feed and Bleed" (VTHPIH)	N/A	0.05

Table 3.9 Random Failure and Operator Error Mean Frequencies

	Mean Frequency of F	ailure (per year)	Ratio	
Sequence	Alternative Case*	Base Case*	Base Case	
1. EĀQU ₂	0.10E-4	0.99E-5	1.01	
2. EĀQŪ ₂ X	0.89E-7	0.80E-7	1.11	
3. EAQKBX	0.39E-5	0.44E-5	0.89	
4. EAQK	0.11E-5	0.16E-5	0.69	
5. EAQKBU3	0.54E-4	0.48E-4	1.13	
6. EAQKBU ₃ X	0.222-6	0.79E-7	2.78	
7. EAU ₁	0.13E-4	0.13E-4	1.00	
8. EAU ₁ x	0.20E-5	0.58E-6	3.45	
Total	0.82E-4	0.75E-4	1.09	

Table 3.10 Effect of Alternative Random Failure and Operator Error Frequencies

*See Table 3.9 for assumed random failure and operator error frequencies.

Sequence		Base Case	Bounding Case
1.	EĀQU2	0.992-5	0.12E-4
2.	EAQU2X	0.80E-7	0.64E-5
3.	EAQKBX	0.44E-5	0.20E-5
4.	EAOK	0.16E-5	0.71E-6
5.	EAQKBU 3	0.48E-4	0.13E-3
6.	EAQKBU 3X	0.79E-7	0
7.	EAU1	0.13E-4	0.14E-4
8.	EAU ₁ X	0.58E-6	0.788-5
Total Core Melt		0.75E-4	0.17E-3

Table 3.11 Bounding Cases: No Credit for Operators Actions Mean Frequency of Failures (per year)

	Mean Frequency of Failure	e (per year)	Ratio	
Sequence	Case with Relay Chatter	Base Case*	Base Case	
1. EAQU2	0.10E-4	0.99E-5	1.01	
2. EAQU2X	0.58E-7	0.80E-7	0.73	
3. EAQKBX	0.43E-5	0.44E-5	0.98	
4. EAQK	0.13E-5	0.168-5	0.81	
5. EAQKBU3	0.50E-4	0.48E-4	1.04	
6. EAQKBU ₃ X	0.63E-7	0.79E-7	0.80	
7. EAU1	0.13E-4	0.13E-4	1.00	
8. EAU ₁ X	0.42E-6	0.58E-6	0.72	
Total Core Melt	0.78E-4	0.75E-4	1.04	

Table 3.12 Effect of Relay Chatter

	Mean Frequency of Failure (per year)/Ratio*				
Sequence	Increased Block a(5) = 2.0g a(9) = 0.9g	Wall Capacity a(5) = 10.0g a(9) = 10.0g	Increased Block Wall and Condenser Capacity a(5) = 2.0g a(9) = 0.9g a(3) = 0.5g		
1	0.80E-5/0.81	0.79E-5/0.80	0.48E-5/0.48		
2	0.95E-7/1.19	0.95E-7/1.19	0.17E-6/0.21		
3	0.57E-5/1.30	0.57E-5/1.30	0.63E-5/1.43		
4	0.12E-5/0.69	0.12E-5/0.75	0.12E-5/0.75		
5	0.28E-4/0.58	0.28E-4/0.58	0.24E-4/0.50		
6	0.96E-7/1.22	0.96E-7/1.22	0.202-6/0.25		
7	0.12E-4/0.92	0.12E-4/0.92	0.87E-5/0.67		
8	0.71E-6/1.22	0.71E-6/1.22	0.11E-5/0.19		
Total Core Melt	0.54E-4/0.7	0.54E-4/0.72	0.45E-4/0.60		

Table 3.13 Effect of Alternative Block Wall and Condenser Capacities

*Ratio of mean frequency of alternative case to Base Case. (See Table 3.7 for Base Case frequencies.)

		Mean Frequ	uency of Core Meit	(per year)/Ratio*
Sequence		Eliminating Jocassee Dam	Eliminating Pressurizer	Eliminating Pressurizer and Condenser
1.	EAQU2	0.99E-5/1.0	0.12E-4/1.21	0.13E-4/1.31
2.	EAQU2X	/ 0	0.84E-7/1.05	0.84E-7/1.05
з.	EAQKBX	/ 0	0.45E-5/1.02	0.45E-5/1.02
4.	EAQK	0.16E-5/1.0	0.17E-5/1.06	0.18E-5/1.13
5.	EAQKBU 3	0.48E-4/1.0	0.54E-4/1.13	0.57E-4/1.19
6.	EAQKBU 3X	/ 0	0.83E-7/1.05	0.83E-7/1.05
7.	EAU1	0.13E-4/1.0	0.36E-5/0.28	0.51E-6/0.04
8.	EAU ₁ X	/ 0	0.49E-7/0.08	0.19E-8/ 0
Tot	al Core Melt	0.71E-4/0.95	0.74E-4/0.99	0.74E-4/0.99

Table 3.14 Effect of Eliminating Jocassie Dam, Pressurizer, and Reactor Coolant Pump Components from Analysis

*Ratio of mean frequency of alternative case to Base Case (see Table 3.7 for Base Case frequencies).

Components Removed		Analysis Cases							
From Base Case Model*	1	2	3	4	5	6	7	8	
3		X	X	X	X	X	X	X	
5	X	X	Х	X	Х	X	X	X	
13	Х	X	X	X	X	X	X	X	
14	X	X	Х	X	X	Х	X	X	
18	X	X	Х	Х	Х	X	Х	Х	
20	Х	Х	Х	X	Х	Х	X	X	
21	X	Х	X	X	Х	X	X	Х	
25	X	Х	Х	X	Х	X	X	Х	
26	X	X	X	Х	Х	X	X	Х	
27	Х	X	X	Х	X	Х	X	X	
28	Х	X	X	X	Х	Х	X	X	
30	Х	Х	Х	Х	Х	Х	X	Х	
4			X	X	Х	X	Х		
6				Х	Х	X	X		
11					Х	Х	X		
17						X	Х		
1							X	X	

Table 3.15 Components Removed From Base Case Plant System Model

*For component names, see Table 3.5.

				A	nalysis Case*				
Sequence	Base Case	1	2	3	4	5	6	7	8
1. EAQU2	0.99E-5	0.53E-5	0.18E-5	0.14E-5	0.14E-5	0.12E-5	0.15E-5	0.3%E-13	0.138-5
2. EAQU2X	0.80E-7	0.14E-12	0.16E-12	0.19E-12	0.20E-12	0.20E-12	0.21E-12	0.29E-12	0.18E-1
3. EAQKBX	0.445-5	c	0	0	0	0	0	0	0
4. EAQK	0.16E-5	0	0	0	0	0	0	0	0
5. EAQKBU 3	0.48E-4	0.25E-4	0.212-4	0.20E-4	0.19E-4	0.19E-4	0.23E-4	0.43E-12	0.11E-4
6. EAQKBU ₃ X	0.798-7	0	0	0	0	0	0	0	0
7. EAU1	0,13E-4	0.12E-4	0.82E-5	0.82E-5	0.80E-5	0.69E-5	0.25E-5	0.39E-13	0.67E-5
8. EAUIX	0.58E-6	0.47E-12	0.58E-12	0.58E-12	0.60E-12	0.638-12	0.498-13	0.11E-12	0.68E-12
Total Core Meit	0.75-4	0,42E-4	0.31E-4	0.30E-4	0.28E-4	0.27E-4	0.27E-4	0.90E-12	0.20E-4

Table 3.16 Effect of Removing Components From Base Case System Model

*See Table 3.15 for components removed from Base Case.

4. REVIEW OF OTHER EXTERNAL EVENTS

The OPRA has provided an analysis of the effects of the following external events:

- Fires (Section 9.3 of OPRA)
- Tornado (Section 9.2 of OPRA)
- External Floods (Section 9.4 of OPRA)
- · Aircraft Impact (Section 9.6 of OPRA).

Compared to the analysis of the accident sequences due to internal floods and seismic events, the OPRA analysis for the above listed external events presents much less detail and information to be reviewed. Because of the nature of the analysis and because of the constraints of this review, only a scoping review was performed for fires, tornadoes, external floods, and aircraft impact; Sections 4.1 through 4.4 present the results of this scoping review.

4.1 Review of Fire Events

4.1.1 Summary of OPRA Analysis

The OPRA authors themselves recognize that "the fire analysis was carried out under resource limitations that limited the scope of the analysis activities," and thus, major assumptions had to be made in order to perform the analysis. The following remarks directly quoted from the OPRA, together with those presented in Appendix G, place the results of the OPRA into perspective:

- "The analysis was limited to areas where the analysts believed the most damage can be anticipated. Many more areas of the plant would have to be investigated in more detail for a complete fire risk analysis. The degree to which additional analysis is warranted must be balanced by the importance to the overall study results and an understanding of the limitations associated with the state of the art in the analysis of fire event sequences.
- 2. The frequencies of fires were derived from the experience of all U.S. nuclear power plants. The extent to which they reflect the conditions at Oconee Unit 3 is not entirely certain. For example, it is debatable whether fires like the Browns Ferry incident should be included in the data base because modifications have been implemented as a result of that fire. Nevertheless, all fires were included in the data base.
- Simple models were used to assess the propagation of fires in cable trays and the temperature rise in compartments due to the heat released by the fire.
- 4. The analysis of the fire-initiated sequences was not detailed. Such an analysis would explicitly include the timing of events, the possibility of restoring lost functions, the possibility of errors of commission, and a detailed analysis of local actions outside the control room.

5. Whenever a fire is postulated in an area where it can affect instrumentation, the question of completeness of the analysis becomes very important. It is very difficult to know what information would be presented to the operators and how they would respond. However, the impact of such events on the fire risk is judged to be included in the uncertainties assessed for the dominant sequences."

Starting with the above assumptions, the OPRA fire analysis concentrated on loss-of-coolant accidents (LOCAs) and transients that could be initiated by fires in the electrical-equipment room and in the cable-shaft area. This analysis concluded that:

- Equipment-room fire sequences would result in a core damage frequency equal to about 1.5E-8/yr, which is an insignificant contribution to the total core damage frequency due to fires.
- Cable-shaft fire sequences, which were analyzed by using the event tree reproduced here in Figure 4.1, result in a core damage frequency equal to 1.0E-5/yr (bin I CDF = 6.5E-6/yr and bin III CDF = 3.6E-6/ yr).

4.1.2 BNL Review

Battelle Columbus Laboratories (BCL) was retained by BNL to review the methods used in the fire-risk analysis performed to identify critical locations and the frequency of fire hazards, and the modeling of fire growth and suppression with its resulting effects on initiating events and equipment loss due to fires. This review, which is presented in Appendix G of this report, concluded that "the overall plant fire risk analysis performed for the Oconee 3 nuclear power plant appears to be a reasonable first approximation based on the results of the review with exception of the limitation of fire risk areas being limited to the auxiliary building, specifically the equipment room and the cable-shaft area."

For the areas reviewed, i.e., using the Oconee event tree with probabilities/unavailabilities unchanged with the exception of the frequency of cableshaft fire (first top event in the event tree presented in Figure 4.1), Table 4.1 (reproduced from Table 2 in Appendix G) summarizes the various factors that have been identified in the BCL review, together with an expected range of effects on the accident sequence requencies; these factors provide the range within which a more detailed study would be expected to yield results.

The review of the event tree and accident sequences presented in Figure 4.1 was performed by BNL and the following comments are appropriate:

- BNL is in qualitative agreement with the fire event tree (Figure 4.1).
- The quantification of the top event HPI, i.e., failure of the operators to indicate HPI, is based on two assumptions:
 - a. The actions to start the HPI function can be remotely performed from the control room.
 - b. The safety injection signal is not generated automatically.

The first assumption does not seem to be correct, because it is stated in page 9-102 of OPRA that the cables for the HPIS valves 3HP-24 and 3HP-25, which must be open to provide suction from the BWST, are assumed to be located in the cable shaft area. So, if these cables are damaged by the same fire that caused the LOCA, the operators need to open those valves locally, and this was not considered. Note that if the pumps are started without suction, the pumps will be damaged in a short period of time.

The second assumption, event though it seems conservative, is not necessarily true because of the location of the cables to valves 3HP-24 and 3HP-25; i.e., if the safety injection signal automatically starts the HPI pumps but the suction valves (3HP-24 and 3HP-25) do not open, the pumps may be damaged if not stopped in time.

For the above reasons, it is the opinion of the BNL reviewers that the probability of not having the HPIS to mitigate the LOCA (i.e., probability of failure for top event HPI in Figure 4.1) may approach 1.0 if no credit is taken for operator actions to locally open valves 3HP-24 and 3H-25. Note that the OPRA does not indicate that the operators are aware of this problem. To evaluate the failure probability for the top event HPI, a much more detailed analysis would be necessary. Therefore, this review will use the probability equal to 1.0 (i.e., certainty) for the top event HPI as an upper bound, noting that this value may not necessarily be a very conservative assumption.

A summary of the core damage frequency obtained in this review is presented in Table 4.2, where it is shown that the total core damage frequency from fire sequences may lie between 6.9E-6/yr and 2.2E-4/yr. In this table, a breakdown from the OPRA results and for the three cases obtained in this review is also presented.

Duke Power Company has stated, in a comment to this review, that a sequence added in this report is theoretically possible but it was not appropriate or within the bounds of the OPRA scoping fire analysis because of the low probability combination of burned/not burned control and power cables. However, the information provided to BNL was nr. sufficient for elimination of such sequence. If this sequence were to be eliminated, the core damage frequency from fire events would change from between 6.9E-6/yr to 2.2E-4/yr to between 2.5E-6/yr to 8.1E-5/yr.

4.2 Review of Tornado Events

4.2.1 Summary of OPRA Analysis

The analysis by OPRA to estimate the frequency of core damage due to sequences initiated by a tornado was divided into two parts, namely, the vulnerability of the plant to damage from tornado missiles and the vulnerability to damage from tornado-wind effects.

The OPRA analysis of the plant's vulnerability to damage from tornado missiles is a scoping analysis based on extrapolation of the results of two other detailed simulation studies of tornado-missile hazards, namely, the EPRI NP-768¹ and an SAI report for the Pilgrim Station Unit 2.² From those two studies, the OPRA concludes in its Appendix K that the unnual freque cy at which plant safety is compromised (core damage results) by tornado missiles is less than 1.0E-9. Given this low core damage frequency and given the potential for damage caused by tornado winds (discussed below), the sequences resulting from missiles were judged to be negligible.

The OPRA analysis of the tornado-wind effects on vulnerable structures was performed by using the event tree reproduced here as Figure 4.2. To determine the frequency of tornadoes which could damage plant structures the OPRA states that "an evaluation of the capacity of equipment and structures to withstand tornado-wind effects and an understanding of the core-cooling functions that must be defected for a tornado sequence to lead to core-melt, led to the choice of a tornado with wind speeds of 150 mph or more as the initiating event." This wind speed was judged to be a threshold above which two key features, the BWST and the west penetration room, become unavailable. Based on this, the data given in Appendix K of OPRA yield a frequency of $3.5 \times 10^{-5}/yr$ for tornadoes with wind exceeding 150 mph (see Figure 4.2).

The other top events in the event-tree are self-explanatory and a detailed description of these events with the explanation of how probabilities/unavailabilities were obtained is given in Section 9.2 of the OPRA. From the event tree given in Figure 4.2 the following annual core damage frequencies for each bin were obtained:

Bin I : 2.2E-6 Bin III: 1.1E-5 Total : 1.3E-5.

4.2.2 BNL Review

This review attempted to address the two parts of the OPRA tornado analysis described in the preceding section. However, the information provided in the analysis of tornado-missiles made a meaningful review impossible. Also, the results of previous PRAs seem to indicate that the accident sequences generated by tornado missiles will not be an important contribution to the Oconee-3 core damage frequency.

In the review of the tornado-wind effects the following steps were performed:

- The frequency of initiating events was reviewed on the basis of the table presented in Appendix K of OPRA, and assuming that the threshold wind speed (150 mph) and the structures damaged by the wind are those given in the OPRA.
- Using the assumptions above, BNL reviewed the event tree. It is the reviewers' opinion that given a loss of 4-kV power (event B in Figure 4.2), seal leakage will occur (loss of makeup and loss of component cooling). Thermal barrier cooling is lost when component cooling is lost. Additionally, seal injection fails because HPI is not available (loss of 4-kV power and loss of BWST) and because the possibility of makeup from the SSF is also lost as the result of tornado damage to the west penetration room (see OPRA page 9-60, event F). Implications of these failures have been discussed in internal events treatment, where it is shown that core damage will occur in several hours. Note that loss of the BWST means that the seal LOCA cannot be mitigated

even if 4-kV power is restored before core uncovery. Therefore, sequences 3 through 6 in Figure 4.2 are substituted by the sequence T_0AB , which is a bin I sequence with a frequency equal to 1.7 x $10^{-5}/yr$. The following revised core damage frequencies are then obtained:

Bin I : 1.8E-6/yr Bin III: 5.0E-6/yr Total : 2.3E-5/yr.

4.3 Review of External Flood

4.3.1 Summary of OPRA Analysis

The OPRA identified two potential sources of external flooding of the Oconee plant: a general flooding of the rivers and reservoirs in the area due to rainfall in excess of the probable maximum precipitation (PMP), and a random failure of the upstream Jocassee Dam.

For the analysis of the first sources, i.e., a general flooding, a mathematical model to predict the frequency of a storm given the duration of rainfall and the cummulative precipitation was developed. With this method, which is described in Section 9.4 of the OPRA, the following distribution was obtained:

Cumulative Probability	(Probable Maximum Precipitation) PMP yr ⁻¹ Frequency
0.05	4.9E-8
0.50	2.9E-7
0.95	8.9E-7

Since the calculated frequency of flooding of the Oconee plant due to the PMP storm is much lower than that due to a random failure of the Jocassee Dam, the OPRA concluded that precipitation-induced external flooding is a negligible contribution to core melt frequency.

For the analysis of the random failure of the Jocassee Dam, the OPRA states that:

"An analysis was performed to determine an annual frequency of failure for earth, earth-rockfill, and rockfill dams due to events other than overtopping and earthquake ground shaking, which were considered in separate analyses. Also, based on dam design information, structural failure of the spillway during discharge and failure associated with seepage along an outlet works have been eliminated as a possible failure mechanism. The following principal modes of failure were considered:

- 1. Piping.
- 2. Seepage.
- 3. Embankment slides.
- 4. Structural failure of the foundation or abutments."

From the above mechanisms of failures, data for dam failures were obtained and a model to estimate the frequency of failure of the Jocassee Dam was developed (see Section 9.4 of the OPRA). The dam-failure frequencies associated with the lower, median, mean, and upper bounds of the probability distribution estimated in the OPRA are:

Cumulative Probability	(Probable Maximum Precipitation) PMP yr ⁻¹ Frequency		
0.05	7.9E-8		
0.50	2.3E-7		
0.95	5.5E-7		
Mean	2.5E-5		

After obtaining this frequency of failure of the Jocassee Dam, the OPRA assumes bounding values of 1.0 for the conditional probability of flooding of the Oconee site given a catastrophic failure of the dam, and for the conditional probability of core melt given flooding at the Oconee site; that is, the OPRA estimates a mean core damage frequency equal to 2.5E-5/yr due to external floodings.

4.3.2 BNL Review

Because of the nature of the analysis performed in the OPRA, the lack of information on the Jocassee Dam, and the resources allocated to this review, the contribution of the external flooding to the core damage frequency could not be reviewed. The OPRA model extracts a failure frequency for Jocassee by compiling data for similar types of dam failures, and assuming that the failure frequency depends essentially on the period during which it was constructed, with more recently constructed dams displaying a lower failure frequency. A functional form was assumed for time-dependent dam failure rate, and a Bayesian analysis was performed to obtain the parameter values entering the assumed functional form. However, if we use only the data provided in Tables 9-20 and 9-21 of the OPRA, i.e., the estimated number of dam failures and the cumulative dam years, it is possible to state that the frequency of dam failures is given by:

Period	of Dam	Construction	Frequency (yr-1)
	1900 -	1975	4.7E-5
	1940 -	1975	2.9E-5
	1960 -	1975	2 3F-5

These results show that the model used by the OPRA is estimating a frequency of dam failures not much different from that obtained, straightforward, from previous experience.

In summary, since BNL did not review the contribution of external floodings to the core damage frequency Oconee-3, the OPRA frequency is used in this review, i.e., 2.5E-5/yr.

4.4 Review of Aircraft Impact Events

The OPRA developed a method to derive the probability distribution for aircraft-accident frequency. This method, described in Section 9.6 of the OPRA, is similar to what has been used in previous PRAs.

Using this method, the OPRA calculates the following distribution for the frequency of aircraft impact.

Confidence Level	Impac: Frequency per year
0.05	4.9E-10
0.50	2.5E-9
0.95	1.3E-8

Because the calculated impact frequencies are very low, it was used directly as the frequency of core damage due to aircraft impact.

Because of the negligible contribution of aircraft impact to core damage frequency, this review did not address this topic.







Figure 4.2 OPRA event tree for sequences initiated by a severe tornado.

Table 4.1 Summary of BCL Review

	Finding	Potential Effect on Sequence
1.	Frequency of fire in identified zones	x 0.5 to x 1.0
2.	Additional fuel sources	x 1 to x 1.25
3.	Upper-layer shielding of detectors	x 1 to x 1.6
4.	Effect of smoke opacity of propagation	x 0.5 to x 1
5.	As mptions of synergism of areas and fraction of large fires	x 1 to x 4

Table 4.2 Summary of Fire Core Damage Annual Frequency

		Bin I	Bin III	Total
Case 1:	OPRA	6.5E-6	3.6E-6	1.0E-5
Case 2: (incl with pheno	OPRA event tree uding probabilities) BCL review of fire menology			
(Fact	ors given in Table 4.1):			
	Lower Limit Upper Limit	1.6E-6 5.2E-5	9.0E-7 2.9E-5	2.5E-6 8.1E-5
Case 3:	Same as core 1, with BNL requantification of top event HPI in event tree of Figure 4.1 (may be an upper limit)	2.4E-5	3.6E-6	2.7E-5
Case 4:	Modification of cases 2 and 3 combined:			
	Lower Limit Upper Limit	6.0E-6 1.9E-4	9.0E-7 2.9E-5	6.9E-6 2.2E-4

5. SUMMARY

The OPRA has provided a detailed analysis for flooding events from sources within plant buildings and for seismic events. A less detailed analysis was presented for fires, tornadoes, external floodings, and aircraft impact events. The core damage frequencies calculated in the OPRA for each of these external events are presented in Table 5.1.

BNL allocated time to review each of the external events analyzed in the OPRA according to the level of detail in the OPRA analysis and the information available. Most of the time in this review was dedicated to internal flooding and seismic events, where a detailed reanalysis was performed. The results are also presented in Table 5.1, and the following comments are appropriate:

 Internal Flooding - In this review, detailed reanalysis was performed for flooding in the turbine building. The resulting core damage frequency calculated in this review was about the same as that in the OPRA (8.0E-5/yr vs 8.8E-5/yr in the OPRA). Differences did show up in the binning of the sequences and this is discussed in Section 2.

Note that even though the OPRA addresses in some detail the flooding from other sources in the turbine building and, in a scoping study, the floods in the auxiliary building, those two contributions to core damage frequency were not included in the OPRA results.

- 2. Seismic In this review, a reanalysis was performed for seismic events. The calculated core damage frequency was equal to 8.2E-5/yr as compared to 6.3E-5/yr in the OPRA. Note that this review tried to replicate the OPRA analysis, and a core damage frequency of 7.5E-5 was obtained; this result indicates a small difference due to the growth-fraction tools used. In Section 3 of this review, a limited sensitivity analysis is provided.
- 3. Fires In this review, a limited assessment of the modeling of fire growth and suppression and its effects on initiating events was performed. Several areas were found where a more detailed analysis than that performed in the OPRA would change the core damage frequency; for these areas, upper and lower bound factors were used in this review. Also, in the analysis of the fire event tree this review found an area where a more detailed analysis (with much more information than BNL had access to) would change the results of the fire sequences. Using these facts, this review assessed the core damage frequency from fire events to be between 6.9E-6/yr to 2.2E-4/yr, as compared with the 1.0E-5/yr calculated in the OPRA.
- 4. Tornadoes In this review, only a scoping analysis of tornado events was performed. The review of the tornado-wind accident sequences found a sequence not presented in the OPRA (see Section 4.2), and this difference is responsible for the change in core damage frequency from 1.3E-5/yr to 2.3E-5/yr.

- 5. External Floods No reanalysis was done for the external floods because of the lack of information and the restricted scope of this review. The same core damage frequency as that in the OPRA is used here, i.e., 2.5E-5/yr. This core damage frequency comes exclusively from the random failure of the Jocassee Dam and the assumption that a failure of the dam will result in core damage.
- Aircraft Impact Because of the negligible contribution to core damage, no effort was spent in the reassessment of the core damage frequency due to aircraft impact.

	OPRA	BNL Review
Turbine-Building Floods		
• CCW Floods • Other Floods	8.8E-5 _a	8.0E-5 4.8E-6
Auxiliary-Building Floods	_a	1.2E-5
Seismic	6.3E-5	8.2E-5
Fires	1.0E-5	6.9E-6b
		2.2E-4
Tornadoes	1.3E-5	2.3E-5
External Floods	2.5E-5	2.5E-5
Aircraft Impact	Negligible	Negligible

Table 5.1 Summary of Annual Core Damage Frequencies for External Events

aThe OPRA addresses these floods but does not include them

in its final results. bBNL presents upper and lower bounds based on the fire phenomenology modeling (see Appendix G).
APPENDIX A

REVIEW OF OPRA FAULT TREE APPROACH TO QUANTIFICATION OF THE FLOOD-INITIATING EVENTS

This appendix addresses the fault trees constructed by OPRA to evaluate the flood initiating events and for construction of broad groups of flood-initiator categories for the subsequent analysis of the accident sequences.

BNL found the analysis to be generally correct. Most of the comments are minor or have a small effect on the results. Table A.1 summarizes the BNL initiators, their flow rates, and their frequencies. This analysis was redone by BNL because of the different grouping of flooding categories used by BNL and because the flow rates BNL has calculated included the modifications made to the CCW crossover and condensate coolers which were not explicitly given in Table 9.31 (page 9.150 of OPRA). The results of Table A.1 are compared with those of OPRA in Table 2.4 given in the main report. It is seen that BNL used finer grouping in the very large and large flood categories but the sum of their frequencies is similar to those of OPRA. Some changes are seen in the medium flood category.

Table A.2 lists specific comments on the OPRA fault trees generated during the BNL review. Table A.1 Flood-Initiating Events Categorization, Frequencies and Flow Rates

DESIGNATOR FLOW RATE

ERECHENCY

1. Very Large Floods (Greater than 170,000 gpm)
Very large flood 1 -- VL1: > 350,000 gpm
Very large flood 2 -- VL2: 170,000 to 349,000 gpm

1.1 Non Isolable Floods

		CLOTONICI OIL	Lanm	Turlet
	1.1.1 Gross rupture of any of 18 condenser outlet valves bodies	FVL2N	175,000	1.8×10^{-4}
	inclusion implacement on the provinctic values	EVI 1N	440.000	0.0.10-6
	1.1.3 Outlet valve fails removed and improverly reinstalled	FVLIN	440,000	9.0x10-0
	$(0.1 \times 3 \times 10^{-3} * 1.7 \times 10^{-2})$	FVL2N	175,000	5.1×10 ⁻⁶
	1.1.4 Outlet water boxes removed and improperly reinstalled and			
	impingement on pneumatic valve (10%)	FVLIN	440,000	4.0x10-7
	TOTA	AL: FVLIN	# 440,000	-1.0x10-5
11	그는 것 같아요. 그는 것이 가지 않는 것이 없는 것 같아요. 이 것 같아요. 것이 같아요. 이 것이 없는 것	FVL2N	= 175,000	1.8×10-4
1.	2 Isolable Flood Break on Inlet Side of Condenser			
	1.2.1 Gross rupture of any 18 condenser inlet pipes	FVL111	465,000	1.7×10-4
	1.2.2 Water boxes (inlet) removed for retubing and improperly			
	reinstalled (10%)	FVLIII	465,000	7.5x10-6(a
	1.2.3 Gross rupture of any condenser inlet valve body	FVL2II	180,000	1.1×10-5
	1.2.4 Inlet valve fails, removed and improperly reinstalled (10%)	FVL2II	180,000	6.9×10-6
	1.2.5 Inlet valve fails, removed and intake valve spuriously open	FVL211	180,000	2.1×10-6
	TOTA	AL: FVLIT	465,000	1.8×10-4
		EVI 211	180,000	2.0x10-5
1.	3 Isolable Flood, Break on Outlet Side of Condenser			
	1.3.1 Gross rupture of any of 18 condenser outlet pipes	FVL110	440,000	1.8x10-4
	improperly reinstalled (10%)	EVI 110	440,000	8.0×10-6
	TOTA	VI: FVL1TO	440,000	1.9x10-4
			110,000	1. 3410
- La	ge Floods (Greater than 60,000 and less than 169,000 gpm) rge Flood 1 L1: > $120,000$, < $169,000$ gpm			
La	rge Flood 2 L2: > 60,000, < 119,000 gpm			
2.	Non Isolable Floods			
	2.1.1 Outlet water boxes removed for retubing and improperly			
	reinstalled and impingement on putlet pneumatic valve (30%)	FL1N	132,000	1.1×10-5

		DESIGNATOR	FLOW RATE (gpm)	$\frac{FREQUENCY}{(yr)^{-1}}$
2.1.2	Gross rupture of any outlet expansion joint and impingement		1.11	
	on pneumatic valve (100%)	FL2N	56,000(d)	2.3x10-4
2.1.3	Large rupture of any outlet pipe and impingement (30%)	FL1N	132,000	2.7x10-5
2.1.4	Outlet isolation valve fails while expansion joint is out			
	for replacement	FL2N	56,000	3.4x10-5
2.1.5	Other cases of CCW open for maintenance and outlet			
	valve fails to open	FL2N	90,000	1.7x10-5
2.1.6	RCW outlet piping rupture (10%)	FL2N	80-115,000	5.4x10-6
2.1.7	RCW outlet valve rupture (10%)		77,000	2.5x10-6
2.1.8	Manways removed and 2 out of 54 improperly reinstalled.			
	and impingement on pneumatic valve (10%)		70,000	3.0x10-5
2.1.9	Two manways out of six (on floor-nonisolable) improperly			
	reinstalled (once in 10 years)(b)		100,000	< 10-6
2.1.10	Gross rupture of condensate coolers piping (10%)	н	90,000	3.6×10-5
2.1.11	Condensate coolers in maintenance and improperly			
	reinstalled outlet side (10%)	н	81,000(b)	6.0x10-5
	TOTAL:	FLIN	130,000	3.8x10-5
		FL2N	~60,000	4.1x10-4
Isolab	le Large Floods, Break on Inlet Side			
2.2.1	Large rupture of any of 18 Condenser inlet pipes (30%)	FLITI	140.000(b)	5.0x10-4
2.2.2	Gross rupture of crossover pipes (all assumed to be 42 in.) (10%) FL11I	140,000(b)	7.0×10-6
2.2.3	Gross rupture of any of 4 CCW crossover valves (10%)	F1211	81,000	2.5×10-6
2.2.4	Two or more condenser inlet manways improperly reinstalled (1)	0%) FL2II	70,000	6.0x10-4(a
2.2.5	Gross rupture of any inlet expansion joints (100%)	FL2II	76,000(d)	4.5×10-3
2.2.6	Removal of inlet water boxes and improper reinstallation (30%) FLIT	140,000	2.3×10-5(a
2.2.7	Removal of inlet expansion joints for replacement and intake	,		C
	valve opens spuriously	F1211	76,000	2.3×10-7
2.2.8	LPSW Lines		10,000	C
	- Gross runture of LPSW inlet lines (10%)	FL2II	103.000	3.8×10-5
	- Gross rupture of LPSW pump casing	FL 211	103,000	10-4
	- Gross rupture of any of 7 inlet valves (10%)	FL211	68,000	4.3×10-5
	- LPSW inlet valves fail, removed and improperly			
	reinstalled (10%)	FL2II	68,000	1.5x10-6
	LPSW sub1	total		1.8x10-4
	2.1.2 2.1.3 2.1.4 2.1.5 2.1.6 2.1.7 2.1.8 2.1.9 2.1.10 2.1.11 10 2.1.11 10 2.1.11 10 2.1.11 2.2.2 2.2.3 2.2.4 2.2.5 2.2.6 2.2.7 2.2.8	 2.1.2 Gross rupture of any outlet expansion joint and impingement on pneumatic valve (100%) 2.1.3 Large rupture of any outlet pipe and impingement (30%) 2.1.4 Outlet isolation valve fails while expansion joint is out for replacement 2.1.5 Other cases of CCW open for maintenance and outlet valve fails to open 2.1.6 RCW outlet piping rupture (10%) 2.1.7 RCW outlet valve rupture (10%) 2.1.8 Manways removed and 2 out of 54 improperly reinstalled. and impingement on pneumatic valve (10%) 2.1.9 Two manways out of six (on floor-nonisolable) improperly reinstalled (once in 10 years)(b) 2.1.10 Gross rupture of condensate coolers piping (10%) 2.1.11 Condensate coolers in maintenance and improperly reinstalled outlet side (10%) 2.2.4 Gross rupture of any of 18 Condenser inlet pipes (30%) 2.2.2 Gross rupture of any of 4 CCW crossover valves (10%) 2.2.4 Two or more condenser inlet mamways improperly reinstalled (11 2.2.5 Gross rupture of any inlet expansion joints (100%) 2.2.6 Removal of inlet water boxes and improper reinstallation (30%) 2.2.7 Removal of inlet water boxes and improper reinstallation (30%) 2.2.8 LPSW Lines Gross rupture of LPSW inlet lines (10%) Gross rupture of LPSW pump casing Gross rupture of any of 7 inlet valves (10%) LPSW subities (10%) 	DESIGNATOR 2.1.2 Gross rupture of any outlet expansion joint and impingement on pneumatic valve (100%) FL2N 1.3 Large rupture of any outlet pipe and impingement (30%) FL2N 2.1.4 Outlet isolation valve fails while expansion joint is out for replacement FL2N 2.1.5 Other cases of CCW open for maintenance and outlet valve fails to open FL2N 2.1.6 RCW outlet valve rupture (10%) FL2N 2.1.7 CKW outlet valve rupture (10%) FL2N 2.1.7 CKW outlet valve rupture (10%) " 2.1.7 CKW outlet valve rupture (10%) " 2.1.8 Manways removed and 2 out of 54 improperly reinstalled. and impingement on pneumatic valve (10%) " 2.1.9 Two manways out of six (on floor-nonisolable) improperly reinstalled (once in 10 y-ars)(b) " 2.1.11 Condensate coolers in maintenance and improperly reinstalled outlet side (10%) " TOTAL: FLIN FLIN 2.2.6 Gross rupture of any of 18 Condenser inlet pipes (30%) FL111 2.2.3 Cross rupture of any of 18 Condenser inlet pipes (30%) FL211 2.2.4 Two or more condenser inlet manways improperly reinstalled (10%) FL211 <td< td=""><td>2.1.2 Gross rupture of any outlet expansion joint and impingement on pneumatic valve (100%) FLOW RATE (gpm) 2.1.3 Large rupture of any outlet pipe and impingement (30%) FLIN 132,000 2.1.4 Outlet isolation valve fails while expansion joint is out for replacement FL2N 56,000(d) 2.1.5 Other cases of CCW open for maintenance and outlet valve fails to open FL2N 90,000 2.1.6 RCW outlet piping rupture (10%) FL2N 90,000 2.1.7 RCW outlet valve rupture (10%) FL2N 80-115,000 2.1.8 Manways removed and 2 out of 54 improperly reinstalled. and impingement on pneumatic valve (10%) "70,000 2.1.9 Two manways out of six (on floor-nonisolable) improperly reinstalled (once in 10 years)(b) "100,000 2.1.10 Gross rupture of condensate coolers piping (10%) "90,000 2.1.11 Condensate coolers in maintenance and improperly reinstalled outlet side (10%) "91,000(b) 2.2.2 Gross rupture of any of 18 Condenser inlet pipes (30%) FLIN 140,000(b) 2.2.3 Gross rupture of any of 4 CCW crossover valves (10%) FL2N 76,000 2.2.4 Two or more condenser inlet maways impoperly reinstalled (10%) FL2N 76,000 2.2.5 Gross rupture of any of 4 CCW crossover valves (10%) FL2N 76,000 2.2.4 Two or more condenser inlet maways imp</td></td<>	2.1.2 Gross rupture of any outlet expansion joint and impingement on pneumatic valve (100%) FLOW RATE (gpm) 2.1.3 Large rupture of any outlet pipe and impingement (30%) FLIN 132,000 2.1.4 Outlet isolation valve fails while expansion joint is out for replacement FL2N 56,000(d) 2.1.5 Other cases of CCW open for maintenance and outlet valve fails to open FL2N 90,000 2.1.6 RCW outlet piping rupture (10%) FL2N 90,000 2.1.7 RCW outlet valve rupture (10%) FL2N 80-115,000 2.1.8 Manways removed and 2 out of 54 improperly reinstalled. and impingement on pneumatic valve (10%) "70,000 2.1.9 Two manways out of six (on floor-nonisolable) improperly reinstalled (once in 10 years)(b) "100,000 2.1.10 Gross rupture of condensate coolers piping (10%) "90,000 2.1.11 Condensate coolers in maintenance and improperly reinstalled outlet side (10%) "91,000(b) 2.2.2 Gross rupture of any of 18 Condenser inlet pipes (30%) FLIN 140,000(b) 2.2.3 Gross rupture of any of 4 CCW crossover valves (10%) FL2N 76,000 2.2.4 Two or more condenser inlet maways impoperly reinstalled (10%) FL2N 76,000 2.2.5 Gross rupture of any of 4 CCW crossover valves (10%) FL2N 76,000 2.2.4 Two or more condenser inlet maways imp

			DESIGNATOR	FLOW RATE (gpm)	FREQUENCY (yr)-1
	2.2.9	Condensate Coolers Inlet			all and see a
		- Gross rupture of condensate coolers inlet			
		piping (10%)	FL2II	90,000	2.0×10-5
		- Maintenance on coolers and improper reinstallation (10%)	FL2II	90,000	6.0x10-5
		TOTAL:	FLIII		5.3×10-4
			FL2II		5.4x10-3
2	.3 Isolat	le Large Floods, Break on Outlet Side			
	2.3.1	Large rupture of any of 18 condenser outlet pipes (30%)	FL1I0	130,000	5.3×10-4
	2.3.2	Two or more manways improperly reinstalled (10%)	FL210	66,000	6.0x10-4
	2.3.3	Water boxes on outlet side improperly reinstalled (30%)	FL1I0	130,000	2.3x10-5
	2.3.4	Gross rupture of any outlet expansion joint (100%)	FL210	56,000(d)	4.5×10^{-3}
		TOTAL :	FLIIO	130,000	5.5x10-4
			FL210	66,000	5.1x10-3
	Medius Flo	od (Greater than 12 000, and less than 59 000			
	1 Nonico	Jahla Medium Floode			
	3.1.1	large runture of any 18 outlet condenser valve			
		hodies (30%)	EMN	52 000(b)	5 4×10-4(f)
	3.1.2	Medium rupture of any 18 outlet condenser		52,000.	J. TA10
		valve bodies (60%)	EMN	15 000(b)	1 1×10-3
	3.1.3	Medium runture of any 18 outlet nines with		15,000.	1.1.10
	01110	impingement on pneumatic valve (60%)	FMN	40,000	5 5x10-5
	3.1.4	Outlet water boxes removed for maintenance		10,000	0.00.00
		improperly reinstalled and impingement (60%)	FMN	40,000	2 5x 10-6
	3.1.5	Gross flow from improperly reinstalled manways (one)		10,000	
		and impingement (10%)	EMN	33,000	3.0x10-4
	3.1.6	Outlet valves failed, removed and improperly reinstalled		00,000	0.0410
		(large error = 30%)	EMN	40.000	1.7×10-5(f
		(medium error = 60%)	FMN	12,000	3.4×10-5
	3.1.7	One of six unisolable manways (on basement floor)		12,000	J. TALO
		improperly reinstalled (one in ten years) (10%)	FMN	50,000	6x10-5(a)
	3.1.8	Same as above (30%)	FMN	15,000	1-8×10-4(a
	3.1.9	Condensate Coolers:			
		- Gross rupture of condensate coolers, 12 outlet valves (10%)	FMN	55,000	7.6x10-6(f
		- Large rupture of the above (30%)	FMN	16,000	2.3×10-5

TONATOO

-

			DESIGNATOR	(gpm)	(yr)-1
		- Any outlet valve on condensate coolers failed removed			
		and improperly reinstalled (10%)	FMN	55,000	7x10-0
		- The above (30%)	FMN	16,000	2.1x10-5
		- Condensate coolers in maintenance and improperly			1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 - 1990 -
		reinstalled (30%)outlet side	FMN	27,000	1.8x10-4
		 Medium rupture of condensate coolers outlet piping 	FMN	27,000	1.1x10-4
		Subtota	l condenser co	olers	3.5x10-4
	3.1.10	Outlet isolation valve fails while inlet expansion			
		joint replaced	FMN	50,000	3.6×10-7
	3.1.11	Medium nonisolable flood from RCW outlet	FMN	30,000(b)	6.0x10-5
	3.1.12	LPSW Discharge			
		- Gross rupture of LPSW outlet piping (10%)	FMN	50,000	1.1x10-5
		- Large rupture of LPSW outlet piping (30%)	FMN	15,000(b)	3.3x10-5
		- Gross rupture of LPSW 4 manual valves (10%)	FMN	20,000	2.5x10-6
		TOTAL:	FMN	a an	2.7×10-3
.2	Isolab	le Medium Flood Inlet Side Break			
	3.2.1	Medium rupture of condenser inlet piping (10%)	FMII	40,000	1.0×10^{-3}
	3.2.2	Large rupture of an inlet valve body (30%)	FMII	54,000	3.4×10-5
	3.2.3	Medium rupture of an inlet valve body (60%)	FMII	16,000	6.8x10-5
	3.2.4	Inlet valve removed, and large reinstallation error (30%)	FMII	54,000	1.6×10-5
	3.2.5	The above (60%)	FMII	16,000	3.2x10-5
		CCW pip	ing subtotal		1.5x10-4
	3.2.6	Gross rupture of RCW inlet piping (10%)	FMII	34,000	1.6x10-5
	3.2.7	Gross rupture of RCW valve inlet (10%)	FMII	14,000	1.3×10^{-6}
	3.2.8	Unwatering system gross rupture of pipe (10%)	FMII	20,000	1.5x10-5
	3.2.9	Unwatering system gross rupture of valve (10%)	FMII	14,000	1.3×10^{-6}
		Unwater	ing and RCW sul	btotal	3.4×10-5
	3.2.10	Condenser Coolers			
		- Removal of valve CCW-75 for maintenance			
		and improper reinstallation (10%)	FMII	55,000	1.1×10-6
		- Coolers in maintenance and improperly reinstalled (30%)	FMII	27,000	1.8×10-4
		- Large flood from (CW-75 (30%)	FMII	16,000	3.3×10-6
		- Any of 6 valves in inlet of condensate coolers is			
		removed and improperly reinstalled (10%)	FMII	55,000	~10-6
		- Gross rupture of any of the 9 inlet valves (10%)	FMII	55,000	5.7×10-6
		MINGO INFORTE OF MILE OF STREET AND THE TAXE AND			

			DESIGNATOR	FLOW RATE	FREQUENCY
		그 것 같은 것 이 것 같은 것 같은 것 같은 것 같은 것 같은 것 같이 많이 많이 많이 했다.		(gpm)	$(yr)^{-1}$
		- Large rupture of any of the 9 inlet valves (30%)	FMII	16,000	1.7×10-5
		- Large rupture of condensate coolers inlet piping (30%)	FMII	27,000	5.9x10-5
		- 3 expansion joints of condensate coolers rupture (10%)	FMII	44,000	7.5x10-4(C)
		Conden	sers Coolers su	ubtotal	1.0x10-3
	3.2.11	Maintenance in CCW Piping			
		- Water boxes improperly reinstalled (60%)	FMII	40,000	5.0x10-5
		- Manways open for maintenance and improperly reinstalled (or	ne) FMII	30,000	6.0x10-3(d)
		- Inlet expansion joint out for replacement and outlet			
		isolation fails	FMII	60,000	3.4x10-5
		Subto	tal Maintenance	e of CCW	6.1x10-3
	3.2.12	CCW Crossover			an en alta de la complete de la complete
		- Medium flood from CCW crossover piping rupture (60%)	FMII	12,000(b)	4.2x10-5
		- Large flood from CCW crossover piping rupture (30%)	FMII	40,000	2.1×10-5
		- Large flood of any of 4 values of crossover (30%)	FMII	24,000	7.6x10-6
	3.2.13	LPSW/HPSW Systems			
		- Gross rupture of HPSW inlet piping (10%)	FMII	36,000	3.9x10-6
		- HPSW pump casing rupture	FMII	36.000	5.0x10-5
		- Gross rupture of a suction valve	FMII	36,000	1.3x10-6
		- Large rupture of any of 7 inlet LPSW valves (30%)	FMII	20,000	1.3×10^{-4}
		- Large flood due to pump casing rupture (30%)	FMII	25,000	3.0x10-4
		- Large rupture of any LPSW inlet piping (30%)	FMII	25,000	1.1x10-4
		- LPSW valve maintenance and improper reinstallation (10%)	FMII	20,000	4.0x10-6
		Subto	tal HPSW/LPSW	and Crossover	6.7×10-4
		TOTAL	FMII		8.0×10-3
3	Isolab	le Medium Flood Outlet Side Break			
	3.3.1	Medium rupture of any 18 condenser outlet lines (60%)	FMIO	40,000	1.1x10-3
	3.3.2	Water box removed and improperly reinstalled (60%)	FMIO	40,000	5.0x10-5
	3.3.3	Condenser opened for maintenance and improperly			
		reinstalled. One manway out of total of 54 (10%)	FMIO	33,000	6.0x10-3
	3.3.4	Any of the 18 outlet isolation valves tranferred			
		closed and ruptures expansion joint by water hammer (10%)	FMIO	50,000	4.2x10-3(e)
	3.3.5	CCW emergency discharge pipe rupture (10%)	FMIO	20,000	1.9x10-5
		T	OTAL EMIO:		1.1×10-2

3.

Notes

^aBNL value is higher than OPRA because unrecovered improper installation error probabilities on inlet and outlet sides are assumed to be the same.

bThis initiator added in the BNL reevaluation.

COPRA refers to this initiator but it did not appear in the fault tree developed for initiators grouping. dFlow rate taken from OPRA because it was specifically calculated by a detailed computer program. Similar to OPRA, it is assumed to be included in the large flood category even though flow rate is slightly less than the 60,000 gpm boun-dary.

eThis item was included in the inlet rather than the outlet side of the condenser in OPRA.

fThis initiator transferred in the BNL review. It appears in a higher flood category in OPRA.

Table A.2 BNL Comments on OPRA Flood-Initiating Events Fault Trees (Figures 9.63 to 9.70)

CW20AVOT: Fig. 9-64	Internal leak or rupture of a valve was assumed to be similar to the external leak or rupture failure probability with retubing for 1000 hr. This resulted in 10^{-5} for internal rupture with little or no impact.
CW20SWT: Fig. 9-64	340 hr used instead of 1000 hr for water box retubing, which was assumed in all other cases. The change has no impact.
CW1ORVT: Fig. 9-64	This event is calculated as 1.2×10^{-6} . It should be 1.2×10^{-5} . This has no effect and apparently is just a typo (see Figure 9.62, event TVL3).
CW75XXX: Fig. 9-65	All events related to valve rupture or improper assembly in the condensate coolers were calculated to have flow rate of 55,000 gpm and therefore were included only in the medium flood category (FMII).
CWIWBOXLH Fig. 9-65	Water box fails to be properly installed on inlet sidewas assumed to be the same as in the case for the outlet side, i.e., $5x10^{-3} \times 0.1 \times 0.3 = 9x10^{-4}$, rather than $9x10^{-5}$. This has no effect.
CWMWAYILH: Fig. 9-65	Failure of inlet side manways to be properly installed was assumed equal to the outlet side, i.e., $3x10^{-4}$. This has a small impact.
CWOWBOXM: Fig. 9-66	Typoshould be 2.5x10 ⁻² .
CWMWAYOLH: Fig. 9-67	Two or more misplaced manways were assumed to have lower proba- bility than one misplaced manway in the BNL review. Frequency reduced.
CWCONDM: Fig. 9-67	This case is a medium flood and was transferred to the FMN tree.
CW68LVVF: Fig. 9-67	Should be 2.5x10-6. A typo.
CW87LVVF: Fig. 9-67	This case and other condensate coolers valve failure have a flow rate that corresponds to a medium flood. Transferred to FMN in BNL review. Small effect.

This event appears in the case of FMN (Figure 9.70) for the case of large misinstallation error (30%). A case of gross misinstallation error (10%) was added to FLN by BNL, to be consistent with the methodology used in all other similar cases. This adds $2(yr^{-1}) \ge 3 \ge 10^{-5} = 6 \ge 10^{-5}$ which has a small effect.
This event, of an "outlet valve transfers closed and water hammer causes outlet expansion joint to rupture," is put in OPRA to FMII. BNL considered this to be part of FMIO, because break is on outlet side.
The "and gate" of CW20EJM and TVL3 is not appropriate. TL61 should be used instead of TVL3, because when expansion joint is out for replacement, the inlet manual valve (CW14) is closed, and if the intake MOV inadvertently opens, the closed inlet manual isolation valve will prevent flooding. This was assumed by BNL and this event was removed from consideration.
An event "any of 3 rubber expansion joints in the inlet of the condensate coolers ruptured" was added to FMII (flow rate cal- culated in OPRA to be 25,000 gpm). This has a frequency of 7.5×10^{-4} and is missing in OPRA Figure 9-68. It has some effect.
Outlet expansion joints fail due to water hammer following a pneumatic outlet valve transferring closed. This was added to FMIO. In OPRA it is included in FMII.
Medium rupture of discharge piping. Apparently 24- and 16-in. pipe contributions are not accounted for. Also gross rupture of the RCW 36-in. pipe is not included.
(TVU4 should be TVL14.) It should be a factor of about 30 smaller because = 12 hr (not 340 hr as assumed in calculating the 2.6 $\times 10^{-6}$ value). BNL also included internal rupture of CW20AVOT. Overall, no effect.

APPENDIX B

TURBINE BUILDING FLOODING: QUANTIFICATION OF THE SEQUENCES, CORE DAMAGE BINS, AND CONTAINMENT-SAFEGUARD STATES

This appendix summarizes the detailed results of the BNL review of the OPRA accident sequence quantification. The approach and data used to derive these results are discused in the main report. Figure 2.2 in the main report shows the eight core damage sequences considered. This appendix is arranged to show the results on a sequence-by-sequence basis according to Figure 2.2. Thus, in the BNL study, the same sequence is sometimes contributing to more than one bin (early and late bins) according to the time of the failure. In OPRA, the heading of a group of core damage sequences sometimes does not fully correspond to the sequences listed (e.g., bin II FQ_VUYX of the OPRA includes cases in which LPSW failed and is therefore not correctly covered by Y, or the bin IV FQ_SQ_VBUX of the OPRA includes cases in which B did not fail, i.e., \overline{B}).

For each sequence, the specific core damage accident sequences are given in an order of appearance similar to that in Appendix D.5.4 of the OPRA (page D-46). Thus, the numerals in square brackets refer in most cases to a corresponding core damage accident sequence in App. D.5.4. When [Ia], [Ib] is given, it is because some additional variants of sequences appear in the BNL review. Otherwise, some additional sequences are given at the end of a list with numbers higher than the last one appearing in App. D.5.4.

In most cases the same names are used by BNL in its list of the sequences. The names of events that do not appear in OPRA sequences can be found in Table 9-40 of the OPRA. Events that do not appear in that table but are derived in OPRA in Table 9-41, and appear in many sequences of the BNL review, are ASWLTF, CM12, SFMPPSH, REHSTK and LPSWPPSH.

The "ASWLTF" is the result of the quantification of the event tree in Figure 9-93 of OPRA, shown in Figure 2.3 of this report. This figure, with the values given here in Table 2.7, can be used for its quantification. The ASWLTF is the conditional probability of losing ASW long-term suction given any flood initiator.

The CM12 is the conditional probability that an early core damage would occur in one of the other units while it did not occur in unit 3. The quantification of this value in BNL review is based on BNL judgment on the sequences given in bins I and III, which do not cause a core damage in all three units. It is about two-thirds of the sum of the conditional probabilities in bins I and III for floods that can reach critical level 3 and one-half of that sum for all other initiators. A large contribution to CM12 comes from the case in which a stuck-open relief-valve sequence has occurred in unit 1 (or 2) and did not occur in unit 3.

The SFMPPSH is explained in OPRA, page 9-272, and used in the same way in the BNL review. This event corresponds to failure to replenish water to the spent-fuel pool to provide long-term suction to the SSF-makeup system.

The REHSTK is an event introduced in OPRA sequences which does not appear in Table 9.40. It stands for operator failure to replenish the elevated storage tank of the HPSW system after it is depleted.

The above two operator actions are quantified in OPRA at 0.01. On the basis of this and BNL judgment, the third case of operator action of this type--establishing long-term suction to the ASW pumps--was also quantified by BNL at 0.01 (OPRA uses 5×10^{-3} for this in its quantification in Figure 9.93). This change is in part responsible for differences in the values of the BNL ASWLTF and the corresponding SFAPPS used in OPRA.

The LPSWPPSH corresponds to SWPPS in OPRA, for which no specific value is given. It appears in the BNL review (without SFAPPS) in one sequence only. It stands for assuring suction to the LPSW.

Very few changes were made to the data given in Table 9.40. Two numerical errors were corrected (CCWAVC should be 0.049 and HSHPIF should be 0.018). CCWAVIC of OPRA was divided into its two components in the BNL review (CC20AVC + CWSRRYN) because the first is for a specific valve and the second was assumed to be a common cause for failure of all six valves. Also BNL calculated a slightly higher unavailability of HPR. Finally, the operator action RC4MVH was made time dependent. All other data were consistent with the generic OPRA data which were reviewed separately.

The fraction of LPSW system floods for large and medium floods given by L21F, L311F, etc. was calculated by BNL from its results given in Table A.1.

Sequence No. 1 $F_n Q_s Q_v BX_{11}$, Bin IV, Frequency = 1.8 x 10⁻⁵ yr⁻¹ (OPRA frequency = 1.9 x 10⁻⁵ yr⁻¹)

A. Background

These core damage sequences are characterized by a flood initiator that inundates the HPI in addition to the EFW and LPSW, requiring reliance on the SSF. The SSF eventually fails in the long term because of failure to maintain a suction supply for the ASW pump (ASWLTF) or for the SSF makeup pump (CM12 + SFMPPSH).

Core damage sequences [1a], [2a], and [3a] correspond to the sequences given in OPRA, page D-52. The other sequences are additional variations considered by BNL. Note that this sequence is ascribed to bin III in OPRA. It should be in bin IV (and the OPRA results are therefore conservative).

B. Core Damage Sequences

[1a] 1.4 x 10-5 FVL2N (1.8 x 10-4) * [CM12 (0.07) + SFMPPSH (0.01)] 1.0 x 10-6 FVL1N (1.0 x 10-5) * [CM12 (0.09) + SFMPPSH (0.01)] [16] 1.4 x 10⁻⁶ FLIN (3.8 x 10⁻⁵) * [CM12 (0.027) + SFMPPSH (0.01)] [2a] 5.0 x 10^{-7} FVL2N (1.8 x 10^{-4}) * [SSFMUPPR (1.4 x 10^{-3}) + SSFASWPPR (1.4 x 10-3)] 2.8 x 10-8 FVL1N (1.0 x 10-5) * [SSFMVPPR (1.4 x 10-3) + SSFASWPPR (1.4 x 10-3)] 1.1 x 10-7 FL1N (3.8 x 10-5) * [SSFMVPPR (1.4 x 10-3) + SSFASWPPR (1.4 x 10-3)7 [2b] 4.3 x 10⁻⁸ FL1I0(5.5 x 10⁻⁴) * [CWSRRVN(0.028)] * [SSFMVPPR (1.4 x 10-3) + SSFASWPPR (1.4 x 10-3)] [3a] 3.4 x 10-7 FVL110 (1.9 x 10-4) * [CCWMVC (0.036) + CWSRRVN (0.028) + CC20AVC (0.0035)] * [CM12 (0.008) + SFMPPSH (0.01)] 3.4 x 10-7 FVL111 (1.8 x 10-4) * [CCWMVC (0.036) + CWSRRVN (0.028) + CCWAVC (0.021)] . [CM12 (0.008) + SFMPPSH (0.01)] 1.8 x 10-8 FVL2II (1.8 x 10-5) . [CCWMVC (0.035) + CWSRRVN (0.028)] • [CM12 (0.006) + SFMPPSH (0.01)] 1.7 x 10-7 FL1IO (5.5 x 10-4) . [CWSRRVN (0.028)] • [CM12 (0.001) + SFMPPSH (0.01)] [3b] 3.7 x 10⁻⁸ FL210 (5.1 x 10⁻³) . [CWSRRVN (0.028) + CC20AVC (0.0035)] • [HSHPIF (1.8 x 10-2) + SWHPSWF (4.8 x 10-3)] • [SFMPPSH (0.01)]

1.2 x 10⁻⁷ FL2II (5.4 x 10⁻³) * [CCWMVC (0.036) + CCWAVC (0.021) + CWSRRVN (0.028] * [HSHPIF (1.8 x 10⁻²) + SWHPSWF (4.8 x 10⁻³)] * [SFMPPSH (0.01) + CM12 (0.001)] 1.4 x 10⁻⁷ FL2N (4.1 x 10⁻⁴) * [HSHPIF (1.8 x 10⁻²) + SWHPSWF (4.8 x 10⁻³)] * [CM12 (0.005) + SFMPPSH (0.01)] * [CM12 (0.005) + SFMPPSH (0.01)] * [CM12 (0.034) + (M321F (0.042) + CCHXFPF (1.8 x 10⁻³)] * RESW12 (0.0)] * [HSHPIF (1.8 x 10⁻²) + SWHPSWF (4.8 x 10⁻³)] * [SFMPPSH (0.01)] 2.2 x 10⁻⁸ FMIO (1.1 x 10⁻²) * [HPIF (2 x 10⁻⁴)] * [SFMPPSH (0.01)]

C. Containment Safeguard States

Failure of RBCS and RBSS, in cases: [1], [2], [3a] Failure of RBCS only, in cases: [3b], [4].

D. Discussion

The OPRA sequence [3] (OPRA, page D-52) has a higher frequency because a high value is derived for CM12. A smaller CM12 value was derived by BNL in Table 2.7 for the same initiator (FVLI).

In the BNL review, additional core damage sequence contributors compensate for the above reduction so that the total frequency in both cases is similar.

The main difference for this sequence is that BNL considers it to be in bin IV, i.e., late core damage.

Sequence No. 1A $F_n Q_s \overline{Q_v} BY\overline{X}$, Bin IV, Frequency = 4.6 x $10^{-6} yr^{-1}$ (OPRA frequency = 1.0 x $10^{-5} yr^{-1}$)

A. Background

The core damage accident sequences below are the BNL review sequences which correspond to the OPRA case of Bin IV FQ_Q_BUX in Appendix D, page D-54. Even though the OPRA refers to B, all the sequences listed are in fact cases in which the B function is a success, i.e., SSF is available for the short term, preventing the loss of RCS heat removal via the steam generators. Therefore, they are considered by BNL as part of sequence No. 1 (see Figure 2.2). In all these sequences the LPSW is lost because the flood exceeds critical level 2 or fails LPSW; as a consequence, the RBCS is lost and the RBSS would be actuated to remove heat from the containment, depleting the BWST. Because there is no loss of RCS inventory, the ASW will be able to remove decay heat while HPI will provide makeup. When the lake is lowered for flood isolation or when the ASW pump fails to run (both assumed by BNL to occur between 12 and 72 hr), the unit would be put in the feed-and-bleed mode, the remainder of the BWST will soon be depleted, and HPR will fail in a few hours if cooling by LPSW or ASW is not recovered. OPRA does not consider a recovery in these sequences. BNL believes that time to fail the HPR will be greater in an accident sequence starting at ~15 hr, compared with the sequences in which HPR is needed at 2 hr. Therefore, a recovery factor of 0.5 was used in BNL requantification of these core damage sequences.

OPRA sequences [1], [2], [3], [4], [7], [8] on pages D-54 and D-55 correspond to the following sequences.

B. Core Damage Sequences

[1]	2.5 x 10 ⁻⁶	FL2N (4.1 x 10 ⁻⁴) * [ASWLTF (0.011) + SSFASWPPR (1.4 x 10 ⁻³)] * 0.5	
[2a]	9.4 × 10-7	FL2II $(5.4 \times 10^{-3}) * [CCWMVC (0.036) + CCWAVC (0.021) + CWSRRVN (0.028)] * [ASWLTF (0.0027) + SSFASWPPR (1.4 × 10^{-3})] * 0.5$	
[22]	8.3 x 10 ⁻⁸	FL1II (5.3 x 10 ⁻⁴) * [CCWMVC (0.036) + CWSRRVN (0.028)] * [ASWLTF (3.5 x 10 ⁻³) + SSFASWPPR (1.4 x 10 ⁻³)] * 0.5	
[3]	1.2 × 10-7	FL2IO (5.1 x 10 ⁻³) * [CWSRRVN (0.028) + CW20AVC (3.5 x 10 ⁻³)] * SSFASWPPR (1.4 x 10 ⁻³) * 0.5	
[4]	1.6 x 10-7	FL2II (5.4 x 10 ⁻³) * [L2IF (0.013) + (L311F (0.005) + CCHXFPF (1.8 x 10 ⁻³)) * RESW12 (0.1)] * [ASWLTF (2.7 x 10 ⁻³ * SSFASWPPR (1.4 x 10 ⁻³)] * 0.5)

[5], [6] See sequence No. 18.

[7] 1.3×10^{-7} FMN (2.7 x 10⁻³) * [{M361F (0.043) + CCHXFPF (1.8 x 10⁻³)} * RESW12 (0.1) + 0.006] * [ASWLTF (0.008) + SSFPSWPPR (1.4 x 10⁻³)] * 0.5 [8] 6.1 x 10⁻⁷ FMII (8.0 x 10⁻³) * [M11F (0.034) + {M321F (0.042)} + CCHXFPF (1.8 x 10⁻³)} * RESW12 (0.1)] * [ASWLTF (2.6 x 10⁻³) + SSFASWPPR (1.4 x 10⁻³)] * 0.5

C. Containment Safeguard States

Failure of RBCS in all the above cases.

D. Discussion

The heading of B sequence in OPRA is inappropriate. BNL considers this sequence to be part of sequence No. 1 of Figure 2.2. In general, the results are similar, but BNL has made two changes that compensated each other: 1) consideration of operator error in the recovery of long-term ASW suction and 2) consideration of some recovery of SSF, feedwater, or BWST because longer grace times are available.

Note that BNL assumes recovery of LPSW in unit 2 also for the case in which condensate coolers break occurred on their outlet side (see FMN * M361F sequence [7]).

The reduction by a factor of 2 in the BNL review results is because the frequency of FLN is lower in the BNL reevaluation, and because in sequences [2] and [3] SFAPPSLN for FLN is used in OPRA, rather than a value about one third smaller, which results from using OPRA, Table 9.41 and Figure 9.93, for FLII.

Sequence No. 1B $F_n \overline{Q}_S \overline{Q}_V BYX$, Bin IV, Frequency = 1.2 x 10⁻⁵ yr⁻¹ (OPRA frequency = 7 x 10⁻⁶ yr⁻¹)

A. Background

These core damage sequences are the same as in sequence 1A, but LPSW is not lost because the flood level does not reach critical level 2 and does not fail the LPSW system. The SSF is assumed available and so is HPI makeup. If the lake is lowered to isolate the flood and ASW and LPSW, suction may be lost, or if the ASW pump fails to run, the operator will have to establish feed and bleed. When it depletes the BWST, suction to LPSW and ASW may not be recovered either because of operator error (FCC2 or FCI) or because of core damage in another unit (CM12). There is a longer grace time in this case because the feed-and-bleed mode is required in the long term when decay heat is a fraction of 1%. A recovery of 0.5 is applied by BNL for this reason. The sequences correspond to [5] and [6] on page D-55 of the OPRA.

B. Core Damage Sequences

[5a] 2.0 x 10⁻⁸ FL1II (5.3 x 10⁻⁴) * [CCWAVC (0.021)] * [ASWLTF (3.5 x 10⁻³)] * 0.5

[6a] 1.1 x 10⁻⁵ FMN (2.7 x 10⁻³) * ASWLTF (0.008) * 0.5

Additional Sequences

[6b] 8.8×10^{-7} FMII (8.0 x 10⁻³) * [CCWMVC (0.036) + CCWAVC (0.021) + CWSRRVN (0.028)] * [ASWLTF (2.6 x 10⁻³)] * 0.5

C. Containment Safeguard States

Failure of the RBCS.

D. Discussion

The above core damage sequences are additional cases of sequence 1, in which both the LPSW and ASW are available in the short term. They fail only after more than 10 hr. as the result of actions taken to lower the lake for the isolation of the flood.

The results for sequence [6] in the BNL reevaluation are higher than those of the OPRA. BNL calculated some contribution for the FMII flood with isolation, but in which isolation malfunctions have occurred. A case of medium flood with isolation failure is considered by BNL to be similar to a nonisolable flood apart from the possibilities to locate and isolate this flood at a later time, a possibility that is taken into account in the ASWLTF.

The difference in sequence [6a], which corresponds to sequence [6] in the OPRA, is due to (1) a higher frequency of FMN calculated by BNL as shown in Appendix A and (2) a higher value used for failure to provide ASW suction after lowering the lake (0.01 in BNL instead of 0.005 in OPRA).

Sequence No. 2 $F_n \overline{O_s} \overline{O_v} B \overline{U_3} Y X$, Bin IV, Frequency = 1.0 x 10⁻⁷ (OPRA frequency: no sequence in OPRA)

A. Background

This sequence is referenced on pages D-54 and D-55 of the OPRA. However, as discussed before (sequences 1A and 1B), the B is not included in the core damage sequences shown in the OPRA under this heading. The following sequences are the results of the sequence in the event tree shown in Figure 2.2.

These sequences are characterized by a medium flood which does not affect the LPSW. SFW and ASW are lost in the first 30 minutes. Feed and bleed is initiated and the BWST suffices for 12 hr because the RBCS is available. After 12 hr. HPR is initiated if ASW was not recovered in the meantime. Lowering the lake for the purpose of isolating the flood can cause loss of LPSW suction which is needed for successful HPR.

The difference between this sequence and sequence 1B is that here ASW failed at the beginning of the accident and was not recovered, whereas in sequence 1B the ASW cooling is successful until the lake is lowered and interrupts the backflow. Before lowering the lake, a CCW flood to at least one unit's LPSW must be maintained or else ASW must be recovered. The sequences that follow correspond to [5] and [6] of sequence 1B described before, but with B rather than \overline{B} .

- B. Core Damage Sequences
- [5] (None significant)
- [6] 9.2 x 10⁻⁸ FMN (2.7 x 10⁻³) * [SSF30H (0.03) * NORECV (0.03) + SSFASWF (2.5 x 10⁻³)] * [LPSWPPSH (0.01)]

C. Containment Safeguard State

Failure of RBCS.

D. Discussion

Contribution from sequence No. 2 of Figure 2.2 is very small and practically covered by sequence [6] of 1B. Note that if the lake is lowered before recovery of ASW or without improvising any flow of water from the lake to LPSW suction, a core damage state will result. In quantifying LPSWPPSH = 0.01, it was assumed that the above would be recognized and the action to isolate the flood would be postponed until ASW is available or LPSW suction is assured. Sequence No. 3 $F_n \overline{Q}_s \overline{Q}_v \overline{BU}_3 YX$, Bin IV, Frequency = 3.0 x 10⁻⁷ yr⁻¹ (OPRA frequency: no sequences in OPRA)

A. Background

This sequence is referenced on page D-54 of the OPRA, similarly to sequence No. 2 discussed before. The difference from sequence 2 is that in this case critical level 2 is reached by the flood and LPSW is lost early in the sequence. It also includes the sequences in which LPSW is lost because of a break in the system or because of diversion of the backflow that provides suction to the LPSW.

The following sequences correspond to [1] to [4] and [7] to [8] on the OPRA, pages D-54 and D-55, but with B rather than \overline{B} . Thus, the sequences correspond to sequence 1 A discussed before with B instead of \overline{B} .

In all the following sequences, SSF was not initiated within 30 min, necessitating feed and bleed and RBSS actuation. However, SSF was recovered before 2 hr. i.e., before depletion of BWST and ASW cooling of the RCS through the SGs is reestablished.

In the long term, the ASW fails to run or the lake may be lowered for flood isolation and the operator fails to assure ASW suction. Because the BWST is practically empty, feed and bleed would not be successful.

B. Sequences

[1] 1.5×10^{-7} FL2N $(4.1 \times 10^{-6}) * [SSF30H (3 \times 10^{-2})] * [ASWLTF (0.011) + SSFASWPPR (1.4 \times 10^{-3})]$ [2] 5.7×10^{-8} FL2II $(5.4 \times 10^{-3}) * [CCWMVC (0.036) + CCWAVC (0.021) + CWSRRVN (0.028)] * [SSF30H (3 \times 10^{-2})] * [ASWLTF (2.7 \times 10^{-3}) + SSFASWPRR (1.4 \times 10^{-3})]$ [3] 6.8×10^{-9} FL2IO $(5.1 \times 10^{-3}) * [CCWSRRVN (0.028) + CC20AVC (3.5 \times 10^{-3})] * SSF30H (0.03)] * [SSFASWPPR (1.4 \times 10^{-3})]$ [4] 9.1×10^{-9} FL2II $(5.4 \times 10^{-3}) * [L21F (0.013) + (L311F (0.C35) + CCHXFPF (1.8 \times 10^{-3})] * RESW12 (0.1)] * [SSF30H (3 \times 10^{-2})] * [ASWLTF (2.7 \times 10^{-3}) + SSFASWPPR (1.4 \times 10^{-3})]$ [5]-[6] see sequence No. 2

[7] -1.0×10^{-8} FMN (2.7 x 10^{-3}) * [small, because recovery of LPSW from unit 2 is assumed]

[8] 3.7×10^{-8} FMII (8.0 x 10^{-3}) * [M11F (0.034) + {M321F (0.042) + CCHXFPF (1.8 x 10^{-3})} * RESW12 (0.1)] * [SSF30H (3 x 10^{-2})] * [ASWLTF (2.2 x 10^{-3}) + SSFASWPPR (1.4 x 10^{-3})]

C. Containment Safeguard States

Failure of RBCS.

D. Discussion

The contribution from sequence No. 3 of Figure 2.2 is small and practically covered by sequence 1A. These sequences are a variation of the sequences 1A in which ASW is initiated in time, but the BWST water would also be partially consumed by the actuation of the RBSS for containment heat removal (LPSW is lost and consequently RBCS). Sequence No. 3 $F_p \overline{Q_s} \overline{Q_y} \overline{B} \overline{U_3} Y X_{36}$, Bin III, Frequency = 1.1 x 10⁻⁵ yr⁻¹ (OPRA frequency = 1.2 x 10⁻⁵ yr⁻¹)

A. Background

This sequence is referenced on page D-53 and D-54, and is included in bin III in the OPRA as well. It is a case of a large flood that results in a loss of feedwater and LPSW. SSF fails to start or operator fails to initiate it in 2 hr. Thus, when the BWST is depleted by the RBSS and the HPI in 2 hr., it is assumed that the HPR will not be successful, resulting in an early core damage.

The sequences that follow correspond to the [1] through [3] on pages D-53 and D-54.

B. Core Damage Sequences

[1] 3.1 x 10⁻⁶ FL2N (4.1 x 10⁻⁴) * [SSFASW2H (5 x 10⁻³) + SSFASWF (2.5 x 10-3)] [2] 3.6 x 10^{-6} FL2II (5.4 x 10^{-3}) * [CCWMVC (0.036) + CCWAVC (0.021) + CWSSRRVN (0.028) + CCWI20H (3 x 10-3)] * [SSFASW2H (5 x 10-3) + SSFASWF (2.5 x 10-3)] FL1II $(5.3 \times 10^{-4}) * [CCWMVC (0.036) + CWSRRVN (0.028)] * [SSFASW2H (5 x 10^{-3})]$ 2.5 x 10-7 + SSFASWF (2.5 x 10-3)] [3] 1.2 x 10⁻⁶ FL2IO (5.1 x 10⁻³) * [CWSRRVN (0.028)] + CC20AVC (3.5 x 10-3) * [SSFASW2H (5 x 10-3) + SSFASWF (2.5 x 10-3)] [4] 5.7 x 10^{-7} FL2II (5.4 x 10^{-3}) * [L2IF (0.013) + {L311F (0.005) + CCHXFPF (1.8 x 10⁻³)) RESW12 (0.1)] * [SSFASW2H (5 x 10⁻³) + SSFASWF (2.5 x 10-3)7 [5] 2.3 x 10⁻⁶ FMII (8.0 x 10⁻³) * [M11F (0.034) + (M321F (0.042)) + CCHXFPF (1.8 x 10⁻³)) * RESW12 (0.1)] * [SSFASW2H (5 x 10⁻³) + SSFASWF (2.5 x 10-3)]

It is assumed that the LPSW from unit 2 would be connected within 2 hr, with 0.1 probability of failure.

Additional sequences

[6] 1.4×10^{-7} FVL110 $(1.9 \times 10^{-4}) * [CCWI5H (0.1)]$ * [SSFASW2H (5×10^{-3}) + SSFASWF (2.5×10^{-3})] 1.3 x 10⁻⁷ FVL111 (1.8 x 10⁻⁴) * [CCW15H (0.1)] * [SSFASW2H (5 x 10⁻³) + SSFASWF (2.5 x 10⁻³)] [7] 2.0 x 10⁻⁷ FMN (2.7 x 10⁻³) * [M361F (0.043) + CCHXFPF (1.8 x 10⁻³)] * RESW12 (0.1) + 0.006] * [SSFASW2H (5 x 10⁻³)] + SSFASWF (2.5 x 10⁻³)]

C. Containment Safeguard States

Failure of RBCS in all cases.

D. Discussion

The core damage sequences for accident sequence No. 3 are similar in both the OPRA and the BNL review. The change of groupings of flood categories and consideration of flow rates after isolation malfunctions in intake and outlet are seen to have resulted in some small changes only in individual sequences. However, the sum of these sequences remained unchanged.

Note that sequence No. 3 has some contribution to bin IV when the SSF is successfully initiated at 2 hr. This is given in the previous page for sequence No. 3--contribution to bin IV which amounts to about 3×10^{-7} . This is small compared to the contribution to bin III. Therefore, sequence no. 3 is mainly contributing to bin III, and this was correctly accounted for in the OPRA.

Sequence No. 4 $F_n Q_s Q_v BU_3$, Bin III, Frequency = 2.6 x 10⁻⁶ yr⁻¹ (OPRA frequency = 5.6 x 10⁻⁶ yr⁻¹)

A. Background

This sequence is shown on pages D-52 and D-53. It is also included in the OPRA in bin III. It is a case of a large flood that causes the failure of EFW. Operator failure to initiate feed and bleed or the SSF ASW system in 30 minutes results in a loss of core cooling. BNL has judged that the operator error in not isolating the flood and in not initiating the HPI, and in not manning and initiating SSF, all three being clearly separated instructions of the flooding emergency procedure, is smaller than considered in OPRA.

B. Core Damage Sequences

The following sequences correspond to the [1] through [6] sequences in OPRA pages D-52 and D-53.

[1] 1.2 x 10⁻⁷ FL2N (4.1 x 10⁻⁴) * [UTHPIH * SSF30H (3 x 10⁻⁴)]

 $(SSF30H = 3 \times 10^{-4}: 30 \text{ minutes from the loss of EFW})$

[2]	1.4 x	10-7	FL2II	(5.4	х	10^{-3})	*	(CCWMVC	(0.036) + CCWAVC (0.021)
							+	CWSRRVN	$(0.028) + CCW30H (10^{-3})$]
		1111					*	[UTHPIH	* SSF30H (3 x 10-4)]
	3.6 x	10-8	FLIII	(5.3	х	10-4)	*	[CCWMVC	(0.036) + CWSRRVN (0.028)
							+	CCWI15H	(3×10^{-3})]
							*	[UTHPIH	* SSF15H (10-3)]

 $(SSF15H = 10^{-3}: 15 \text{ minutes from the loss of EFW})$

- [3] 1.3×10^{-7} FL2IO (5.1 x 10^{-3}) * [CCWMVC (0.036) + CCWAVC (0.021) + CCWI30H (10^{-3}) + CWSRRVN (0.028)] * [UTHP2H * SSF30H (3 x 10^{-4})]
- [4] 2.7 x 10⁻⁷ FMN (2.7 x 10⁻³) * [UTHPIH * SSF60H (1.0 x 10⁻⁴)]

(SSF60H = 10⁻⁴: 60 minutes from loss of EFW)

- [5] 8.0 x 10⁻⁷ FMII (8.0 x 10⁻³) * [CCWI60H * UTHPIH * SSF60H (1.0 x 10⁻⁴)]
- [6] 1.1 x 10⁻⁶ FMIO (1.1 x 10⁻²) * [CCWI60H * UTHPIH * SSF60H (1.0 x 10⁻⁴)]
- C. Containment Safeguard States

Failure of RBCS in cases [1] through [3] No failure of RBCS in cases [4] through [6]

D. Discussion

The sequences of the OPRA were also found in the BNL analysis of this sequence No. 4. BNL considered that given the time until EFW fails (30 to 60 minutes) the probability of operator failure to initiate HPI and then also the SSF is smaller than considered by the OPRA; as explained before, this is one of the first steps in the flooding emergency procedures. In addition, BNL treatment considered that a large flood could not continue unnoticed for a long time and the operator will trip the CCW pumps with the probability of CCWI20H = 3×10^{-3} independent from his actions to initiate HPI and SSF. Therefore, for the large floods the probability of isolation malfunctions was included in BNL sequences. For the medium isolable floods, BNL followed the OPRA assumption that these floods may continue unnoticed for long times, and did not include the isolation measures malfunction. However, even if the isolation was included in these cases, it does not prevent a backflow of 20,000 gpm coming to the break if the LPSW and HPSW do not operate, so in this sequence, the operator error in controlling the LPSW flow may also be assumed.

Sequence No. 5 $F_n \overline{Q}_s Q_v \overline{UYX}_{54}$, Bin II, Frequency = 4 x 10⁻⁷ yr⁻¹ (OPRA frequency: 1.3 x 10⁻⁶ yr⁻¹)

A. Background

This sequence is shown on pages D-50 and D-51 in the OPRA. It is also a bin II there. However, even though it refers to Y in its heading, the first three sequences are cases of loss of LPSW and, therefore, are covered by BNL as part of sequence No. 6--sequence $F_nQ_sQ_vUYX_{54}$ (see next sequence). All other sequences, [4], [5], and [6], are considered below.

The sequences are characterized by a medium flood that does not exceed critical level 2; i.e., only EFW is lost. Since for medium flood the EFW is lost in a time period longer than 50 minutes, it is assumed that if the PORV is opened, it is sufficient to relieve the pressure without lifting the SRVs. Thus, only operator failure to open the PORV block valve is considered. BNL considered that in a time frame of about 1 hr, the stress on the operator will be less than at the beginning of the transient, and the block valve will be opened in 90% of the cases. If the block valve remains closed, the SRV will be challenged by liquid, also because HPI may be on for injection according to procedures. A 0.1 probability of any SRV stuck open was used as in the OPRA. Then. HPR will be required with additional ASW or LPSW cooling. If HPR initation in 12 hr fails or the lake is lowered, with failure to assure the ASW and LPSW suction, a core damage state will result.

B. Core Damage Sequences

[1]-[2]	included II.	in seque	nce No	. 6 ($F_nQ_sQ_vUY_{54})$ contribution to bin
[4]	2.7	× 10-7	FMN (2.	7 x 10	- ³) *	[RC4MV4 (0.1) * RCSRVLC (0.1)] [HPRF (2 x 10 ⁻³) + XHPR12H (3 x 10 ⁻⁴) ASWLTF (0.008)]
[5]	3.3	x 10-8	FMII (8	3.0 x 1	0- ³)	<pre>* [CCWMVC (0.036) + CCWAVC (0.021) + CWSRRVN (0.028)] * [RC4MV4 (0.1) * RCSRVLC (0.1)] * [HPRF (2 x 10⁻³) + XHPR12H (3 x 10⁻⁴) + ASWLTF (2.6 x 10⁻³)]</pre>
[6]	2.2	x 10 ⁻⁸	FMIO (1	.1 x 1	0-2)	<pre>* [CCWMVC (0.036) + CCWAVC (0.021) + CWSRRVN (0.028)] * [RC4MV4 (0.1) * RCSRVLC (0.1)] * [HPRF (2 x 10⁻³) + XHPR12H (3 x 10⁻⁴)</pre>
Addit	iona	al Sequend	ces			
[7]	5.3	× 10-8	FVL1II	(1.8 x	10-4	*) * [CCWI1H (0.5) * {1 - CCWI5H}] * [(RC66RV0F1 + RC4MV1] (0.88)]

* [RCSRVLC (0.1)]

* [HPRF (2 x 10⁻³) + HPR12H (3 x 10⁻⁴) + ASWLTF (0.006)]

Flood reach level 1 and is "isolated" before level 2.

[8]	1.1 x	10-8	FL2I0	(5.1 x	(10^{-3})	*	[CCWMVC (0.036) + CCWAVC (0.021)]	
						*	[RC4MV4 (0.17) * RCSRVLC (0.1)]	
						*	[HPRF (2 x 10 ⁻³) + XHPR12H (3 x 10 ⁻⁴)]	

C. Containment Safeguard States

In these sequences, RBSS is assumed to fail if HPR fails, and RBCS is assumed to fail if long term suction supply to the LPSW fails. The relative fraction can be calculated for each sequence separately.

D. Discussion

In core damage sequences [1] and [2], the LPSW is lost. They are discussed in sequence No. 6. Core damage sequence [3] is a special case of a medium flood with failure of HPR after 12 hr, similar to sequences [4] to [6] of page D-51. For a large flood (FLII or FLIO) to have a flow rate corresponding to a medium flood, it needs to be "isolated." Thus, the OPRA sequence refers to an isolated flood, i.e., with all inlet and outlet valves closed, and the break flow then corresponds to the backflow through the condensate coolers. However, in another place (page 9-157) OPRA apparently considers that the LPSW (which is available in this sequence) will consume a large part of the backflow so that the break discharge will be about 10,000 gpm which is insufficient to reach critical level 1. This assumption was confirmed in the BNL meeting with Duke Power. Furthermore, in unit 1 there is no backflow through the condensate coolers (and RCW backflow is less than 12,000 gpm), and in unit 3 the backflow can be isolated and LPSW cooling supplied from unit 2. Thus, even if the LPSW consumption is not considered (as in the case of sequence [3]), only one third of the contribution may not be isolated before the loss of EFW. Note that at a flow rate of ~15,000 gpm, the time to reach critical level 1 is more than 3 hr. Furthermore, in the BNL review the operator failure to open the block valve is assumed to be smaller than in OPRA (by a factor of -2), so this sequence is insignificant in the BNL review.

Note that sequence [3] on page D-51 and sequence [5] on page D-55 are the same except for the SRV stuck open or successfully reclosed, respectively. However, in the latter case the isolation measure malfunctions were considered by OPRA.

Sequence [4] is similar in OPRA and the BNL review, and is affected by a higher frequency of FMN in the BNL review. Sequences [5] and [6] in the OPRA do not give credit to the backflow isolation in unit 1 and the potential isolation in unit 3 as well as to the LPSW consumption of part of the backflow before it reaches the break (the important inlet and outlet breaks are at a higher elevation than LPSW suction allowing for about 5 ft head for the LPSW suction).

The above modifications by BNL resulted in a reduced contribution in the BNL reevaluation.

Sequence No. 6 $r_n \overline{Q}_S Q_V \overline{U}YX_{54}$, Bin II, Frequency = $3x10^{-7} \text{ yr}^{-1}$ (OPRA Frequency = $3x10^{-7} \text{ yr}^{-1}$)

A. Background

This sequence is shown on pages D-50 and D-51 in the OPRA. It is also a Bin II there. However, the heading for this sequence on the OPRA, page D-50 is \overline{Y} although in fact the sequence described in cases [1], [2], and [3] are Y sequences. These sequences are considered here by BNL.

The sequences are characterized by a large flood that exceeds critical level 2 but not critical level 3. Thus, EFW, LPSW, and HPSW pumps are assumed to be failed by the flood. The flood would fail EFW in about 20 to 30 minutes; therefore, the probability of challenging the SRVs with the PORV open is very small. BNL accepted OPRA considerations that in 80% of the cases the block valve is closed and the operator probability of not attempting to open the block valve is 0.2; thus a probability of 0.1 for a SRV sticking open is assumed. Cooling is provided by HPI with the elevated storage tank supplying the HPSW tank supplying cooling to the HPI motor bearings. These sequences consider the failure in the long term either of HPI pumps because the elevated tank is depleted and fails to be replenished [REHSTK(0.01)], or because of failure of ASW cooling of the SGs due to loss of suction or pump failure to run, i.e., no means of decay heat removal is available, since LSPW pumps are flooded.

B. Core Damage Sequences

[1] 7.0 x 10⁻⁸ FL2N (4.1 x 10⁻⁴) * (RC4MV4 (0.17) * RCSRVLC (0.1)] * [REHSTK (0.01]

Additional Sequence

	7.8x10 ⁻⁸	FL2II (5.4 x 10-) * [CCWMVC (0.036) + CCWAVC (0.021) + CCSRRVN (0.028)] * (RC4MV4(0.17) * RCSRVLC (0.1) * REHSTK (0.01)]	ALL LAND
[2]	8.6x10 ⁻⁸	FL2N (4.1 x 10 ⁻⁴)	<pre>* [(RC4MV4 (0.17) * RCSRVLC (0.1)] * [(ASWLTF (0.011) + SSFASWPPR (1.4×10⁻³)]</pre>	

Additional Sequence

3.2 x 10 ⁻⁸	FL?II	(5.4)	(10^{-3})	*	[(CCWMVC (0.036) + CCWAVC (0.021)	
				+	CCWRRVN (0.028)]	
				*	[RC4MV4 (0.17) * RCSRVLC (0.1)]	
				*	[ASWLTF (2.7 x 10-3)	
				+	SSFASWPPR (1.4 x 10-3)]	

2.5 x 10⁻⁸ FVL1II (1.8 x 10⁻⁴) * [CCWI5H (0.1) * {1-CCW15H}] * [(RC66RV0F1 + RC4MV1) (0.88)] * [RCSRVLC (0.1)] * [REHSTK (0.01) + ASWLTF (0.006)]

C. Containment Safeguard States

RBCS is unavailable as a result of the loss of LPSW.

D. Discussion

The differences between OPRA and BNL in these core damage sequences are small. They are a result of a lower frequency in BNL for the FLN flood initiator. This reduction is compensated by the inclusion in the BNL review of flood sequences of FLII with isolation measure malfunctions which are not shown in the OPRA. Sequence No. 6 $F_n \overline{Q}_S Q_V \overline{U}YX_{57}$, Bin I, Frequency = 2.3 x 10⁻⁷ Yr⁻¹ (OPRA Frequency = 3.6 x 10⁻⁷ Yr⁻¹)

A. Background

This sequence is shown on page D-50 in the OPRA. It is a case of a large flood that does not reach critical level 3, and therefore, does not fail the HPI system. LPSW is lost, and the EFW is lost very early causing a demand for pressure relief through the PORV or the SRVs. If the PORV is not sufficient to relieve the pressure or the block valve is shut and not opened in time, then the SRVs will be challenged with a probability of 0.1 assumed for their sticking open. The BWST will be depleted in about two hours by the HPI and RBSS, and both ASW and HPR (without LPI coolers) must operate to avoid loss of cooling. Because loss of cooling may occur as early as two hours into the sequence, it is assigned to early core damage class--bin I.

B. Core Damage Sequences

[1]	8.7 x 10-8	FL2N	(4.1 x	10-4)	*	[RC4MV4 (0.17) * RCSRVLC	(0.1)]
					*	[SSFASW2H (5 x 10-3)	
					+	SSFASWF (2.5 x 10-3)	
					+	XHPR2H (3 x 10 ⁻³) + HPRF	(2×10^{-3})]

Additional Sequence

[2a]	2.5 x 10 ⁻⁸	FVL1I0 (1.9 x 10 ⁻⁴) * [CCWI5H (0.1) + CCWAVC (0.021)] * [(RC66RV0F1 + RC4MV1) (0.88) * RCSRVLC (0.1)] + [XHPR2H (3. x 10 ⁻³) + HPRF (2 x 10 ⁻³) + SSFASW2H (5 x 10 ⁻³) + SSFASWF (2.5 x 10 ⁻³)]	
[2b]	2.0 x 10 ⁻⁸	FVL1II (1.8 x 10 ⁻⁴) * [CCWI5H (0.1)] * [(RC66RV0F1 + RC4MV1) (0.88) * RCSRVLC (0.1)] * [XHPR2H (3. x 10 ⁻³) + HPRF (2 x 10 ⁻³) + SSFASW2H (5 x 10 ⁻³) + SSFASWF (2.5 x 10 ⁻³)])
[2c]	9.8 x 10 ⁻⁸	FL2II (5.4 x 10^{-3}) * [CCWMVC (0.036) + CCWAVC (0.021) + CWSRRVN (0.028)] * [RC4MV4 (0.17) * RCSRVLC (0.1)] * [XHPR2H (3. x 10^{-3}) + HPRF (2 x 10^{-3}) + SSFASW2H (5 x 10^{-3}) + SSFASWF (2.5 x 10^{-3})]	

C. Containment Safeguard States

Failure of RBCS in all sequences because LPSW is unavailable. It is assumed that the RBSS will be lost whenever the HPR hardware fails or the operator fails to transfer suction to containment sump. This case is about 40% of the total.

D. Discussion

There are no significant differences between OPRA and BNL in these sequences. In BNL, a large flood which was not isolated because of malfunction of the inlet or outlet isolation valves is also a contributor. Sequence No. 7, $F_n \overline{Q}_S Q_V U$, Bin ' Frequency = 1.6 x 10⁻⁵ yr⁻¹ (OPRA Frequency = 1.8 x 10⁻⁵ yr⁻¹)

A. Background

This sequence is shown on pages D-46 and D-47 of the OPRA. It is also a bin I case. This sequence is characterized by one of the following:

- (a) A very large flood that causes total loss of feedwater in about 5 minutes, with the PORV unable to provide sufficient relief to prevent SRVs challenging, and any of the two SRVs fail to reclose. The flood reaches critical level 3 and fails the HPI in about 30 minutes.
- (b) A large or medium flood that causes total loss of feedwater, with PORV block valve not opened by the operator in time to avoid challenging the SRVs, and any of the two SRVs fail to reclose. The flood reaches critical level 2 and fails the LPSW or it reaches critical level 1 but the break originated in the LPSW system. In addition to the loss of LPSW the alternate HPSW cooling path of the HPI motor bearing fails.

The OPRA gives six core-damage sequences. The core-damage sequences of the BNL review that correspond to those of OPRA are given below.

B. Core Damage Sequence

[1] 1.1 x 10⁻⁵ FVL2N (1.8 x 10⁻⁴) * [(RC66RV0F2 + RC4MV2) (0.6)] * [RCSRVLC (0.1)] FVL1N (1.0 x 10⁻⁵) * [(RC66RVOF1 + RC4MV1) (0.88)] 8.8 x 10-7 * [RCSRVLC (0.1)] FVL111 (1.8 x 10-4) * [(RC66RVOF1 + RC4MV1) (0.88)] [2a] 1.0 x 10⁻⁶ * [RCSRVLC (0.1)] * [CCWMVC (0.036) + CWSRVVN (0.028)] 6.9 x 10-8 FVL2II (1.8x10-5) * [(RC66RV0F2 + RC2Mv (0.6)] * [RCSRVLC (0.1)] * [(CCWMVC (0.036) + CWSRVVN (0.028)] [2b] 1.1 x 10⁻⁶ FVL110 (1.9 x 10⁻⁴) * [(RC66RVOF1 + RC4MV1) (0.88)] * [RCSRVLC (0.1)] * [CCWMVC (0.036) + CWSRRVN (0.028) + CC20AVC (3.5 x 10-3)] [2c] 2.6 x 10-7 FL1IO (5.5 x 10-4) * [(RC66RV0F3 + RC4MV3) (0.17)] * [RCSRVLC (0.1)] * [CWSRRVN (0.028)] [2d] 6.5 x 10-7 FL1N (3.8 x 10-5) * [(RC66RV0F3 + RC4MV3) (0.17)] * [RCSRSVLC (0.1)]

[3] 1.6 x 10⁻⁷ FL2N (4.1 x 10⁻⁴) * [RC4MV4 (0.17) * RCSRVLC (0.1)] * [HSHPIF (1.8 x 10-2) + SWHPSWF (4.8 x 10-3)] [4] 1.3 x 10⁻⁸ FL1II (5.3 x 10⁻⁴) * [CCWMVC (0.036) + CWSRRVN (0.028)] * [(RC66RV0F3+RC4MV'3) (0.17) * RCSRVLC (0.1)] * [HSHPIF (1.8x10-2) + SWHPSWF (4.8x10-3)] 1.8×10^{-7} FL2II (5.4 x 10^{-3}) * [CCWMVC (0.036) + CCWAVC (0.021) + CWSRRVN (0.028)] * [RC4MV4 (0.17) * (RCSRVLC (0.1)] * [HSHPIF (1.8 x 10-2) + SWHPSWF (4.8 x 10-3)7 6.3 x 10⁻⁸ FL2IO (5.1 x 10⁻³) * [CWSRRVN (0.028) [5] + CC20AVC (3.5 x 10-3)] * [RC4MV4 (0.17)] [CSRVLC (0.1)] * [HSHPIF (1.8 x 10-2) + SWHPSWF (4.8 x 10-3)] [6] 1.4×10^{-7} FMII (8.0 x 10^{-3}) * [M11F(0.034) + M321F(0.042) + CCHXFPF(1.8 x 10-3)] * [RC4MV5(0.1) * RCSRVLC(0.1)] * [HSHPIF(1.8 x 10-2) + SWHPSWF(4.8 x 10-3)] 3.0 x 10-8 FMN (2.7 x 10-3 * [M361F (0.043) + 0.006 + CCHXFPF (1.8 x 10-3)] * [RC4MV5(0.1) * RCSRVLC(0.1)] * [HSHPIF(1.8 x 10-2) + SWHPSWF(4.8 x 10-3)]

It is assumed here that time (30 min) is insufficient to recover the LPSW from unit 2.

C. Containment Safeguard States

Sequences [1], [2]: Both RBCS and RBSS unavailable. Sequences [3] to [6]: Failure of RBCS.

D. Discussion

The difference between the OPRA and the BNL sequences is small. Some comments on these differences are:

- 1. For a very large flood 2, BNL assumed that the EFW will be lost in about 10 min, and a smaller SRV challenge probability was assumed.
- Because of BNL groupings, FVLI is split into inlet and outlet side contributions.
- The contribution from large flood 1, which was separated out in BNL groupings of flood categories, is not large compared to FVLN.

4. OPRA did not consider the recovery of LPSW from unit 2 in the case of FMII. This is apparent, because of the short time (30 minutes) available in this sequence. Also, for medium flood, the time to lose EFW and to challenge the SRV is larger than in the large flood case, and BNL judges that at the time of 60 minutes after initiation of the incident the operator will have less stress. Thus, BNL used a somewhat lower value for operator not attempting to open the block valve (0.1 instead of the 0.2 in OPRA). OPRA used the same value of 0.2 for both very large and medium floods. Sequence No. 8 F_nQ_SU , Bin I, Frequency = 1.4 x 10⁻⁵ yr⁻¹ (OPRA Frequency = 1.3 x 10⁻⁵ yr⁻¹)

A. Background

This sequence is shown on pages D-47 to D-49 in OPRA. It is also a bin I sequence. This sequence is characterized by one of the following:

- A very large flood that results in failure of EFW, LPSW, and HPI. Therefore, seal cooling is lost. Failure to trip the RCP in time or failure to initiate the SSF in 30 minutes (or 90 minutes for the FL1type flood) or failure of the SSF pumps to start, either in the makeup or the ASW systems, leads to a loss of subcooling. The operator has a procedural requirement to restart the RCPs (RCPPSH). This will cause a seal LOCA beyond SSF capability.
- A large flood that results in loss of EFW and LPSW pumps. If the alternate HPSW path to HPI motor cooling fails, the HPI will be lost and the sequence proceeds as in (1) above.
- A medium flood which is in the LPSW system and therefore results in loss of EFW and LPSW as before.

The OPRA gives nine core damage sequences. The BNL core damage sequences corresponding to them are given below.

B. Core Damage Sequences

[1a] 8.7 x 10⁻⁶ FVL2N(1.8 x 10-4) * [SSF30H (0.03) + SSFSIF (0.016)] + SSFASWF (2.5x10-3)] * [RCPPSH (1.0)] 4.8 x 10-7 FVL1N (1.0 x 10-5) * [SSF30H (0.03) + SSFSIF (0.016)] + SSFASWF (2.5x10-3)] * [RCPPSH (1.0)] 8.9×10^{-7} FLIN (3.8 x 10⁻⁵) * [SSF90H (5 x 10⁻³) + SSFS1F (0.016) + SSFASWF(2.5 x 10-3)] * RCPPSH (1.0) [1b] 1.8×10^{-7} FVL1II (1.8 x 10⁻⁴) * HPRCP30H (10⁻³) FVL1N $(1.0 \times 10^{-5}) * HPRCP15H (10^{-2})$ 1x10-7 ${FVL1II (1.8 \times 10^{-4}) + [CCWMVC (0.036) + CCWAVC (0.021) + FVL2II (1.8 \times 10^{-5})}$ [2a] 8.2 x 10-7 + CWSRRVN (0.028)] * [SSF30H (0.03) + SSFSIF (0.016) + SSFASWF (2.5 x 10-3)] * RCPPSH (1.0) [2b] 6.2 x 10-7 FVL110 (1.9 x 10-4) * [CCWMVC (0.036) + CWSRRVN (0.028) + CW20AVC (3.5x10-3)] * [SSF30H (0.03) + SSFSIF (0.016) + SSFASWF (2.5x10-3)] * RCPPSH (1.0)

[2c] 3.0 x 10⁻⁷ FL1IO (5.5 x 10⁻⁴) * [CWSRRVN (0.028)] * [SSF90H (5 x 10-3) + SSFSIF (1.6 x 10-2) + SSFASWF (2.5 x 10-3)] * RCPPSH (1.0) [3a] 1.5×10^{-7} FVL1II (1.8 x 10⁻⁴) * [CCWMVC (0.036) + CCWAVC (0.021) + CWSRRVN (0.028)] * HPRCP15H (10-2) [3b] 1.3 x 10⁻⁷ FVL110 (1.9 x 10⁻⁴) * [CCWMVC (0.036) + CWSRRVN (0.028) + CW20AVC (3.5x10-3)] * HPRCP15H (10-2) FL1IO (5.5 x 10-4) * [CWSRRVN (0.028)] [3c] E * [HSHPIF (1.8 x 10-2) + SWHPSWF (4.8 x 10-3)] * [HPRCP60H (10-3)] [3d] 9.4 x 10⁻⁸ FL2N (4.1x10⁻⁴) * [HSHPIF (1.8 x 10⁻²) + SWHPSWF (4.8 x 10-3)] * [HPRCP60H (10-3)] [3e] 7.0 x 10⁻⁹ FMII (8 x 10⁻³) * [M11F(0.034) + [M321F (0.042) + CCHXFPF (1.8 x 10⁻³)} * RESW12 (0.1)] * [HSHPIF (1.8 x 10⁻²) + SWHPSWF (4.8x10⁻³)] * [HPRCP60H (10-3)] [4] 2.2 x 10^{-7} FL2N (4.1 x 10^{-4}) * [HSHP1F (1.8x10⁻²) + SWHPSWF (4.8x10⁻³)] * [SSF90H (5.10⁻³) + SSFS1F (1.6x10⁻²) + SSFASWF (2.5x10-3)] * RCPPSH (1.0) [5a] 2.5 x 10⁻⁷ FL2II (5.4 x 10⁻³) * [CCWMVC (0.036) + CCWAVC (0.021)] + CWSRRVN (0.028)] * RCPPSH (1.0) * [HSHPIF (1.8 x 10-2) + SWHPSWF (4.8 x 10-3)] * [(SSF90H (5 x 10-3) + SSFS1F (1.6 x 10-2) + SSFASW (2.5 x 10-3)] [6] 8.6 x 10^{-8} FL2IO (5.1 x 10^{-3}) * [CWSRRVN (0.028) + CW20AVC (3.5 x 10-3)] * RCPPSH (1.0) * [HSHPIF (1.8 x 10-2) + SWHPSWF (4.8 x 10-3)] * [(SSF90H (5 x 10-3) + SSFS1F (1.6 x 10-2) + SSFASW (2.5 x 10-3)] [7] 5.8 x 10⁻⁸ FL2II (5.4 x 10⁻³) * [L311F (0.005) + L21F (0.013) + CCHXFPF (1.8 x 10⁻³)] * [HSHPIF (1.8 x 10⁻²) + SWHPSWF (4.8 x 10-3)] * [S3F90H (5 x 10-3)

+ SSFSIF (1.6×10^{-2}) + SSFASWF (2.5 x 10-3)] * RCPPSH (1.0) [8] 7.4 x 10⁻⁸ FMN (2.7 x 10⁻³) * [M361F (0.043) + 0.006 + CCHXFPF (1.8X10-3)] * [HSHPIF (1.8 x 10-2) + SWHPSWF (4.8 x 10-3)] * [SSF90H (5 x 10-3) + SSFSIF (1.6 x 10-2) + SSFASWF (2.5 x 10-3)] * RCPPSH (1.0) [9] 3.3 x 10⁻⁷ FMII (8.0 x 10⁻³) * FM11F (0.034) + M321F (0.042) + CCHXFPF (1.8.10-3)] * [HSHPIF (1.8 x 10-2) + SWHPSWF (4.8 x 10-3)] * [SSF90H (5 x 10-3) + SSFSIF (1.6 x 10-2) + SSFASWF (2.5 x 10-3)] * RCPPSH (1.0)

It is assumed in the last three sequences that time (30 minutes) is insufficient to recover LPSW from unit 2.

C. Containment Safeguard States

Sequences [1] to [3] fail the LPSW and RBSS pumps, so both RBCS and RBSS are unavailable.

Sequences [4] to [9] fail LPSW only, and only RBCS is unavailable.

D. Discussion

These seal LOCA sequences are similar to the OPRA results with small differences. The addition of subcategory FLIN that can reach critical level 3 in the BNL reevaluation has a small effect of less than 10%. This is because more time to initiate the SSF is available in this case (i.e., 90 min) so that a lower value, consistent with OPRA value for not initiating the SSF in 2 hr, was used by BNL. It can be concluded, from this case and the sequences discussed earlier, that the grouping made by BNL to refine the flooding categories has an overall small impact indicating that OPRA flooding categories were adequate.

The contribution of <u>large</u> floods in the BNL analysis is lower because it is assumed that 90 minutes is available to the operator to initiate SSF before HPI pumps will be flooded. OPRA assumed the same 30 minutes as in the case of a very large flood, which is unrealistic.

APPENDIX C

FAULT TREES FOR THE POSSIBILITIES OF FLOODING REACHING CRITICAL LEVELS


in the

. 9.

























1

























APPENDIX D

REVIEW OF SEISMIC GROUND MOTION HAZARD OCONEE NUCLEAR POWER PLANT SITE OCONEE, SOUTH CAROLINA

> Professor Pradeep Talwani University of South Carolina Columbia, South Carolina

GENERAL COMMENTS

I. Seismotectonic Regions and Seismicity Parameters

In the delineation of the seismotectonic zones, and seismicity parameters there were a few serious ehertcomings.

a. Lack of documentation

In the report it is stated that they have "....identified seven source areas or "seismotectonic regions" in the eastern United States....". However, no documentation is provided to justify their delineation. No geologic, geophysical or seismicity data were given. This division appears to be based primarily on the known physiographic provinces, except in the case of the 'Charleston epicentral area' and the 'New Madrid faulted zone', where the identification appears to be based on historical seismicity. The seismotectonic regions as given in the report are shown in Figure 1 (Fig. 2-1 of the report), and we note a broad agreement with the generally accepted physiographic provinces shown in Fig. 2 (taken from Hatcher, 1978). However, there are some differences and these are discussed in a later section.

b. Source Zones Based on Seismicity

Bollinger (1973, 1975b) reviewed and analyzed the historical seismic data for southeastern United States (SEUS). Based on the historical record of seismicity, he divided SEUS into four seismic zones. Very pertinent to this study is the Southern Appalachian seismic zone (Figures 3 and 4). This

D-2



D-3



D-4





From Bollinger, 1975a



SOUTHERN APPALACHIAN SEISMICITY

Figure 4 Seismic zones in the southeastern U.S. (Bollinger, 1975b)

zone includes parts of both Blue Ridge and the Valley and Ridge tectonic provinces (the latter is included in Deformed Appalachian highlands in Fig.1). Although the tectonic provinces extend over large distances to the NE and SW (Figures 1 and 2), the historical seismicity is not as widespread. Bollinger (1973) separated the seismicity in southern Appalachians and in northern Virginia and Maryland into different seismic zones.

In my judgement, source zones based on selemicity should have been considered, especially the Southern Appalachian seismic zone. (This is discussed later in Section II.b.)

c. Probabilistic Acceleration Maps

Dr. Algermissen and his group at the U.S. Geological Survey have been studying the seismic hazards in U.S. and have brought out several reports. There are two reports which are particularly pertinent to this study (Algermissen and Perkins, 1976; and Algermissen <u>et al</u>., 1982). The authors divided the United States into various seismic zones based on geologic and seismicity data and then calculated horizontal accelerations in those zones for various exposure times based on some attenuation relationships. Figures 5-7 are taken from their work. The results of the 1976 study should have been incorporated.

D-7









D-11

1


d. Comparison of Different Approaches

For a complete analyses of the seismic hazards, the results of choosing different seismotectonic zones (see a. and b. above) and analyzing by modified Cornell's (1968) method; and the results of Algermissen probabilistic maps and the LLL . study should have been compared.

e. Current Literature Not Used

Although the report is dated May 1981, the latest reference to seismicity is Bollinger's (1975a) catalog that lists seismicity to 1974. In the interim, in 1977 the SEUS Selsmicity Bulletin began publication, and a magnitude 5+ event occurred in July 1980 in Kentucky. Thus the enalyses in the report is not based on up-to-date data at the time the <u>report was prepared</u>. (The data quality and quantity have improved considerably since 1981 with the installation of additional seismographs.)

f. Reservoir Induced Seismicity (RIS)

The Oconee Nuclear Power Plants are located adjacent to Lake Keowee (Figure 8), their source of cooling water. Swarms of microearthquakes occurred at Lake Keowee in Jan.-Feb., 1978 (Talwani <u>et al</u>., 1979), an intensity IV event occurred on Jan. 19, 1979 (Stover <u>et al</u>., 1980). Another intensity IV event occurred in the vicinity of the plant on July 13, 1971 (Sowers and Fogle, 1979). Talwani <u>et al</u>., (1979) suggested that the 1971 event and an earlier event on



Fig. 8 Location of epicenters near Lake Keowee for the period December 30, 1977 to January 16, 1978. The activity lies near the Oconee nuclear station, and trends NW-SE. Suismograph sites are shown by stars while the shallow earthquakes ($Z \leq 1$ km) are shown by solid circles.

From Talwani et al., 1979

December 13, 1969 may have been associated with the filling of Lake Keowee.

At Lake Jocassee (Figure 9), located upstream of Lake Keowee a M_{bLg} 3.7 event occurred in August 1979, nearly six years after impoundment. This shallow event (depth 2-3 km) was associated with intensity VI in the epicentral area (Taylor and Talwani, 1979). Although Lake Keowee had been filled several years prior to the construction of the Oconee plant, the possibility of RIS should perhaps have been considered. Typically in the Piedmont, RIS is shallow and associated with large ground accelerations.

DETAILED REVIEWS

II. Seismotectonic Zones

a. Section 2.1.1 Piedmont and the Upper Coastal Plain

I am in basic agreement with the delineation of this seismotectonic zone. The northwest boundary is along the Brevard zone (Figure 2). The southeast boundary of this zone lies in the Atlantic Coastal Plain. This boundary in portions of Georgia and South Carolina has a general spatial association with a change in character of the aeromagnetic anomalies, and has been interpreted as a terrane boundary by various authors (Popence and Zietz, 1977; Williams and Hatcher, 1982, 1983; and Higgins and Zietz, 1983).

However, as noted in comments Ia and Ie above, no data are provided and the references are out of date. This



F(6,9) Locations occupied by seismograph stations of the Lake Keowee and Lake Jocaciee networks are shown by solid squares.

From Talwani et al., 1979.

zone includes the Central Virginia seismic zone of Bollinger (1973) - the location of the highest seismicity in this seismotectonic zone (Bollinger and Sibol, 1985). It should perhaps have been mentioned. The Union County, S.C. earthquake of 1913 was assigned an intensity VIII (R.F.) by Taber (1913), and VII to VIII (M.M.) by Bollinger (1975a) rather than the value of VII used. This event was located some 70 miles <u>Southeast</u> of the Oconee site and not northeast as stated in the report.

b. <u>Sections 2.1.2</u> and 2.1.4. Blue Ridge and Deformed Appalachian Highlands.

The Blue Ridge seismotectonic zone (Section 2.1.2.) incorporates the Blue Ridge physiographic province (Figures 1 and 2) whereas the Deformed Appalachian Highland zone (Section 2.1.4.) incorporates the Valley and Ridge physiographic province and the eastern portion of the Cumberland plateau (Figures 1 and 2). Both these zones trend northeast along the Appalachians from Alabama to Maryland. However, the historical seismicity (Figures 3 and 4) (Bollinger, 1975a,b) and the current seismicity (Figure 10) (SEUSSN Contributors, 1985) is not so continuous. In both cases, the seismicity is dominated by the Southern Appalachian seismic zone. (SEUSSN Contributors, 1985; Johnston <u>et al</u>, 1985). Clearly this zone is the most active in the region. There were six events with magnitudes \geq 4.0 between November 1928 and June 1981 (Johnston <u>et al</u>, 1985) including the magnitude 4.6 East Tennessee



Fig. 10 Seismicity of the southeastern United States (July 1977-June 1983); earthquake ($M \ge 0$) epicenters shown by open circles (southeastern United States seismicity (July 1977-June 1983); tectonic events only).

From SEUSSN Contributors, 1985

earthquake in November 1973 (Bollinger <u>et al</u>., 1976). In the same period (10/1928 to 6/1981) there were 19 events between magnitude 3 and 4. After the installment of a local seismographic network there, Johnston <u>et al</u>., (1985) report an additional 10 events with magnitudes \geq 3.0 between 9/1981 and 12/1983. Further, like the seismicity in Giles County, Va., (Bollinger and Wheeler, 1983), which is also included in the Southern Appalachian seismic zone, the earthquakes in eastern Tennessee occur at midcrustal depths (\approx 20 km) compared to shallower than 15 km at all other locations considered in this report. (See e.g. Bollinger and Wheeler, 1983).

In view of the above, I recommend that the seismicity in the Southern Appalachian seismic zone be treated as a part of a single seismotectonic province rather than split into two provinces based on their physiographic characteristics.

c. <u>Sections 2.1.3 and 2.1.5</u> <u>Charleston Epicentral Area and</u> the New Madrid Faulted Zone

These seismotectonic source zones are based on the historically anomalous seismicity, and the ongoing seismicity at these locations. I am in basic agreement with the delineation of these zones. These zones include in them the 1886 epicentral intensity MM X (Bollinger, 1977) earthquake near Charleston and the epicentral intensity XII events in

1811-1812 near New Madrid, Mo. The Oconee site probably encountered MM VII intensities due to these shocks.

d. Section 2.1.6 Central Stable Region

This seismotectonic source zone is bounded to the southeast (Figure 1) by the Deformed Appalachian Highlands and surrounds the New Madrid Faulted Zone. The largest event considered in this region was the 1929 Attica, N.Y. event with an intensity value of VIII. No mention was made of the M_{blg} 5.0 (MMI VII) Sharpsburg, Kentucky earthquake of July 1980 (Mauk <u>et al</u>., 1982). This event was associated with MMI II-IV shaking in portions of South Carolina, although not at the Oconee site. However, I agree with the conclusion in the report that "....Because of the distance involved, the Central Stable region does not contribute significantly to seismic hazard at the Oconee site...".

c. Section 2.1.7 Florida Platform and Lower Coastal Plain

I am in agreement with the assessment in the report - both in terms of delineation of the boundaries of this zone and the absence of any significant hazard at Oconee site due to seismicity in the zone.

III. Section 2.2 Average Annual Activity Rate

In the report it is claimed that "....historical data base...." with the company was used to compute for each seismotectonic region the average number of earthquakes per year with epicentral intensities exceeding IV. The data used covered the period 1870 to 1979 - for which, it is claimed, data are complete for MM intensities greater than IV. Some of the shortcomings/omissions are listed below.

i. It is not apparent what the significance of 'activity rate' is - if it is not normalized to some specified areal extent. In the report they consider the number of earthquakes over a 110 year period - but give no consideration to the area covered in a region. Thus the Central <u>Stable</u> Region (with the largest area) has the largest annual activity rate of earthquakes with MM intensity greater than IV (.882), whereas Charleston (.173) was the second lowest (the lowest being Florida platform). However, if these observations are normalized with respect to area (Table 1) the results are perhaps more meaningful. The unnormalized activity rute is really just a measure of the number of events in that region (Table 1).

ii. It would perhaps have been more useful (and supplied a consistency check) if normalized activity rates had been obtained, and compared with the a-values for the various regions.

iii. As noted earlier, no data were provided.

iv. It is not clear if events with intensity IV were used or not. It was pointed out that historical data for events with epicentral intensity less than IV were incomplete, and rates were computed for events with intensity exceeding IV - no mention was made of events with intensity IV.

v. It was not specified in calculating these rates if aftershocks of larger events had been removed. Two of the larger events in the study, the 1886 Charleston earthquake and the 1897 Giles County, Virginia earthquake were both followed by aftershocks with intensity \geq IV.

vi. No effort was made to compare the calculated activity rates with those available in the literature. For example, Bollinger (1973, 1974) obtained the following results.

The rates of occurrence of intensity VII and stronger events are about 2.5 per century for the South Carolina-Geor_mia Seismic Zone compared to 8 per century for the Southern Appalachian region. Comparing all the zones, Bollinger (1974) noted that "....over the past century, for I_o \geq IV, the Southern Appalachian seismic zone and Central Virginia seismic zone have been equally active at 6.2 and 6.3 events per 10,000 km², respectively, while the South Carolina-Georgia seismic zone has had 3.2 events per 10,000 km². If all reported earthquakes are utilized, then the activity figures are: Southern Appalachian seismic zone, 9.8; Central Virginia seismic zone, 17.0; South Carolina-Georgia seismic zone, 31.8." The current activity (Figure 10) appears to bear out these trends.

vii. For meaningful results I recommend that seismicity rates should also be calculated for the seismic zones of Bollinger (1973), in particular for the Southern Appalachian

seismic zone. These should then be compared with the corres-

100	-	3		-	
1	a	D		e	- 1
-		~	-	-	

Seismotectonic Region*	Area (sq.mi.)*	Annual Activity Sate* Io > IV MM	Annual Activity Rate** per 10,000 sq. mi. x 10 ⁻²	No. of Events** Io > IV MM
Piedmont and Upper Coastal Plain	127,400	0.609	4.8	67
Blue Ridge	25,850	0.282	10.9	31
Charleston Zone	8,800	0.173	19.7	19
Deformed Appala- chian Highlands	63,160	0.473	7.5	52
Central Stable Highlands	309,400	0.882	2.9	97
New Madrid Zone	12,140	0.764	62.9	84
Florida Platform	210,000	0.054	0.3	6
And the state and the size that part that part has been built and the size that the				

* From Report
**This review

.

IV. Section 2.3 Epicentral Intensity Frequency Distribution

The approach used in this section is basically straightforward and standard. However, the following observations may be pertinent.

i. Most of the historical data in eastern United States are in the form of epicentral intensities. It is therefore not uncommon to list the Gutenberg-Richter relationship in terms of intensities (e.g. Chinnery, 1979). That approach has been taken in the report (eq. 1 on page J-3-16). Various empirical magnitude - intensity relationships have been obtained by different authors, and as Chinnery (1979, p. 765) notes, they are of form

Chinnery further notes that for eastern United States, the appropriate value of a₂ is 0.6. Thus for the 'intensity b-values' of 0.54 to 0.6 obtained in the report, the corresponding 'magnitude b-values' are 0.9 to 1.0 respectively.

ii. The 'intensity b-values' obtained for the various regions are in general agreement with those obtained for larger areas by Chinnery (1979). In particular, the 'intensity b-values' obtained in the report are:

	Blue Ridge	0.59
	Piedmont and Upper Coastal Plain	0.56
	Deformed Appalachian Highlands	0.59
	Charleston Epicentral Area	0.48
Whereas, those	obtained by Chinnery (1979) are :	
	South Carolina-Georgia and Scuthern Appalachian	0.46 to 0.55
	Mississippi Valley	0.59
	Southern New England	0.59
Bollinger (197	3) obtained (for the whole):	
	Southeastern United States	0.56.

iii. As noted in Section II.a. the January 1913, Union County earthquake was assigned an intensity (MM) of VII to VIII (Bollinger, 1975a). If the larger value is used, how would it effect the b-value calculated for the Piedmont and Upper Coastal Plain region? This is relevant, because the Oconee plant lies in this seismotectonic zone.

iv. As noted in Section I.b., the Southern Appalachian seismic zone should have been considered separately for calculations of seismicity parameters. Even in the modified zones discussed on page J-3-27 of the report, this has not been done.

v. Long (1974) obtained b-values for portions of southeastern United States using instrumentally recorded events and magnitude values. He obtained a (magnitude) b-values ranging between 0.9 and 1.7 for microearthquakes and their aftershocks. This should have been mentioned/incorporated in the report.

V. Section 2.4 Maximum Earthquakes

In the absence of knowledge of a causative seismogenic feature, three hypotheses concerning maximum earthquakes were considered. These will be discussed individually.

i. In the first hypothesis the maximum earthquake in each seismotectonic region is equal to the historic maximum in that region. They make the assumption that the New Madrid Faulted Zone (XII) and Charleston Epicentral Area (X) have experienced their maximum earthquakes in historical times.

Although no supporting documentation was provided, this assumption is supported by the following observations. The results of paleoseismic investigations in the New Madred Faulted Zone (Russ, 1981) suggested a recurrence rate of \approx 600 years for earthquakes with body wave magnitude equal to, or greater than 6.2. In the Charleston Epicentral area, recent paleoseismic investigations (Talwani and Cox, 1985) also suggest a recurrence time of 1500-1700 years for earthquakes with m_b \geq 6.2.

Also in magnitude-frequency plots for these regions, both the 1811-1812 New Madrid and the 1886 Charleston earthquakes are outliers - suggesting that the historic data do not cover a long enough period for these events to lie on the b-value curve.

ii. As noted earlier the largest event in the Piedmont, the Union County, S.C. earthquake of January 1913 had an epicentral intensity of VII to VIII. If the larger value is taken it will change the entry in Table I of the report from VII to VIII.

iii. The 1897 Giles County, Virginia event had an epicentral intensity of VIII. This should have been mentioned in connection with Deformed Appalachian Highland seismotectonic zone.

iv. The second hypothesis is based on Nuttli and Herrmann's (1978) procedure of assigning maximum earthquake on the basis of a 1000 year return period.

A priori this is not a very defensible method, as noted in the report (page J-3-20). However, in light of the results of paleoseismic investigations mentioned above, this method may be valid for eastern United States.

v. Using the second hypothesis, and an intensity value of VIII for the Union County earthquake will give a higher value (IX) for the maximum earthquake in the Piedmont province.

vi. In the third hypothesis, the maximum earthquake is equal to the intensity of event with a 1000 year return period (hypothesis 2) plus one intensity unit. This hypothesis was "....considered least likely hypothesis and has been assigned a probability of 0.2...".

vii. In summary, assigning an intensity VIII to the Union County event, changes the maximum earthquake in the Piedmont and Upper Coastal Plain as follows:

		Maximu Earthqual	num uake	
Hypothesis	1	VIII		
Hypothesis	2	IX	?	
Hypothesis	3	х	?	

. VI. Section 2.7 Sensitivity Analysis

Given the choice of seismotectonic zones, attenuation functions, maximum earthquakes, values of the parameter b' etc, the analyses to determine the sensitivity of the results to various hypotheses were satisfactorily performed.

Some observations:

Perhaps other attenuation relationships for eastern
 United States should also have been considered. For the
 generally accepted form of the attenuation relation:

Is = Io + a + bR + clog R

Table 2 gives a comparison of the different values available at the time the report was written.

Table 2

Region	<u>a</u>	b	C	Rema	rks	Ref
Charleston, S.C.	2.87	-0.00052	-2.88	R >	10	1
Central U.S.	3.7	-0.0011	-2.7	R >	20	2
Eastern U.S.	3.278	-0.0029	-2.277			3
Eastern U.S.	3.2	-0.00106	-2.7			4
The report	2.44	0	-3.08	R >	6 m i)	les

1. Bollinger (1977)

2. Gupta-Nuttli (1976)

3. Howell & Schultz (1974)

4. Anderson (1978)

iii. As noted earlier, the choice of source zones could have been different. It would have been useful if in the sensitivity analyses where different zones were combined, they had analyzed the Southern Appalachian seismic zone.

VII. Evaluating the Final Results

It is difficult to be sure how the 'final result' would be effected if some of the changes suggested in the previous section were incorporated.

Clearly, if intensity VIII is used as the maximum earth-. quake intensity in the Piedmont, and thus at the site, the calculated accelerations would be increased. Also, using a small concentrated Southern Appalachian seismic zone will amount to higher a-values than when that seismicity is divided into two seismotectonic zones.

Using different attenuation relations will perhaps influence the contribution of the nearer seismotectonic zones but not of the more distant ones.

Assumption of an intensity VII Reservoir Induced earthquake at Lake Keowee will provide large accelerations at higher frequencies - a factor not important for the more distant and deeper tectonic events.

References

- Algermissen, S.T. and D.M. Perkins 1976. A probabilistic estimate of maximum acceleration in rock in the contiguous United States: U.S. Geol. Surv. Open-File Report 76-416, p. 45.
- Algermissen, S.T. <u>et al</u>. 1982. Probabilistic estimates of maximum acceleration and velocity in rock in the contiguous United States: U.S. Geol. Surv. Open-File Report 82-1033.
- Anderson, J.G. 1978. On the attenuation of modified Mercalli intensity with distance in the United States: Bull. Seis. Soc. Am., <u>v. 48</u>, p. 1147-79.
- Bollinger, G.A. 1973. Seismicity of the Southeastern United States: Bull. Seis. Soc. Am., <u>v. 63</u>, p. 1785-1808. Also see 1974, Errata, <u>v. 64</u>, p. 733-734.
- Bollinger, G.A. 1975a. <u>A Catalog of Southeastern United States</u> <u>Earthquakes, 1754 through 1974</u>, <u>in Research Div. Bulletin</u> 101, Dept. of Geological Sciences, Virginia Polytechnic Institute and State University, Blacksburg, VA 24061, 68 pp.
- Bollinger, G.A. 1975b. Seismic activity in the Southeastern United States: Eq. Eng. Conf., Univ. of S.C., p. 17-47.
- Bollinger, G.A. 1977. Reinterpretation of the intensity data for the 1886 Charleston South Carolina, earthquake, <u>in</u> Rankin, D.W., ed., <u>Studies related to the Charleston, South</u> <u>Carolina, earthquake of 1886 - A preliminary report</u>: U.S. Geol. Surv. Frof. Paper 1028, p. 17-32.
- Bollinger, G.A. and Russell L. Wheeler 1983. The Giles County, Virginia, Seismic Zone: Science, v. 219, p. 1063-1065.
- Bollinger, G.A. and M.S. Sibol 1985. Seismicity, seismic reflection studies, gravity and geology of the central Virginia seismic zone: Part 1. Seismicity: Geol. Soc. of Am. Bull., <u>v. 96</u>, p. 49-57.
- Bollinger, G.A., C.J. Langer, and S.T. Harding 1976. The Eastern Tennessee earthquake sequence of October through December, 1973: Bull. Seis. Soc. Am. <u>v. 66</u>, p. 525.547.
- Chinnery, M.A. 1979. A comparison of the seismicity of three regions of the Eastern U.S.: Bull. Seis. Soc. Am., <u>v. 69</u>, p. 757-772.
- Cornell, C.A. 1968. Engineering seismic risk analysis: Bull. Seis. Soc. Am., <u>v. 58</u>, p. 1583-1606.

Gupta, I.N. and O.W. Nuttli 1976. Spatial attenuation of intensities for central U.S. earthquakes: Bull. Seis. Soc. Am., <u>V. 66</u>, p. 743-751. J

- Hatcher, R.D., Jr. 1978. Tectonics of the Western Piedmont and Blue Ridge, Southern Appalachians: Review and speculation: Am. Jour. Sci., <u>v. 278</u>, p. 276-304.
- Higgins, M.W. and I. Zietz 1983. Geologic interpretation of geophysical maps of the pre-Cretaceous "basement" beneath the Coastal Plain of the Southeastern United States, <u>in Hatcher</u>, R.D., Jr., Williams, H., and Zietz, I., eds., <u>Contributions</u> <u>to the Tectonics and Geophysics of Mountain Chains</u>: Geol. Soc. Am. Memoir 158, p. 125-130.
- Howell, B.F., Jr. and T.R. Schultz 1974. Attenuation of modified Mercalli intensity with distance from epicenters: Bull. Seis. Soc. Am., <u>V. 65</u>, p. 651-665.
- Long, L.T. 1974. Earthquake sequences and b-values in the Southeast U.S.: Bull. Seism. Soc. Am., <u>v. 64</u>, p. 267-273.
- Mauk, F.J., D. Christensen, and S. Henry 1982. The Sharpsburg, Kentucky, earthquake 17 July 1980: Main shock parameters and isoseismal maps: Bull. Seis. Soc. Am., <u>y. 72</u>, p. 221-236.
- Popence, P., and I. Zietz 1977. The nature of the geophysical basement beneath the Coastal Plain of South Carolina and northeastern Georgia, in Ranking, D.W., ed., Studies related to the Charleston, South Carolina, earthquake of 1886 - A preliminary report: U.S. Geological Survey Professional Paper 1028. p. 119-137.
- Russ, D.P. 1982. Model for assessing earthquake potential and fault activity in the New Madrid Seismic Zone, <u>in</u> Earthquakes and Earthquake Engineering in Eastern U.S., ed. J.E. Beavers, Ann Arbor Science Publishers, Michigan, 48106 <u>v.</u> <u>1</u>, p. 309-335.
- SEUSSN Contributors 1985. Availability of a six-year (1977-1983) earthquake catalog for the Southeastern United States derived from network monitoring: Bull. Seis. Soc. Am., <u>V. 75</u>, p. 629-633.
- Sowers, G.F., and G.H. Fogle 1979. Intensity survey of the Seneca, South Carolina, Earthquake, July 13, 1971: Earthquake Notes, <u>v. 50</u>, no. 1, p. 25-36.
- Stover, C.W., J.H. Minsch, G.B. Reagor, and P.K. Smith 1980. <u>Earthquakes in the United States, January-March 1979</u>: U.S. Geol. Sur. Circular 836-A, 42 pp.

Taber, Stephen 1913. The South Carolina earthquake of January 1. 1913: Bull: Seis. Soc. Am., <u>v.</u> <u>3</u>, p. 6-13.

- Talwani, P. and J. Cox 1985. Paleoseismic evidence for recurrence of earthquakes near Charleston, South Carolina: Science, <u>v. 228</u>, p. 379-381.
- Talwani, P., Donald Stevenson, David Amick, and Jin Chiang 1979. An earthquake swarm at Lake Keowee, South Carolina: Bull. Seis. Soc. Am., <u>V. 69</u>, p. 825-841.
- Taylor, K.B. and P. Talwani 1979. An isoseismal study of the August 25, 1979 Lake Jocassee earthquake, Oconee County, South Carolina: Earthquake Notes, <u>v. 50</u>, no. 3, p. 41-42.
- Williams, H., and R.D. Hatcher, Jr. 1982. Suspect terranes and accretionary history of the Appalachian orogen: Geology, <u>v. 10</u>, p. 530-536.
- Williams, H., and R.D. Hatcher, Jr. 1983. Appalachian suspect terranes in Hatcher, R.D., Jr., Williams, H., and Zietz, I., eds., <u>Contributions to the tectonics and geophysics of</u> <u>mountain chains</u>: Geol. Soc. of Am. Memoir 158, p. 33-53.

APPENDIX E

A REVIEW OF THE SEISMIC ANALYSES FOR JOCASSE DAM AND OCONEE DIKES USED IN OCONEE PRA (A PROBABILISTIC RISK ASSESSMENT OF OCONEE UNIT 3)

August 1985

Prepared by: C. Costantino and K. Bandyopadhyay

Structural Analysis Division Department of Nuclear Energy Brookhaven National National Laboratory Upton, NY 11973

E.1 INTRODUCTION

The Structural Analysis Division (SAD) of BNL has evaluated the seismic and failure analyses performed for the Jocassee Dam and Oconee Dikes. These analyses were performed as part of the seismic PRA study conducted by EPRI and Duke Power Co. for the Oconee Nuclear Station, Unit 3, and are presented in Report NSAC/60, June 1984. This review consists only of our evaluation of the description presented in the PRA report of the various studies performed. The results of this evaluation and conclusions reached are presented in the following sections of the report.

E.2 ANALYSES PERFORMED

The intent of the seismic analyses was to obtain an evaluation of the seismic fragility curves for both the Jocassee Dam and Oconee Dikes at the plant intake basin, which are to relate peak bedrock acceleration to probability of failures which can lead to flooding of the site. We will focus our comments herein more to an evaluation of the applicability of the analyses performed, rather than on the probability studies themselves, since this evaluation will lead directly to an assessment of the applicability of the derived results of the PRA study.

The slope stability analysis used is based on the Simplified Bishop Method of analysis which has been used for many years to estimate the stability of slopes in various deterministic studies. For those unfamiliar with this approach, it is based upon the equilibrium analyses of the Method of Slices applied to an assumed trial circular failure surface passed through a two-dimensional cross-section of the earth dam. Failure is assumed to occur by a rigid body rotation of the block of soil within the circular arc about the center of the circle. The safety factor of this particular failure mode is determined by comparing the resisting moments developed by shear strength countering the rotation to the driving motion initiating the rotation. For simple static problems, the driving motions are caused by gravity forces and seepage forces tending to push the soil down the slope (or about the center of Pseudodynamic effects are included by adding additional body rotation). forces proportional to the peak ground acceleration assumed for the dam. In slope stability analyses, very many trial failure surfaces are evaluated, with the one yielding the lowest factor of safety being the most critical. In addition, static analyses using other assumed failure surfaces (log spirals, block wedges, etc.) are also tried in an attempt to determine the critical failure mode, as well as the lowest factor of safety.

In the analyses described in the PRA report, the following primary assumptions were made for the stability analyses for both the dam and the dikes.

(a) Only potential circular failure surfaces were considered for both the dam and dikes. No wedge studies, which may be of particular interest for the dam, were included.

- (b) In choosing potential failure circles for the dam, only those circles which break out at or near the level of the unpounded water (and which would thus lead to overtopping of the dam) were considered. Other failure circles which may be associated with internal piping or liquefaction failure modes were not considered.
- (c) In the rigid body moment equilibrium analyses performed for each circle (for both the dam and dikes), seismic effects were included only in the driving moment computation, but not in the resisting moment computation. In addition, the impact of the vertical seismic component on the calculation was not mentioned.
- (d) Seepage forces were neglected in the analyses on the grounds that they would be of negligible effect except in the core of material of the dam.
- (e) The peak seismic force applied to the rigid soil section is determined by applying an acceleration amplification factor to the peak bedrock acceleration estimated for the site. The amplification factor used is based, apparently, upon natural period estimates of the dam obtained from other elastic anlayses.
- (f) Estimates of the probability that the peak displacement of the soil block will exceed a critical value are apparently based on simple rigid body estimates for the case where safety factor is less than 1. These are included in an attempt to yield information on probability of cracking of the dam core material.
- D.3 EVALUATION OF THE ANALYSES PERFORMED

Considering the various aspects of the analyses performed for both the dam and dikes, the following comments can be made which will lead to an evaluation of the adequacy of the results obtained.

(a) It is stated on page J-7-22 of the PRA report that

". . .the Simplified Bishop Method is generally regarded as an accurate procedure of slope stability analyses. . . Errors have been estimated to be of the order of 10%; consequently the factor λ_M has been taken to have mean value 1 and standard deviation 0.1. . . "

It should be pointed out that the error estimates mentioned above refers to the Simplified Bishop Method (used in the PRA report), as compared to the more complete "Method of Slices", including internal forces in the calculations. Both of these methods are based upon rigid body, circular failure surface analyses, with the Simplified Bishop Method merely reducing the calculation requirements by simplifying assumptions. However, it whould be realized that actual estimates of errors in safety factor are much greater than the 10% mentioned above. The basic assumptions common to both methods lead to assumptions common to both methods lead to errors much greater than 10%. That is the reason for the use of safety factors 1.5 and greater being typically used for slope stability analyses of even the simplest configurations.

- (b) The circular failure surface analyses (of which the Bishop Method is only one) is based upon the primary assumption of rigid block behavior. Its applicability is clearly to static problems wherein the time frame of interest is very large as compared to the time of propagation of stress waves through and around the dam. The method was modified in previous decades by the simple assumption of including a horizontal seismic pseudo-static inertia force. This was done because it was simple and could be handled in the precomputer age. With the advent of large finite element computer programs, more detailed analyses are now performed to try to ascertain estimates of stability of earth slopes.
- (c) By looking at the nonuniform configurations of both the dam and dikes, it is not clear that the potential circular failure surface is most critical.
- (d) In all the analyses performed, seepage effects were neglected. Yet, it is well known that they will have significant effects on safety factor, particularly for configurations similar to the dikes. For the dam, pore pressure and liquefaction effects under seismic loadings are extremely important in the core material. These have not been evaluated.
- (e) In the analyses, seismic effects were not included in evaluating resisting moments (page J-7-9 of the PRA report). Yet, in evaluating the soil shear strength along the failure surface (eq. 2), the seismic coefficient will clearly impact the intergranular stress term.
- (f) The effects of vertical seismic earthquake coefficients have not been mentioned in the analyses, although it clearly will have an impact on the calculations.
- (g) Other failure modes developed by pore pressure and soil liquefaction both through and under the dam have not been addressed. These are clearly important considerations for any dam as they perenially lead to catastrophic dam failures. This is particularly true for configurations similar to the Oconee Dikes.

APPENDIX F

SEISMIC FAULT TREES



F-2



-

.

1.0

196

F-3



F-4







F--7



F-8









F-11




F-13

6 1

34.60

APPENDIX G

A REVIEW OF THE MODELING OF FIRE PHENOMENA

AND THE ESTIMATION OF FIRE FREQUENCIES IN THE

OCONEE UNIT 3

PROBABILISTIC RISK ASSESSMENT

FOR

BROOKHAVEN NATIONAL LABORATORY

BY

BATTELLE COLUMBUS LABORATORIES

SEPTEMBER 1985

FINAL REPORT

1.0 Introduction

At the request of Brookhaven National Laboratory, Battelle Columbus Laboratories has reviewed Section 9.3 and Appendix M of the NSAC-Duke Power Company-sponsored probabilistic risk assessment (PRA) of the Oconee Unit 3 nuclear power plant; that section and appendix present the analysis of risks arising from fires within the plant areas. This report presents the findings of our review, which concentrated on the estimation of the frequency of fires and the modeling of the associated phenomena.

It is recognized in this review that the fire-risk analysis was performed as one element in the overall PRA, and forms a comparatively small part of that study. The authors of the analysis themselves recognize that several simplifying assumptions and limitations exist in the analysis; these are listed in table 9-17 of the Oconee Unit 3 PRA report, which is presented as table 1 of this report for information.

This review has considered the methods used in the fire-risks analysis for the identification of critical locations and the frequency of fire hazards, and the modeling of fire growth and suppression phenomena. As a result, we have estimated the potential effect of the principal limitations of the analysis,

It should be noted that many limitations are known to exist in the tools and methods presently used in fire-risk analyses--these are considered to be limitations in the state-of-the-art of fire analysis. Particularly the computer code, COMPBRN, used in all analyses performed to date to model

the growth of fires with time, is known to contain many simplifications, most but not all of which are conservative. These limitations have been published in previous reviews of fire-risk analyses, such as that performed by Brookhaven National Laboratory of the Limerick Generating Station Severe Accident Risk Assessment (NUREG/CR-3494, July 1984). These somewhat generic limitations will not be repeated in this review; rather, readers are recommended to read the appropriate sections of NUREG/CR-3494, such as section 2.2.1, Deterministic Fire Growth Modeling.

It is recognized that the assumptions and models used in the probabilistic fire modeling effort are of varying degrees of importance and subsequent consequences on the overall plant risk. There is no reanalyis of the fire modeling performed for the Oconee PRA. Rather, this effort is aimed at identifying potential problem areas in the analysis, and then attempting to provide the relative weight of each point in relationship to the frequency of core damage. A list of those areas which could represent nonconservative, inconsistent, or missing data or models which could have a significant impact on the plant risk has been prepared.

Summary of Findings

Overall, we judge that the core-damage frequency of fires is within a factor of four higher to eight lower of that which would be calculated by more extensive and detailed phonemena analyses. Within the state-of-the-art, such an agreement is considered comparatively close, and quite within the ranges of uncertainty expected with an analysis of this kind.

An apparent omission of the study was its lack of consideration of fires in the turbine building. Unlike most other nuclear power plants, the Oconee fite has one turbine building for all 3 units, and which contains several systems considered in the internal-events portion of the PRA. These include the instrument- and service-air systems (common to all units), the main and emergency feedwater systems, and the high- and low-pressure service-water systems. Accident sequence, involving fires in the turbine building are not expected to dominate the fire risk, and are probably negligible.

Beyond these omissions, we find that the analysis has been performed in a manner consistent with other probabilistic fire-risks analyses, and the results are also consistent with those in other studies.

2.0 Review Comments

This section reports the review findings for each of the following areas:

- o identification of critical locations
- o frequency of fire hazards
- o modeling assumptions relative to plant layout
- o fire growth and suppression modeling
- o effect of fire related phonemena on core-melt frequency.

2.1 Identification of Critical Locations

Critical locations for analysis in the fire risks analysis were reportedly selected on the basis of judgment by the analysts; no systematic survey of plant areas is presented in the study.

The two critical locations identified in the analysis are both located in the Unit 3 auxiliary building and are places where extensive cable damage could result in failure to maintain vessel inventory and cooling. The first location is the vertical cable shaft that extends from elevation 796' to elevation 833', and provides routing for cables to the equipment room and the cable room beneath the Unit 3 control room. The second location identified as critical is in the Unit 3 equipment room where, along one wall, many cables associated with safety-related equipment are located.

In addition to these locations, we have identified the turbine building as an area that may potentially be important from the fire-risks perspective. There are several systems located in this building, common to all three units of Oconee, that are important to safety. Specifically, they include the mutually redundant, though well-separated instrument-air and service-air systems, the main and auxiliary feedwater systems, and the high- and low-pressure service water systems. Of these, the two air systems and the high-pressure service water system are shared by systems for all three units.

Other parts of the review of the Oconee Unit 3 PRA have identified total loss of air systems as an important contributor in the internal-events analysis. The normal mode of operation is for the station instrument-air

system to provide control-air supplies to the pneumatically operated valves in the safety-related systems. Should it fail, supplies are automatically switched from the service-air system by a biased pneumatic valve on loss of air pressure. In the event of a fire in the turbine building it is conceivable that the instrument-air system compressors may fail (either due to mechanical or electrical damage) and the change-over valve fails because of fire damage. There is a manual valve connected in parallel with the pneumatic change-over valve located in the turbine building, but access to it may be limited during the postulated fire. The service-air system, while also located in the turbine building, is well separated from the instrument-air system (both in distance and by intervening equipment) that a fire that engulfs both is considered unlikely.

The effects of loss of the station high-pressure service-water pumps are mitigated by the presence of a large header/storage tank outside the station that would passively supply water to the high-pressure service water system (a principal duty of which is the fire-suppression system). In addition, one high-pressure service-water pump is contained within a fire-protected enclosure in the turbine building. The effect of turbine-building fires on the high-pressure service-water system's safety functions is considered small.

In addition, total loss of the Unit 3 low-pressure service-water cooling due to fires is considered very unlikely. In the event of loss of the Unit 3 low-pressure service-water pumps as a result of fire damage, connections to Units 1 and 2 low-pressure service-water system and the high-pressure service-water system are available.

The overall frequency of fires in turbine buildings has been found in other studies (e.g. Limerick Severe Accident Risk Assessment) to be about 0.012 per reactor year. However, only a small fraction of such fires would have any potential to cause major failures. In addition, any vulnerable areas in the turbine building would be limited. As such, we judge that the frequency of fires in the turbine building is about 10-5/reactor year, based on only 10% of the fires being large fires and only 1% of possible fire locations being important to safety. Assuming that this is the frequency of disabling the compressed air system, and giving credit for operator actions to supply feedwater from the safe shutdown facility or to peform "feed and bleed" operations, the contribution to core damage frequency is less than 10-6 per year.

2.2 Frequency of Fires

The initial estimation of the overall frequency of fires at Oconee Unit 3 followed the same type of analysis, and used the same database as other fire-risks analyses. The results of this process, which in essence is to divide the number of reported power-plant fires by the number of U.S. reactor years of operational experience, therefore are in agreement with other similar studies, and are judged reasonable.

This initial estimation was used in the Oconee study as a prior distribution for a two-stage Bayesian analysis to develop a plant-specific frequency distribution, using the plant experience at the time of the study, of zero fires in 4.6 reactor years. This results in a mean frequency of 2.3 \times 10-2 for the posterior distribution compared with prior distribution of 4.1 \times 10-2 per year.

Subsequent to the analysis for Oconee Unit 3, a fire did occur there. This fire occurred in May 1983 with Unit 3 at 100% power; it was caused by welding activities resulting in a high ground current level in a cable, which then caught fire. As a consequence, four engineered-safeguards valves failed in their non-ES position. This event is reported in Oconee Unit 3 Licensee Event Report 83-007, Revision 1.

Adding in the subsequent experience of one fire yields an overall fire frequency of approximately $4 \times 10-2$ per year.

In order to subdivide the frequency of fires occurring in the auxiliary building, the Oconee study pessimistically assigned a conditional likelihood of the fire being in the equipment room as 0.16 (mean value of a lognormal distribution); the same value was used for the cable shaft. Compared with other methods for subdividing building fires to particular rooms (for example, on the basis of the relative mass of combustible material), this value is judged pessimistic. (In comparison, the frequency of fires in the cable room estimated in the Oconee study is approximately double that assessed for the equivalent location in the Limerick Generating Station Severe Accident Risk Assessment. From visits to both plants, it is judged that similar masses of cables are present in both locations.) Overall, we believe that the frequencies of fires in the two critical locations are within a factor of 2 lower than those calculated in the study.

The major assumptions made relative to the conditions found in the plant are identified in those case were it is believed that such assumptions are potentially misleading. We recognize that this is a negative way of presenting the material since those assumptions that were made and believed to be valid are not explicitly identified. However, in order to provide the information in the most concise manner this is the method of presentation adopted for use in this report. Assumptions which are not identified below can be assumed to be found reasonable by the review team. In assessing the effect of each point on the fire modeling the equation for the frequency of large fires is examined:

$$F_1 = FAUX \cdot PER \cdot P_S, ER \cdot QER(tG)$$
(1)

where

F1	:	Frequency of large fires
FAUX	:	Frequency of auxiliary-building fires
PER	;	Fraction of fires in the equipment room
P _s ,ER	:	Fraction of fires that are large and in the critical locations
CER(tG)	:	Probability that a fire grows and is not suppressed

Implicit Assumptions About the Rate of Fire Growth

The location of fuel sources and the ability of a fire to cause a large or small LOCA is dependent on the rate at which a fire is spreading.

The rate of fire growth is a function of the fuel source which is available for ignition. This can be addressed in one of two ways. First, the fraction of fires that are large and in the critical areas can be modified to account for increased or decreased growth rates by increasing or decreasing the critical area by an appropriate amount. Secondly, a probabilistic analysis can be performed to include these effects in the mean growth time, t_g. The increase in the critical area or the decrease in the growth is not expected to be significant because of the good housek sping procedures and the relatively small probaility of an explosion, or extremely rapid burn, taking place.

Mechanical Failure of Valves

The location of the fuel source to the appropriate valve and the rate of fire spread could both potentially lead to the mechanical failure of the valve either to large strain effects or the melting of seal material. Thus the location of fuel sources is critical to the determination of mechanical valve failure.

This event is not expected to have a significant influence because there are no pumps or valves located in the equipment room and the auxiliary and turbine buildings have alternate and backup routing systems for the safety critical components. The largest effect would be in the containment structure for which no results are available.

Location Of Fuel Sources

The location of combustible materials for the design of the plant does not appear to present any problem. However, the material and structures, which are used during plant maintenance and are easily relocated, do have a potential for affecting the results of the probabilistic fire analysis. There are many such instances in the turbine and auxiliary buildings with the following three fuel sources being the most significant: temporary wood work platforms, oil-storage facilities, and resin containers with ammonium nitrates. Additionally, two potential ignition source which are movable are the welding machines and pipe leakage onto the electrical circuitry.

The effect of additional fuel sources on the calculated results enters into the expression for the probability that a fire grows and is not suppressed. Since this probability is already approximately equal to 0.75 (Section 9.3.3.5) the change in the growth time (even to zero) is at most 1.25, assuming the exponential model.

2.4 Fire-Growth and -Suppression Modeling

The physical model used for the fire growth calculations, COMPBRN, is a relatively simple fire model that accounts for fires spreading. While it can be argued that the uncertainties inherent in the fire process do not justify the added complexity of a more detailed fire model it is also true that there is little justification for introducing additional uncertainties which cannot be quantified by using a model which does not account for the individual processes in an accurate fashion. Therefore, the following points are made about the fire-model computer code and its inter-action with the suppression modeling.

Upper Layer Calculations

The version of COMPBRN used in the Oconee probabilistic fire model does not accurately account for layer effects in the upper portions of the room. The temperature differential between the gases will have an effect on the smoke detector response times. These upper-layer effects enter in some of the other points made below.

Smoke Effects

The effect of the smoke is crucial from two standpoints: (1) the interaction of the smoke with the heat transfer and (2) the smoke dispersion relative to the detector locations. In the case of the areas of concern at Oconee, fire detection is primarily by means of smoke detectors. For example, six detectors are located in the Unit 3 cable room (Table 9.5-1, Oconee FSAR). Hence, it is judged unlikely that smoke will interfere with fire detection; it will rather increase the likelihood of detection.

Loss of Fire-Suppression Capability

The analysis does not consider the possibility of the fire reaching a stage at which the energy release exceeds the capacity of the fire suppression system. This can occur for two reasons: (1) the fire grows unchecked for a long enough period of time that the fire suppression system cannot remove sufficient amounts of energy from the compartment to contain the fire, or, (2) the fire causes an explosion or rapid pressure increase that ruptures or incapcitates the suppression. One of the limits to the analysis which has been performed is that the availability of fuel sources has only been considered from a implicitly imposed viewpoint. Movable sources are not directly addressed and thus such rapid releases of energy are not considered.

Elashover

The phenomenon of flashover is not a well understood process, however it is also poorly modeled in COMPBRN. To insure that the critical components to a safe shutdown of the reactor are not damaged during a fire it is necessary to investigate this phenomena more closely.

Fire Doors and Accessways

While the effect of open fire doors is more properly a topic for human reliability analysis it is worthwhile to investigate the effect of doors being left open on the results to as ess their importance to the subsequent consequences of a fire. Additionally a model better able to address the vents and openings that can either provide oxygen to the fire or carry combustible material out of or into the compartment as airborne particles should be examined further.

Toxic Gas Production

It is necessary to assess the production of toxic gases to correctly determine the actual response to a fire by plant personnel. In addition there is evidence that corrosive gas releases can affect circuitry up to 40 feet from the fire. This results in a nonconservative assumption which can also feedback to the calculation of the frequency of fires occurring in the fire critical area, discussed above.

Effect of COMPBRN Assumptions On Core Melt Frequency

In terms of equation (1) there are two ways in which the modeling assumptions in COMPBRN can affect the results of the calculation of the core melt frequency. The first is on the probability that a fire grows and is not suppressed and the second is on the fraction of fires which are large and in the critical locations.

The interaction of the growth model and the suppression model implicitly assumes, as is standard practice in all fire risk analyses, that the fire growth and fire suppression models act independently of each other. This not only has an effect on the physical modeling of the fire but on the probabilistic calculations when variables are assumed to be independent.

The upper layer calculations can affect the modeling in two ways. Either the upper layer temperature increases more quickly than COMPBRN predicts or the upper layer reflects heat (due to smoke and gas generation) back into the room. In the first instance the detection and suppression systems, if they are operational, will detect the fire more quickly and lessen the effect of the fire on the core melt. Because of the short time for growth this will be a minor effect. The second possible response is the more important one. This is because if the upper layer acts as a shield for the detection system then the mean time to detection, and thus activation of the suppression system, increases. If the probability that a fire grows and is not suppressed, Per, is set to its upper limit of 1.0 then the maximum effect that the upper layer effects can have on the core melt frequency is to increase the probability by a factor of 1.6; this is an upper bound.

On the other hand, the generation of smoke may alternate the heat transfer from the flames to the cables. The effect of interfering with the heat

transfer can be simply illustrated. Equation M-5 of the Oconee study identifies that the time to ignition (or failure for components), t*, is given by:

$$t^* = \prod \frac{k(T^* - T_0)^2}{4\chi}$$

where qo" is the heat flux impinging on the object. Smoke between the pilot fire and the object will reduce the heat flux that strikes the object. An opacity of 50% will increase t* by a factor of 2. Using the Electrical Equipment Room as an example, the mean time for propagation of fire to the uppermost tray of cables is 9.3 minutes (Section 9.3.3.3). The heat transfer is both radiative and convective, however, so an opacity of 50% would yield an upper bound increase in the fire-propogation time of 2, to 19 minutes. Such an increase would increase the likelihood of detection and suppression from 0.38 to approximately 0.7. In addition, a somewhat larger pilot fire is required.

Considerable uncertainty exists in the expected cpacity of smoke however, since this would depend on the specific material in the fires. The materials considered in the Oconee study (oil, cable insulation) however, do produce significant smoke. For the purposes of this review, a factor of 2 increase in the time for propagation (and hence a factor of 2 decrease in the accident-sequence frequencies) is considered suitable for illustration.

Another manner in which an even relatively small fire can affect the safety system cabling is in the generation of corrosive gases which can short out equipment without generating sufficient heat to trip the detection system. In this case of Oconee, where multiple smoke detectors are installed, this is considered a small concern. The one area in which the assumption that the fire growth and fire suppression are independent, can have a significant effect on the calculation of the core melt frequency is in the calculation of the probability that the fire is large and near a critical location, Ps,er. This probability is determined from several assumptions, two of which are critical for the current discussion:

- 1. The critical area constitutes a small fraction (about one-tenth) of the total floor area of the electrical equipment room.
- A significant fraction (about one-tenth) of the cable insulation fires reported . . . were estimated to be large fires.

Because the interaction of the growth and suppression models can increase the critical area in the equipment room this probability could be non-conservative. In addition the effect of the combination of the upper layer effects, corrosive gas generation, flashover and longer times to detection can lead to the definition of large fires being modified so that 400 BTU fires are not lower bound fires. While it is outside the scope of this review to develop a more detailed model to assess the effects of such synergestic effects it is judged that these two simplistic modeling assumptions could cause the frequency of large fires in critical areas to be optimistic by a factor of 2-4.

SUMMARY

The overall plant fire risk analysis performed for the Oconee nuclear power plant appears to be a reasonable first approximation based on the results of the review to date with the exception of the limitation of fire risk areas being limited to the auxiliary building, specifically the equipment room and cable spreading area.

Table 2 summarizes the various factors that have been identified in the review, together with an expected range of effect on the accident frequencies -- these effects have been applied to the overall mean frequency of core damage due to fires, to provide the range within which a more detailed study would be expected to yield results. This range is from 2.5 \times 10-6 to 8 \times 10-5 per reactor year based on the overall mean frequency calculated in the Oconee 3 PRA of 1 \times 10-5 per reactor year.

Limiting factor	Comment		
Probability of specific locations of fires	Based on a review of data and an analysis of the specific areas in relationship to the entire auxiliary building. Considerable analyst judgment involved.		
Locations of critical fires	Based on review of systems, areas, and loca- tions of important equipment. The areas identified as important may not be the only ones that could result in fire risk.		
Cable routings	A great deal of uncertainty since detailed information was not available. A number of conservative assumptions had to be made con- cerning vital equipment.		
Failure modes	Hot-short calculations used to identify proba- bility of spurious actuation are heavily influenced by analysts' judgment. Detailed data do not exist.		
Fire growth	Fire progagation is based on physical models, and there are large uncertainties about the results of these models. The analysis included consideration of, but not direct data from, tests on Oconee interlocked armor cable.		
Fire suppression	Fire suppression is based on industry-wide data and is not necessarily directly represen- tative of the actual characteristics of the fire areas of concern.		
Operations staff effects	Errors of commission by the control-room operators as instigated by failures in the instrumentation circuits were not analyzed explicitly. It was judged that the loss of function from fires in the critical areas envelops these potential human errors.		
Smoke progagation	The effects of smoke on the operations staff were not analyzed explicitly.		
Flooding from fire- suppression activities	The effects of flooding from fire-fighting activities were not analyzed explicitly.		

Table 1 Fire-Analysis Limitations

	Tab	le	2
Summa	гу	of	Review

Finding		Potential Effect on Sequence Frequencies (ranges)	
1.	Frequency of fire in identified zones	× 0.5 to × 1.0	
2.	Additional fuel sources	× 1 to × 1.25	
3.	Upper-layer shielding of detectors	x 1 to x 1.6	
4.	Effect of smoke opacity of propogation	× 0.5 to × 1	
5.	Assumptions of synergism of areas and fraction of large fires	× 1 to × 4	

NRC FORM 335 12 84 NRCM 3102 3201 372 SEE INSTRUCTIONS ON THE REVERSE	NUREG/CR-4374, Vol. 2 BNL-NUREG-51917
A Review of the Oconee-3 Probabilistic Risk Assessment: External Events, Core Damage Frequency	3 LEAVE BLANK
	January 1986
N. A. Hanan, D. Ilberg, D. Xue, P. Youngblood, J. W. Reed, M. McCann, T. Talwani, J. Wreathall, P. D. Kurth, K. Bandyopadhyay, C. Costantino	6 DATE REPORT ISSUED
2 PERFORMING ORGANIZATION NAME AND MAILING ADDRESS //www.de Co.	8 PROJECT/TASK/WORK UNIT NUMBER
Brookhaven National Laboratory Upton, New York 11973	B FIN OF GRANT NUMBER FIN A3797
10 SPONSORING ORGANIZATION NAME AND MAILING ADDRESS (Include Zip Code)	114 TYPE OF REPORT
Division of Safety Review and Oversight	Technical
Office of Nuclear Reactor Regulation	PERIOD COVERED (Inclusive detes)
U.S. Nuclear Regulatory Commission Washington, DC 20555	
12 SUPPLEMENTARY NOTES	
ble) the OPRA assessment of the important sequences and lead to core damage. The review included a tec tions and methods used in the OPRA within its state information available. Within this scope, BNL perf the accident sequences generated by internal floods tailed review (in some cases a scoping review) for th fires, tornadoes, external floods, and aircraft impac	s that are "externally" generated chnical assessment of the assump- ed objective and with the limited formed a detailed reevaluation of s and earthquakes and a less de- ne accident sequences generated by tt.
14 DOCUMENT ANALYSIS - + KEYWORDSIDESCRIPTORS	IS AVAILABILITY
Nuclear power plant Oconee Unit 3 Probabilistic risk assessment (PRA) DIDENTIFIERS OPEN ENDED TERMS	e frequency y experience Unlimited If SECURITY CLASSIFICATION IT A APPRIL Unclassified IT A APPRIL Unclassified IN NUMBER OF PAGES IN PRICE